ANALYTIC SOLUTIONS OF SCALAR FIELD COSMOLOGY WITH MINIMAL AND NONMINIMAL COUPLING AND DEFORMED DISCRETE AND FINITE QUANTUM SYSTEMS

by Medine İldeş B.S., Physics, Boğaziçi University, 2003 M.S., Physics, Boğaziçi University, 2006

Submitted to the Institute for Graduate Studies in Science and Engineering in partial fulfillment of the requirements for the degree of Doctor of Philosophy

> Graduate Program in Physics Boğaziçi University 2022

ACKNOWLEDGEMENTS

This thesis is dedicated to my family. For their endless love, support and encouragement...

First, I would like to express my gratitude to my co-advisor Metin Arık for sharing his knowledge, guidance and patience throughout all these years. To work with him is my pleasure and my honor.

I am eternally grateful to Teoman Turgut for being my thesis supervisor after Metin Arık's retirement and for fruitful discussions during all times.

I would like to thank Ali Yıldız, Levent Akant, Savaş Arapoğlu and Dieter Van den Bleeken for reading this thesis.

I would like to also thank Mikhail Sheftel and Aşkın Ankay for accepting me as a researcher in their projects during these years. I will never forget their support and kindness.

Most part of the studies in thesis is financially supported by Bogaziçi University from the Scientific Research Fund (BAP), research project No. 16521 and research project No.11643.

ABSTRACT

ANALYTIC SOLUTIONS OF SCALAR FIELD COSMOLOGY WITH MINIMAL AND NONMINIMAL COUPLING AND DEFORMED DISCRETE AND FINITE QUANTUM SYSTEMS

In this thesis first, we study analytic solutions of cosmology. We investigate the most general cosmological model with real scalar field which is minimally coupled to gravity and Brans-Dicke cosmology. Field equations consist of three differential equations. We switch independent variable from time to scale factor by change of variable $\dot{a}/a = H(a)$. Thus a new set of differential equations are analytically solvable with known methods. a(t) can be explicitly found as long as methods of integration techniques are available. We investigate the dynamics of the universe at early times as well as at late times in light of these formulas. We find mathematical machinery which turns on and turns off early accelerated expansion. On the other hand late time accelerated expansion is explained by cosmic domain walls. ϕ^4 potential is studied in Brans-Dicke Cosmology. In this thesis we also study discrete and finite quantum systems. We define a deformed kinetic energy operator for a discrete position space with a finite number of points. The structure may be either periodic or nonperiodic with well-defined end points. It is shown that for the nonperiodic case the translation operator becomes nonunitary due to the end points. This uniquely defines an algebra which has the desired unique representation. Energy eigenvalues and energy wave functions for both cases are found. In addition, we uncover the mathematical structure of the Schwinger algebra and introduce almost unitary Schwinger operators which are derived by considering translation operators on a finite lattice.

ÖZET

SKALAR ALANLA MİNİMAL VE MİNİMAL OLMAYAN BAĞLANTIDAKİ KOZMOLOJİNİN ANALİTİK ÇÖZÜMLERİ VE DEFORME AYRIK VE SONLU KUANTUM SİSTEMLERİ

Bu tezde öncelikle kozmolojinin analitik çözümlerini inceliyoruz. Gerçek skalar alanla yerçekimi arasında minimal bağlantının olduğu en genel kozmolojik modeli ve Brans-Dicke kozmolojisi araştırıyoruz. Alan denklemleri üç diferansiyel denklemden oluşur. $\dot{a}/a = H(a)$ dönüşümü ile bağımsız değişkeni zamandan ölçek faktörüne değiştiririz. Böylece yeni diferansiyel denklem seti, bilinen yöntemlerle analitik olarak çözülebilir. Entegrasyon teknikleri yöntemleri mevcut olduğu sürece, a(t) açıkça bulunabilir. Evrenin dinamiklerini bu formüller ışığında hem erken hem de geç zamanlar için inceliyoruz. Erken ivmelenen genişlemeyi açan ve kapatan matematiksel mekanizma buluyoruz. Öte yandan, geç zamandaki ivmelenen genişlemeyi, kozmik alan duvarları ile açıkladık. ϕ^4 potansiyeli Brans-Dicke Kozmolojisinde incelendi. Bu tezde ayrıca ayrık ve sonlu kuantum sistemlerini de inceliyoruz. Sonlu sayıda noktaya sahip ayrık bir konum uzayı için deforme olmuş bir kinetik enerji operatörü tanımlıyoruz. Yapı, iyi tanımlanmış uç noktaları olan periyodik veya periyodik olmayan olabilir. Periyodik olmayan durum için, öteleme operatörünün bitiş noktalarından dolayı üniter olmayan hale geldiği gösterilmiştir. Bu, istenen tek temsile sahip bir cebiri tek şekilde tanımlar. Her iki durum için de enerji özdeğerleri ve enerji öz vektörleri bulunur. Ek olarak, Schwinger cebirinin matematiksel yapısını ortaya çıkardık ve sonlu bir kafes üzerindeki öteleme operatörlerini dikkate alarak türetilen neredevse üniter Schwinger operatörlerini tanıttık.

TABLE OF CONTENTS

ACKI	NO	WLED	GEMEN'	TS	iii
ABST	ΓRA	ACT .			iv
ÖZET	Γ.				v
LIST	OF	F FIGU	RES		xi
LIST	OF	F TABI	LES		xii
LIST	OF	F SYMI	BOLS		xiii
LIST	OF	F ACRO	ONYMS/	ABBREVIATIONS	XV
1. IN	ITF	RODUC	CTION .		1
1.	1.	Analyt	ic Solutio	ons of Cosmology	1
		1.1.1.	Problem	s of Cosmology	3
			1.1.1.1.	Flatness Problem	3
			1.1.1.2.	Horizon Problem	3
			1.1.1.3.	Inflationary Cosmology	4
			1.1.1.4.	Cosmological Constant Problem	4
		1.1.2.	Dark Ma	atter and Mond Theory	5
		1.1.3.	Cosmic 1	Domain Walls	6
		1.1.4.	Field Eq	luations	8
1.2	2.	Deform	ned Discr	ete and Finite Quantum Systems	9
2. Al	NA	LYTIC	SOLUT	IONS OF SCALAR FIELD COSMOLOGY WITH MINI-	
М	AL	COUF	PLING .		11
2.1	1.	Field B	Equations	3	11
		2.1.1.	Original	Form	11
		2.1.2.	New For	m of Field Equations	13
2.2	2.	Solutio	on for Fie	ld Equations	15
		2.2.1.	Solution	for Given $V(a)$	16
		2.2.2.	Solution	for Given $\phi(a)$	17
			2.2.2.1.	Singular Case	18
			2.2.2.2.	Non-Singular Case	20

222.2.3.232.2.4.25General Solution for $\rho(a) = \frac{\rho_n}{a^n}$ 2.3.1.252.3.1.1.252.3.1.2.28292.3.2.2.3.2.1.292.3.2.2.32342.4.1.Dark Energy 352.4.1.1. 352.4.1.2.36 2.4.2.Cosmic Domain Walls 38 2.4.2.1. 382.4.2.2. 392.4.3.41 2.4.4.422.4.4.1.Combination with Radiation 43Combination with Domain Walls 2.4.4.2.442.4.4.3.4444 Cosmic Walls 2.5.1.45Dark Energy 2.5.2.51553. ANALYTIC SOLUTIONS OF BRANS-DICKE COSMOLOGY 5757Mathematical Techniques 59

3.3. 60 3.3.1.613.3.1.1.61

3.1.

3.2.

			3.3.1.2. Singular Case $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$
		3.3.2.	Method 2
			3.3.2.1. Non-Singular Case $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$
		3.3.3.	Method 3
			3.3.3.1. Non-Singular Case
			3.3.3.2. Singular Cases $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 65$
		3.3.4.	Method 4
			3.3.4.1. Non-Singular Case $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$
			3.3.4.2. Singular Cases $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$
	3.4.	Soluti	on for Given $V(a)$ and $\phi(a)$
		3.4.1.	Method1
			3.4.1.1. Non-Singular Case $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots $ 71
			3.4.1.2. Singular Case $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$
		3.4.2.	Method 2 \ldots \ldots \ldots $.$ 72
			3.4.2.1. Non-Singular Case $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots $
			3.4.2.2. Singular Cases $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 73$
	3.5.	Soluti	on for Given $H(a)$ and $\phi(a)$
		3.5.1.	Method1
		3.5.2.	Method2
	3.6.	Single	Component Universe
	3.7.	Early	Universe, Dark Energy and Radiation
	3.8.	Late 7	Fime Expansion of the Universe 85
		3.8.1.	Dark Energy Dominated Universe
		3.8.2.	Domain Wall Dominated Universe
	3.9.	Discus	ssion
4.	INF	LATIO	N AND LINEAR EXPANSION IN THE RADIATION DOMINATED
	ERA	IN BI	RANS-DICKE COSMOLOGY
	4.1.	Radia	tion Dominated Era 98
		4.1.1.	The Scale Invariant Solution
		4.1.2.	Linearly Expanding Radiation Dominated Universe 100
		4.1.3.	Creation of Matter in the Late Radiation Dominated Era 102

	4.2. Early Inflation in the Radiation Dominated Era	105
	4.3. Temperature Calculations	108
	4.4. Matching the Solutions	110
	4.5. Discussion	112
5.	QUANTUM MECHANICS IN A SPACE WITH A FINITE NUMBER OF	F
	POINTS	116
	5.1. Nonunitary Translation Operators	118
	5.2. Eigenvalues and Eigenfunctions of the Hamiltonian for the Nonperiodic	
	Space	124
	5.3. Eigenvalues and Eigenfunctions of the Hamiltonian for the Periodic Space	e127
	5.4. Discussion	130
6.	QUANTUM MECHANICS ON PERIODIC AND NON-PERIODIC LATTICE	S
	AND ALMOST UNITARY SCHWINGER OPERATORS	131
	6.1. Mathematical Structure of the Almost Unitary Translation Operators .	132
	6.2. The Schwinger Algebra in Terms of the Almost Unitary Translation	
	Operators	134
	6.3. Almost Unitary Operators in Terms of the Schwinger Algebra	140
	6.4. New Representations for Basis of $M_N(C)$	143
	6.5. Multi-Dimensional Lattice in Terms of Lower Dimensional Lattices	144
	6.6. Discussion	145
7.	CONCLUSION	148
RI	EFERENCES	151
AI	PPENDIX A: APPENDIX FOR CHAPTER 2	172
	A.1. Einstein Tensor	172
	A.2. Action Variation for Non-minimal Coupling	175
	A.2.1. Action Variation with Respect to Metric	175
	A.2.1.1. Action Principle	179
	A.2.2. Action Variation with Respect to the Scalar Field	181
	A.3. Change of Independent Variable	183
	A.4. Integration by Parts	184
	A.5. The Linear First Order Differential Equation	185

A.6. Hypergeometric Functions	185
A.6.1. Some Integrals Expressed as Hypergeometric Function	188
A.7. Integral Expressed as Erfi Function	191
A.8. Cosmological Fluid	193
A.9. Redshift in Cosmology	194
APPENDIX B: APPENDIX FOR CHAPTER 3	195
B.1. Action Variation for Brans-Dicke Cosmology	195
B.1.1. Action Variation with Respect to Metric	195
B.1.1.1. Some Necessary Calculations	196
B.1.2. Action Variation with Respect to the Scalar Field	198
B.2. The Bernoulli Equation	198
B.3. Some Necessary Calculations	199
APPENDIX C: APPENDIX FOR CHAPTER 4	201
APPENDIX D: APPENDIX FOR CHAPTER 5	203
APPENDIX E: APPENDIX FOR CHAPTER 6	204
E.1. Projection Operators in Terms of a and a^{\dagger}	204
E.2. Projection Operators in Terms of U and V	208
E.3. $M_N(C)$ in Terms of a and a^{\dagger}	212
E.4. $M_N(C)$ in Terms of U and V $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots $	213

LIST OF FIGURES

Figure 2.1.	Distance Modulus vs Redshift Plot for Domain Wall Dominated		
	Universe	51	
Figure 2.2.	Distance Modulus vs Redshift Plot for $M = -19.25.$	54	
Figure 2.3.	Distance Modulus vs Redshift Plot for $M = -19.30.$	54	
Figure 3.1.	Distance Modulus vs Redshift Plot for $M=-19.30~{\rm Reproduced}~$.	89	
Figure 6.1.	$d_1 \times d_2$ Lattice for $d_1 = 4$ and $d_2 = 3$	145	

LIST OF TABLES

Table 2.1.	Values of H_0 and M for Given Ω_w	49
Table 2.2.	Values of M and Ω_w for Given H_0	49
Table 2.3.	Values of H_0 and Ω_w for Given M	50
Table 2.4.	Values of H_0 and M for Given Ω_0	52
Table 2.5.	Values of M and Ω_0 for Given H_0	53
Table 2.6.	Values of H_0 and Ω_0 for Given M	54

LIST OF SYMBOLS

$(a)_n$	Pochhammer symbol
\mathscr{A}_d	An algebra on d -dimensional vector space
dx^{lpha}	Coordinate basis one-form
e^i	Orthonormal basis one-form
e_{ij}	Basis of $M_N(C)$
F(a,b;c;z)	The hypergeometric function
$_{p}F_{q}(a;b;z)$	The generalized hypergeometric function
$g_{\mu u}$	Metric tensor
-g	Determinant of the metric tensor $g_{\mu\nu}$
G_N	Newtonian gravitational constant
$G_{\mu u}$	Einstein tensor
H(a)	Hubble function
H_0	Present value of the Hubble function
${\cal L}$	Lagrangian density
\mathscr{L}_d	A linear lattice with d elements
$M_N(C)$	Complex Matrix algebra in N-dimension
$R^{\mu}_{ u\sigma ho}$	Riemann curvature tensor
$R^{\mu}_{ u}$	Ricci tensor
R	Ricci scalar
S	Action
$T_{\mu u}$	Energy momentum tensor
u^{μ}	4-velocity vector for fluid
Φ	Original Brans-Dicke scalar field
ϕ	Scalar field
Ω	Density parameter
Ω^{μ}_{nu}	Curvature two-form
$\Gamma^{\sigma}_{\mu u}$	Christoffel symbol

Brans-Dicke coupling constant Connection one-form

 ω ω^i_j

LIST OF ACRONYMS/ABBREVIATIONS

BDJT	Brans Dicke Jordan Thiry
CMB	Cosmic microwave background
EFE	External field effect
FLWR	Friedmann Lemaitre Robertson Walker
MOND	Modified Newtonian Dynamics
Mpc	Mega parsec
RMOND	Relativistic Modified Newtonian Dynamics
ACDM	Lambda cold dark matter

1. INTRODUCTION

This thesis consists of two main parts: cosmology and deformed quantum mechanical finite systems. In the first part we aim to find exact solutions of field equations both in the most general cosmological model with real scalar field which is minimally coupled to gravity and in Brans-Dicke cosmology. Early epoch and late-time era of the universe are investigated. In the second part we focus on a discrete position space with a finite number of points. We study momentum and translation operators both on periodic and nonperiodic structures. We construct a relation between these translation operators and Schwinger operators.

1.1. Analytic Solutions of Cosmology

The scalar field plays an important role in many parts of modern physics. Its usage in cosmology was seen firstly in Nordstrom's studies after Newton's gravity which has a scalar potential field. Although he introduced scalar theory of gravity [1–4] in 1912-1913, none of them have been verified by observation [5]. Then in 1916 Einstein's theory of general gravity was established. This is a purely tensor theory. Seeds of some alternative theories which incorporates a scalar field were conceived by Jordan [6] and Dirac [7].

Since Einstein's General Theory of Relativity, many new modified theories have been introduced to extend it. Dirac proposed that Newton's constant may be varying with time [7], after influential ideas were developed by Weyl [8,9] and Eddington [10]. Spacetime dependent gravitational constant can be described according to Jordan's study in [6]. Thus his work has been accepted as the origin of the scalar-tensor theory. Scalar fields appear through a nonminimal coupling term. This term arises with the idea of the space-time dependent gravitational constant which was conceived by Jordan [6] and Dirac [7]. A general form of the Lagrangian density for scalar-tensor theory [11–14] can be written as

$$\mathcal{L} = \frac{1}{16\pi} \sqrt{-g} [f(\Phi)R - g(\Phi)\nabla_{\mu}\Phi\nabla^{\mu}\Phi - 2\Lambda(\Phi)] + \mathcal{L}_m(\psi, h(\Phi)g_{\mu\nu}), \qquad (1.1)$$

where f, g, h and Λ are arbitrary functions of the scalar field Φ and \mathcal{L}_m is the Lagrangian density of the matter fields ψ .

Two different revolutionary steps have been taken in cosmology since 1980. First one was the construction of inflationary cosmology by A.H. Guth and A.D.Linde. It had been proposed as a solution for problems of the standard model of cosmology which are the flatness problem, the horizon problem and the monopole problem [15, 16]. In these studies it has been shown that one or more scalar fields drive the early phase of accelerated expansion. Second one was the observational evidence of accelerated expansion of the present universe [17–19]. Standard cosmology explains this behaviour by contribution of dark energy (~ 68%), dark matter (~ 27%) and baryonic matter (~ 5%) to the total density parameter. A Simple explanation of dark energy just as a cosmological constant in standard model of cosmology is problematic [20]. Hence to explain dark energy many different studies have been developed by using scalar fields similar to early inflationary theories. All these models are widely explained in the review article [21]. The second main part of the universe consists of dark matter. This still keeps its secrets. It has not been explained properly yet. A scalar field is again a candidate to reveal its nature [22].

In 1961 Brans and Dicke [23] developed a new approach to general relativity based on Mach's principle. Their main idea was to replace the effective gravitational constant with the reciprocal of a scalar field. The so called Jordan-Brans-Dicke Lagrangian has given inspiration to many scientists. For example the same Lagrangian with spontaneous symmetry breaking potential has been studied by Zee [24] and Smolin [25]. Induced-gravity inflation models have been constructed by using these ideas [26–31]. Moreover in extended inflation, a simple potential causes accelerated expansion [32–34]. On the other hand, more complicated nonminimal coupling terms have been studied in [35–43]. Finally, one should also remember multifield inflationary scenarios which have been developed in the last decade [44–50]. These theories include multiple nonminimally coupled fields.

1.1.1. Problems of Cosmology

Inflationary cosmology was established to solve the problems of Hot Big Bang scenario. Two main problems can be explained as briefly;

<u>1.1.1.1. Flatness Problem.</u> Einstein's equations which will be introduced in Section 2 state that

$$H^{2} + \frac{k}{a^{2}} = \frac{8\pi G}{3} \sum_{i} \rho_{i}(t).$$
(1.2)

The critical energy density is defined for a flat universe as $\rho_c = \frac{3H^2}{8\pi G}$. Then one can arrange (1.2) as

$$\frac{\mid \rho_{total}(t) - \rho_c \mid}{\rho_c} = \mid \Omega_k \mid, \qquad \Omega_k = -\frac{k}{a^2 H^2}.$$
(1.3)

Present observations indicate that the ratio $\Omega = \frac{\rho}{\rho_c}$ is very close to one [51,52]. One can show that at the Planck time $t \sim 5 \times 10^{-44} s$, $|\Omega_P - 1| \leq 10^{-60}$. The coincidence of this fine tuning initial condition is known as the flatness problem.

<u>1.1.1.2. Horizon Problem.</u> Cosmic microwave background radiation implies homogeneity and isotropy of our universe while it has temperature variation $\frac{\delta T}{T} \simeq 10^{-5}$. However if one takes two opposite points which are separated by 180⁰ on the last scattering surface a proper distance between them is $1.96d_{hor}(t_0)$ [53]. Thus there is the problem how causally disconnected regions can be in thermal equilibrium. <u>1.1.1.3.</u> Inflationary Cosmology. To solve these two basic problems it was shown that there should be inflationary era preceding the radiation dominated epoch. An acceleration of the scale factor $\ddot{a} > 0$ or $-\frac{\dot{H}}{H^2} < 1$ are konown as the condition for inflation. The usual assumption states that the Hubble function was constant during this period of time. Thus an exponential expansion $a \sim e^{H_i(t-t_i)}$ of the universe exists [54, 55]. Then at the end of the inflation

$$|\Omega_f - 1| = e^{-2N} |\Omega_i - 1|, \qquad (1.4)$$

$$\mid \Omega_{kf} \mid = e^{-2N} \mid \Omega_{ki} \mid, \tag{1.5}$$

$$N = (t_f - t_i)H_i, \tag{1.6}$$

where *i* and *f* denote beginning and end of the inflation. Calculations in [54] indicate that N > 62. If at the beginning of inflation the universe was curved such that $|\Omega_{\kappa}| \sim 1$ exponential expansion prevents flatness problem.

When we trace back the exponential or accelerated expansion of the universe, today's casually disconnected patches of the cosmic sky have a chance to communicate with each other. Hence inflationary solution also explain the horizon problem.

Inflationary period is assumed to be driven by the scalar field called the *inflaton*. Thus cosmic fluid is dominated by the scalar field. Field equations are solved by two main assumptions,

$$\dot{\phi}^2 \ll V(\phi), \qquad |\ddot{\phi}| \ll H |\dot{\phi}|.$$
 (1.7)

These are called the *slow-roll approximations* [54, 55].

<u>1.1.1.4.</u> Cosmological Constant Problem. Vacuum energy or cosmological constant is known as the term which causes accelerated expansion of the universe although it was

initially introduced to obtain static universe. Einstein field equations were written as

$$R_{\mu\nu} - \frac{1}{2}g_{\mu\nu}R + \Lambda g_{\mu\nu} = 8\pi G T_{\mu\nu}, \qquad (1.8)$$

after adding cosmological constant Λ where metric sign is (-, +, +, +). Hence it contributes to the right side of (1.2) as $\rho_{vac} = \frac{\Lambda}{8\pi G}$. By using observational evidence $\Omega_{\Lambda 0} = 0.7 \pm 0.1$ where $\Omega_{\Lambda} = \frac{8\pi G}{3H^2}\rho_{\Lambda}$ and $H_0 = 70 km s^{-1} Mpc^{-1}$, we found $\rho_{\Lambda} \approx 10^{-29} g/cm^3 \approx 3 \times 10^{-47} GeV^4$ which is consistent with [56].

In quantum field theory the vacuum corresponds to the lowest energy state. Spontaneous virtual particle creation and annihilation occurs in this empty space. These quantum fluctuations contribute to the total energy density of the vacuum. Zero-point energies of the field with mass m is; $\frac{\hbar w}{2}$ and $w = \sqrt{m^2 + k^2}$ where k is the wave number. Summation of all modes with wave number cutoff gives $\rho_{vac} \sim 10^{72} GeV^4$ [20,56]. There is 119 order of magnitude difference between theoretical prediction and observational results. If one is gudied by calculations in quantum chromodynamics which indicate that $\rho_{vac} \sim 10^{-6} GeV^4$ [20] the gap is at the order of magnitude 41.

1.1.2. Dark Matter and Mond Theory

According to Keplerian behavior, rotation speed of stars at a radius r in a galaxy is found as

$$v(r) = \sqrt{\frac{Gm(r)}{r}},\tag{1.9}$$

where m(r) is the total mass contained in a volume with radius r. Since luminous mass is concentrated at the center of the galaxy we expect to see decrease in velocity of the rotating object. However according to observational data velocities of the stars increase as their distance from the center increase. Thus it has been concluded that there must be nonluminous mass such that as r increases m(r) increases [57]. This missing matter is known as the dark matter. MOND (Modified Newtonian Dynamics) was established as an alternative to hidden mass [58]. In this theory instead of a Newtonian force acting on a mass $m, \vec{F} = m\vec{a}$ Milgrom introduced

$$\vec{F} = m\mu(\frac{a}{a_0})\vec{a},\tag{1.10}$$

$$\mu(x \gg 1) \approx 1, \qquad \mu(x \ll 1) \approx x, \tag{1.11}$$

where $a_0 \approx 2 \times 10^{-8} cm s^{-2}$. After performing some fundamental physics one shows that $v = (GMa_0)^{1/4}$ for distant stars of galaxies where accelerations of stars are very small. Although MOND was successful when fitting rotation curves of galaxies, none of the relativistic versions of MOND (RMOND) were compatible with data when CMB and MPS (matter power spectra) were computed. However in 2021 the first RMOND theory which achieved to yield CMB and MPS in agreement with data was established [59]. In addition in 2020 EFE (external field effect) in MOND which was proposed as an alternative to dark matter was investigated and observational evidence was found [60].

1.1.3. Cosmic Domain Walls

In a grand unified theory [61] strong, weak and electromagnetic forces are combined into a single force at extremely high temperatures. As the universe cools phase transitions occur. As a result symmetries between interactions are broken. During this phenomena topological defects may occur [62]. For example the following Lagrangian

$$\mathcal{L} = \frac{1}{2} (\partial_{\mu} \phi)^2 - V(\phi), \qquad (1.12)$$

$$V(\phi) = -\frac{1}{2}m^2\phi^2 + \frac{1}{4}\lambda\phi^4,$$
(1.13)

is invariant under reflection $(\phi \to -\phi)$ symmetry. The potential has two stable minima at $\phi = \pm \sqrt{\frac{m^2}{\lambda}}$, since

$$V(\pm\sqrt{\frac{m^2}{\lambda}}) = -\frac{m^4}{4\lambda}, \qquad V''(\pm\sqrt{\frac{m^2}{\lambda}}) = 2m^2.$$
(1.14)

The case $\phi = 0$ corresponds to an unstable extremum since V''(0) < 0. One can add a constant term to the potential without affecting equations of motion. Thus we can write

$$V(\phi) = \frac{\lambda}{4} (\phi^2 - \eta^2)^2, \qquad \eta^2 = \frac{m^2}{\lambda}.$$
 (1.15)

Hence vacuum energy becomes zero. If we choose one of the vacuum state and set $\phi = \eta + \psi$ our Lagrangian is written as

$$\mathcal{L} = \frac{1}{2} (\partial_{\mu} \psi)^2 - \frac{1}{2} m_{\psi}^2 \psi^2 - \sqrt{\frac{\lambda}{2}} m_{\psi} \psi^3 - \frac{1}{4} \lambda \psi^4, \qquad (1.16)$$

where $m_{\psi} = \sqrt{2}m$. Then the symmetry of the Lagrangian is spontaneously broken. Now space has two ground states. One can visualize a universe consisting of different parts; one has $\phi_{ground} = +\eta$, other part has $\phi_{ground} = -\eta$. At the smooth transition of ϕ from one part to other part there is a region where $\phi = 0$ and $V(\phi) \neq 0$. Domain walls are interpreted as this interpolation region. They are two-dimensional defects while cosmic-strings are one-dimensional defects and magnetic monopoles are zerodimensional defects [63–65]. To include them in cosmology is very appealing because they appear in a field theory which has spontaneously broken discrete symmetries [65].

Domain walls differentiate among various candidates for dark energy. They supply the required accelerated expansion with negative pressure $p = -(2/3)\rho$. Altough cosmic fluids with a negative equation of state have an imaginary sound speed there have been several studies indicating that cosmic walls are not ruled out in cosmology [66–71].

1.1.4. Field Equations

In the last decades cosmologists established several modified theories of gravity to investigate all these concepts [55, 72–80]. All of these theories have different field equations which govern the dynamics of the universe. Thus all these theories have different physical conclusions.

In this thesis we study the most general cosmological model with scalar field which is minimally coupled to gravity and the Brans-Dicke cosmology where there exists non-minimal coupling to gravity.

The field equations which govern the universe are ordinary differential equations. To be able to solve them many different approaches have been developed. For the case of minimal coupling we can briefly sumarize these studies as follows. One of them is the dynamical systems methods in which stabiliy analysis of systems of nonlinear differential equations are investigated. Detailed studies have been performed by this method in [81–84]. Other methods are based on assumptions or approximations. The "slow-roll approximation" is the most common one which is applied in scenarios of the inflationary universe [85, 86]. Lastly, the generating function method is proposed as a method which gives exact solution of the field equations in [87]. Some of the searches for exact solutions of the field equations where the scalar field is minimally cooupled to gravity have been presented in [88–91].

In recent years searching for exact solutions of the field equations of scalar tensor theories also has taken more attention [89–94]. In addition in 1990 it has been shown that there are potentials $V(\phi)$ leading to desired behaviour for the scale factor without the use of 'slow-roll approximation' usually assumed in inflationary models [95].

Up to now, for Brans-Dicke cosmology some exact solutions for specific cases have been presented. While for a radiation filled universe exact solution was given in [96], parametric solution for the case $p = \epsilon \rho$ was given in [97]. For Bianchi type universes with arbitrary barotropic perfect fluid exact solution were found [98]. Parametric exact solutions for potential free universe filled with ordinary matter were given in [99]. On the other hand Noether's theorem was used to determine conservation laws in Brans-Dicke cosmology which includes two scalar fields and exact solutions were found [100]. An exact solution of modified Brans-Dicke theory was studied in [101].

We have three main goals in our cosmological studies. The first is solving field equations analytically. The second is finding mathematical machinery which causes to turn on and turn of accelerated expansion in early universe. Last is explaining late time accelerated expansion without dark energy. We study the most general cosmological model with scalar field which is minimally coupled to gravity in the second chapter. Brans-Dicke cosmology is investigated in the chapter 3 and its complementary part is given in chapter 4.

1.2. Deformed Discrete and Finite Quantum Systems

The conventional formulation [102] of quantum mechanics starts with a position space which is the set of real numbers. This leads to physical states which are vectors of a separable Hilbert space. The observables are then described by Hermitian operators on this Hilbert space. Angular momentum and spin one half are the well known systems in which observables have finite and discrete values. Furthermore, quantum information theory, quantum optics and the lattice models are all realized in a finite dimensional Hilbert space. One can find a guide to the literature on the applications in Vourdas' work [103], where quantum systems with finite Hilbert space are considered and phase-space methods are discussed.

Our motivation is to construct new operators so that one can calculate the energy spectrum. In chapter 5 we define a deformed momentum operator and nonunitary translation operators. Eigenvalues and eigenfunctions of the Hamiltonian for periodic and nonperiodic cases are calculated. On the other hand quantum mechanics on a finite periodic lattice is a well known subject which has been studied repeatedly since Schwinger's famous 1960 paper [104]. He developed the generators of a complete unitary operator basis. Applications of Schwinger approach have been used in quantum optics, quantum communications, quantum probability and Galois quantum systems [105–115]. In addition one can find the review of the literature on quantum systems with finite Hilbert space and the link between this theory and the other research fields in Vourdas [116].

In chapter 6 we recognize that when the lattice is not periodic the end points give rise to almost unitary Schwinger operator. We construct the projection operators in terms of the almost unitary translation operators and in terms of the unitary Schwinger operators. Since projection operators play the key role in relations between these two algebras we investigate their properties. Then we are able to write each algebra in terms of the other one. We also find two new representations where the standard basis of $M_N(C)$ is constructed in terms of the projection operators in each algebra. Finally, we formulate an algebra which is related to representing a multi-dimensional lattice in terms of one-dimensional lattices in each direction.

2. ANALYTIC SOLUTIONS OF SCALAR FIELD COSMOLOGY WITH MINIMAL COUPLING

In this chapter we have three main purposes. The first is solving field equations analytically. The second is finding a mathematical machinery which causes to turn on and to turn of accelerated expansion in early universe. Last is explaining late time accelerated expansion without dark energy. In Section 2.1 we use a mathematical tool which is a change of independent variable. Thus the field equations are converted to a new set of differential equations. In Section 2.2 we exactly solve this new set of equations and present solutions in four different forms. In Section 2.3 we investigate single-component universes. In Section 2.4 we examine two-component universes and we find a new exotic matter which causes mathematical mechanisms which turns on and turns off accelerated expansion in an early universe. In Section 2.5 we show that a universe which contains matter and cosmic walls results in accelerated expansion. Furthermore we compare our results with supernova Ia data. Results are quite satisfactory. Then we examine dark energy dominated universe with the same procedure. Our discussion is given in Section 2.6.

2.1. Field Equations

2.1.1. Original Form

Action of general relativity with scalar field and the cosmological constant is given by

$$S = \int d^4x \sqrt{-g} \left[\frac{1}{2\kappa} (R - 2\Lambda) - \frac{1}{2} g^{\mu\nu} \partial_\mu \phi \partial_\nu \phi - V(\phi) + \mathcal{L}_M\right], \qquad (2.1)$$

where R is the Ricci scalar and $\kappa = 8\pi Gc^{-4}$. We will use FLWR metric with space dominant metric sign (-, +, +, +) and units with $\hbar = 1, c = 1$

$$ds^{2} = -dt^{2} + a^{2}(t)\left[\frac{dr^{2}}{1 - k\frac{r^{2}}{L^{2}}} + r^{2}(d\theta^{2} + \sin^{2}\theta d\varphi^{2})\right],$$
(2.2)

where k = -1, 0, 1 and $a(t) = \frac{R(t)}{R(t_0)}$ is normalized scale factor, with the convention $a(t_0) = 1, L = R(t_0)$ and r has dimension of lenght.

Energy-momentum tensor for the field is defined as

$$T_{\mu\nu} = \frac{-2}{\sqrt{-g}} \frac{\delta S_{\phi}}{\delta g^{\mu\nu}},\tag{2.3}$$

where $T^{\mu\nu} = \{\rho, p, p, p\}.$

In standard cosmology field equations have been found as

$$\frac{\dot{a}^2}{a^2} = \frac{8\pi G}{3}\rho + \frac{\Lambda}{3} - \frac{k}{L^2 a^2},$$
(2.4)

$$\frac{\ddot{a}}{a} = -\frac{4\pi G}{3}(\rho + 3p) + \frac{\Lambda}{3},$$
 (2.5)

where $\rho = \rho_{ord} + \rho_{\phi}$ and $p = p_{ord} + p_{\phi}$ are known as Einstein equations. ord stands for ordinary and represents matter-energy distribution placed in Einstein equations by hand as a function the scale size of the universe.

In addition, variation of the Lagrangian density with respect to the field ϕ gives another equation

$$\ddot{\phi} + 3H\dot{\phi} + \frac{dV}{d\phi} = 0.$$
(2.6)

For scalar field dominated universe we have

$$\rho = \frac{\dot{\phi}^2}{2} + V(\phi), \qquad (2.7)$$

$$p = \frac{\dot{\phi}^2}{2} - V(\phi),$$
 (2.8)

where we assume that ϕ is a function only of t. Then Equation (2.6) is equivalent to continuity equation which is given by

$$\dot{\rho} + 3\frac{\dot{a}}{a}(\rho + p) = 0.$$
 (2.9)

For the details of the continuity equation and evolution of cosmological fluid see the Appendix A.8.

We name our potential as an effective potential in the sense that it may reflect another more basic physical theory. We will call Equation (2.4) as the first Einstein equation, Equation (2.5) as the second Einstein equation and the Equation (2.6) as the ϕ equation.

2.1.2. New Form of Field Equations

If independent variable does not appear explicitly in the differential equation one can define a new variable in terms of dependent variables so that the order of the differential equation can be reduced by one [117]. In our field equations independent variable is "t". We define our new dependent variable as Hubble function

$$H(a) = \frac{\dot{a}}{a}.\tag{2.10}$$

Thus our new independent variable becomes the scale factor a. For this reason we write all other variables in terms of the new variable;

$$\phi = \phi(a)$$
 and $V(\phi) = V(a)$. (2.11)

Expressions for derivatives of a and ϕ with respect to time in terms of derivatives with respect to new independent variable a are given in Appendix A.3. In addition Λ can be included in $V(\phi)$ so we will not carry it anymore.

One can easily write the field equations, energy density of the scalar field and the pressure of the scalar field as

$$H^{2} = \frac{8\pi G}{3}\rho - \frac{k}{L^{2}a^{2}},$$
 (2.12)

$$H'Ha + H^2 = -\frac{4\pi G}{3}(\rho + 3p), \qquad (2.13)$$

$$\phi^{''}a^{2}H^{2} + 4\phi^{'}aH^{2} + \phi^{'}a^{2}HH^{'} + V^{'}\frac{1}{\phi^{'}} = 0, \qquad (2.14)$$

$$\rho(a) = \frac{1}{2} (\phi' a H)^2 + V(a), \qquad (2.15)$$

$$p(a) = \frac{1}{2}(\phi' a H)^2 - V(a), \qquad (2.16)$$

where prime denotes $\frac{d}{da}$.

When this set of differential equations is solved exactly we will obtain all unknown functions; H(a), $\phi(a)$, V(a), $\rho(a)$, p(a) and the deceleration parameter q(a) as a function of scale factor, a. Thus it will be possible to track the dynamical history of the universe backward and forward in time. Indeed in some cases it will be possible to formulate some of these functions as a function of time.

2.2. Solution for Field Equations

When solving this differential equation set one should be careful. By taking time derivative of first Einstein equation and using the continuity equation we reach the second Einstein equation. Thus by taking time derivative of first Einstein equation and using the second Einstein equation we reach the continuity equation. In addition by substitution $\rho(t)$ and and p(t) in the continuity equation one can reach ϕ equation. One of the field equations can be derivable from other two of them. One can combine these 3 equations in 3 different pairs such that when their solutions are plugged in the remaining differential equation it will be satisfied automatically.

First combination is the easiest one. We take the first Einstein equation and the ϕ equation. Then we multiply the ϕ equation by ϕ' and obtain

$$a^{2}\phi'\phi''H^{2} + 4a\phi'^{2}H^{2} + a^{2}\phi'^{2}HH' + V'(a) = 0.$$
(2.17)

We define

$$\gamma(a) = \frac{\phi^{\prime 2} H^2 a^2}{2},\tag{2.18}$$

to be able to solve the last differential equation. Therefore this equation is converted to

$$\gamma' + \frac{6}{a}\gamma = -V'. \tag{2.19}$$

This is a first order linear differential equation and it's solution can be found easily as

$$\gamma(a) = \frac{1}{a^6} \left[\int_{a_{in}}^a (-a^{'6}V^{'}(a^{'})) da^{'} + a^6_{in}\gamma(a_{in}) \right].$$
(2.20)

Hence by rewriting Equation (2.18) we obtain

$$\frac{\phi^{'2}H^{2}a^{2}}{2} = \gamma(a),$$

$$\phi^{'2}H^{2}a^{2} = \frac{2}{a^{6}} [\int_{a_{in}}^{a} (-a^{'6}V^{'}(a^{'}))da^{'} + a_{in}^{6}\gamma(a_{in})].$$
(2.21)

It is apparent that to be able to solve this field equation one needs the knowledge of one of the following functions; V(a), H(a), $\phi(a)$. There is one more function which can be used as a starting point of calculations. This is the energy density. The relation between $\rho(a)$ and Equation (2.21) will be studied in Section 2.2.4.

We have gone further by plugging energy density into the first Einstein equation

$$H^{2} = \frac{8\pi G}{3} \left[\frac{1}{2}\phi^{\prime 2}H^{2}a^{2} + V(a)\right] - \frac{k}{L^{2}a^{2}},$$
(2.22)

$$H^{2} = \frac{\frac{8\pi G}{3}V(a) - \frac{k}{L^{2}a^{2}}}{1 - \frac{4\pi G}{3}\phi'^{2}a^{2}}.$$
(2.23)

We will refer the last equation as our Friedmann equation.

2.2.1. Solution for Given V(a)

In this section we start our calculations by using our Friedmann equation. Substituting (2.23) in (2.21) we obtain

$$\phi^{'2}\left[\frac{\frac{8\pi G}{3}V(a) - \frac{k}{L^2 a^2}}{1 - \frac{4\pi G}{3}\phi^{'2}a^2}\right]a^2 = 2\gamma(a).$$
(2.24)

Then one can reach the following results

$$\phi^{'2} = \frac{2\gamma(a)}{a^2 \left[\frac{8\pi G}{3} (V(a) + \gamma(a)) - \frac{k}{L^2 a^2}\right]},$$
(2.25)

$$\phi(a) = \pm \int_{a_{in}}^{a} \sqrt{\frac{2\gamma(a')}{a'^2 [\frac{8\pi G}{3} (V(a') + \gamma(a')) - \frac{k}{L^2 a'^2}}} da' + \phi_{a_{in}}.$$
 (2.26)

One should decide to pick one of the \pm sign in front of the right side of $\phi(a)$ such that the value of the scalar field increase or decrease as the universe expands. H(a) has been found by using the formula of the scalar field in our Friedmann equation as

$$H(a) = \sqrt{\frac{8\pi G}{3}(V(a) + \gamma(a)) - \frac{k}{L^2 a^2}},$$
(2.27)

where

$$\gamma(a) = \frac{1}{a^6} \left[\int_{a_{in}}^a (-a'^6 V'(a')) da' + a^6_{in} \gamma(a_{in}) \right].$$
(2.28)

It is apparent that knowledge of the potential energy function V(a) is sufficient to formulate the scalar field $\phi(a)$ and the Hubble function H(a) as an exact solution of the field equations.

We would like to mention that the results of this subsection are similar to results of [118]. They have reduced the differential equations to quadrature problems by writing V(a) in a complicated way. Then exponential potentials and hyperbolic potentials were focused in their examples.

2.2.2. Solution for Given $\phi(a)$

In some cases one may need to solve the field equations for a specific scalar field. In this calculation V(a) becomes the unknown dependent variable in (2.24). To be able to go further first we write the solution of the ϕ equation as

$$\gamma(a) = -V(a) + \frac{1}{a^6} \left[6 \int_{a_{in}}^a V(a') a'^5 da' + a_{in}^6 \tilde{\gamma}(a_{in}) \right], \quad \tilde{\gamma}_{(a_{in})} = \gamma(a_{in}) + V(a_{in}),$$
(2.29)

where we have applied integration by parts to Equation (2.20). Details of this calculation are given in the Appendix A.4.

<u>2.2.2.1. Singular Case.</u> Firstly we will investigate the special form of the scalar field which causes this singularity in the denominator of right side of the equation. From (2.23) we have

$$1 - \frac{4\pi G}{3}\phi^{'2}a^2 = 0. (2.30)$$

Therefore

$$\int_{\phi_{in}}^{\phi} d\phi' = \pm \sqrt{\frac{3}{4\pi G}} \int_{a_{in}}^{a} \frac{da'}{a'},$$
(2.31)

and

$$\phi = \begin{cases} \sqrt{\frac{3}{4\pi G}} ln(\frac{a}{a_{in}}) + \phi_{in}, \\ -\sqrt{\frac{3}{4\pi G}} ln(\frac{a}{a_{in}}) + \phi_{in}. \end{cases}$$
(2.32)

Since $a_{in} \leq a$, " + " sign indicates that the scalar field always increases. On the other hand minus sign implies that one will have positive and decreasing scalar field when ϕ_{in} big enough.

One easily obtains the potential by plugging the field into (2.22)

$$V(a) = \frac{3}{8\pi G} \left(\frac{k}{L^2 a^2}\right).$$
 (2.33)

Then the Hubble function is formulated just by substitution of $\phi(a)$ and V(a) into the solution of the ϕ equation which is given by the (2.21)

$$H^{2}(a) = \frac{2}{a^{6}} \{ \int_{a_{in}}^{a} \frac{k}{L^{2}} a^{'3} da^{'} + \frac{4\pi G}{3} a_{in}^{6} \gamma(a_{in}) \},$$
(2.34)

$$H(a) = \sqrt{\frac{k}{2L^2 a^2} + \frac{8\pi G}{3} \frac{a_{in}^6 \tilde{\gamma}(a_{in})}{a^6}}, \qquad \gamma(a_{in}) = \frac{3k}{16\pi G L^2 a_{in}^2} + \tilde{\gamma}(a_{in}). \tag{2.35}$$

At first sight one can say that the spatially flat universe is static by choosing $\tilde{\gamma}(a_{in}) = 0$. However this statement is incorrect because it is incomplete. Firstly we would like to remind that our choice at the change of variable $\frac{\dot{a}}{a} = H(a)$ works only for the dynamic universes where $H(a) \neq 0$. Secondly by using the first Einstein equation one can easily deduce that the static and spatially flat universe must be empty. Therefore complete and correct interpretation says that the spatially flat and dynamic universes have time varying energy density.

Considering the solution for k = 0, it is seen from the Equation (2.33), V = 0 for spatially flat universe. Thus we jump back to Equation (2.19) and it turns to

$$\gamma' + \frac{6}{a}\gamma = 0, \qquad (2.36)$$
$$\int_{\gamma_{in}}^{\gamma} \frac{d\gamma'}{\gamma'} = -6 \int_{a_{in}}^{a} \frac{da'}{a'},$$

and

$$ln\gamma - ln\gamma_{in} = -6(lna - lna_{in}),$$

$$\gamma(a) = \frac{\tilde{\gamma}(a_{in})}{a^6} \quad \text{and} \quad \tilde{\gamma}(a_{in}) = \gamma(a_{in})a_{in}^6.$$
(2.37)

Hence by plugging γ and ϕ in (2.18) we have obtain the Hubble function as

$$H = \sqrt{\frac{8\pi G}{3} \frac{\tilde{\gamma}(a_{in})}{a^6}}.$$
(2.38)

<u>2.2.2.2. Non-Singular Case</u>. In this case we investigate general form of the scalar field where $\phi(a) \neq \sqrt{\frac{3}{4\pi G}} ln(a)$. We have start this case by using (2.29) in (2.25)

$$\phi^{\prime 2} = \frac{2\{-V(a) + \frac{1}{a^6} [6 \int_{a_{in}}^a V(a') a'^5 da' + a_{in}^6 \tilde{\gamma}(a_{in})]\}}{a^2\{\frac{8\pi G}{3a^6} [6 \int_{a_{in}}^a V(a') a'^5 da' + a_{in}^6 \tilde{\gamma}(a_{in})] - \frac{k}{L^2 a^2}\}}.$$
(2.39)

To be able to calculate the potential V(a), one should define a new function

$$\alpha(a) = 6 \int_{a_{in}}^{a} V(a') a'^{5} da' + a_{in}^{6} \tilde{\gamma}(a_{in}), \qquad V(a) = \frac{\alpha'}{6a^{5}}, \tag{2.40}$$

where $\alpha(a_{in}) = a_{in}^6 \tilde{\gamma}_{in}$. Then (2.39) turns into a first order linear differential equation which is obtained as

$$\alpha' + (8\pi Ga\phi'^2 - \frac{6}{a})\alpha = (\frac{3ka^5}{L^2})\phi'^2.$$
(2.41)

The solution is found as

$$\alpha(a) = exp\left[\int_{a_{in}}^{a} \left(\frac{6}{a'} - 8\pi G a' \phi'^{2}\right) da'\right] \\ \times \left\{\alpha(a_{in}) + \int_{a_{in}}^{a} exp\left[\int_{a_{in}}^{a'} \left(-\frac{6}{a''} + 8\pi G a'' \phi'^{2}\right) da''\right] \left(\frac{3ka'^{5} \phi'^{2}}{L^{2}}\right) da'\right\}.$$
(2.42)

Then according to relation (2.40) the potential V(a) is found as

$$V(a) = \frac{\alpha'}{6a^5},$$

$$V(a) = (1 - \frac{4\pi G}{3}a^2\phi'^2)e^{\lambda(a)}\beta(a) + \frac{k\phi'^2}{2L^2},$$
(2.43)

$$\lambda(a) = -\int_{a_{in}}^{a} 8\pi G a' \phi'^2 da', \qquad (2.44)$$

$$\beta(a) = \alpha(a_{in}) + \int_{a_{in}}^{a} e^{-\lambda(a')} \frac{3ka_{in}^{6}\phi'^{2}}{L^{2}a'} da'].$$
(2.45)

H(a) has been found by substituting this potential and the specific scalar field into our Friedmann equation

$$H^{2} = \frac{\frac{8\pi G}{3}V(a) - \frac{k}{L^{2}a^{2}}}{1 - \frac{4\pi G}{3}\phi'^{2}a^{2}},$$
$$H(a) = \sqrt{\frac{8\pi G}{3a_{in}^{6}}}e^{\lambda(a)}\beta(a) - \frac{k}{L^{2}a^{2}},$$
(2.46)

where $\lambda(a)$ and $\beta(a)$ are given by (2.44) and (2.45). Hence for a specific scalar field exact solution of the field equations are given by the last two equations.

Both these two cases have a common physical result. If there is only scalar field without any kinds of matter except the dark energy this universe has dynamic behaviour as a result of it being curved by the scalar field.

For k = 0 the solution changes. Equation (2.41) is solved as

$$\alpha' = (\frac{6}{a} - 8\pi G a \phi'^{2}) \alpha, \qquad (2.47)$$

$$\int_{\alpha_{in}}^{\alpha} \frac{d\alpha'}{\alpha'} = \int_{a_{in}}^{a} (\frac{6}{a'} - 8\pi G a' \phi'^{2}(a')) da', \qquad (2.47)$$

$$ln\alpha - ln\alpha_{in} = (lna^{6} - lna_{in}^{6}) - \int_{a_{in}}^{a} 8\pi G a' \phi'^{2}(a') da', \qquad (2.47)$$

$$\frac{\alpha}{\alpha_{in}} = \frac{a^{6}}{a_{in}^{6}} exp[-\int_{a_{in}}^{a} 8\pi G a' \phi'^{2}(a') da'], \qquad (2.48)$$

Then we formulate the potential by using Equation (2.40)

$$V(a) = \frac{\alpha'}{6a^5},$$

$$V(a) = \frac{\alpha_{in}}{a_{in}^6} [1 - \frac{4\pi G}{3} a^2 \phi'^2(a)] e^{\lambda(a)},$$
(2.49)

where $\lambda(a)$ is given by Equation (2.44). The Hubble function is found just by substuting the potential and the scalar field into our Friedmann equation

$$H(a) = \sqrt{\frac{8\pi G}{3} \frac{\alpha_{in}}{a_{in}^6}} e^{\lambda(a)}, \qquad (2.50)$$

where $\lambda(a)$ is given in (2.44).

2.2.3. Solution for Given H(a)

In this section we will start our calculations by rewriting our Friedmann equation in the form

$$H^{2} = \frac{8\pi G}{3} \left(\frac{1}{2}\phi'^{2}H^{2}a^{2} + V(a)\right) - \frac{k}{L^{2}a^{2}}.$$

This equation is easily converted to

$$\frac{\phi^{\prime 2}a^2H^2}{2} = \frac{3}{8\pi G}(H^2 + \frac{k}{L^2a^2}) - V(a).$$
(2.51)

Therefore one can recognize the first term on the left side of Equation (2.51) as $\gamma(a)$ which is the variable found as a solution of the ϕ equation at the beginning of the Section 2.2. Hence (2.51) turns into the following form

$$\gamma(a) = \frac{3}{8\pi G} \left(H^2 + \frac{k}{L^2 a^2}\right) - V(a).$$
(2.52)

By using the last form of $\gamma(a)$ which is formulated in (2.29)

$$\gamma(a) = -V(a) + \frac{1}{a^{6}} \left[6 \int_{a_{in}}^{a} V(a') a'^{5} da' + a_{in}^{6} \tilde{\gamma}(a_{in}) \right].$$
So we have obtained

$$-V(a) + \frac{1}{a^6} \left[6 \int_{a_{in}}^a V(a') a'^5 da' + a_{in}^6 \tilde{\gamma}(a_{in}) \right] = \left[\frac{3}{8\pi G} (H^2 + \frac{k}{L^2 a^2}) - V(a) \right].$$
(2.53)

As we have done in the previous section we now find the potential energy. The last equation can be easily solved so that $\alpha(a)$

$$\alpha(a) = 6 \int_{a_{in}}^{a} V(a') a'^{5} da' + a_{in}^{6} \tilde{\gamma}(a_{in}) \quad \text{so} \quad V(a) = \frac{\alpha'}{6a^{5}}.$$

Therefore $\alpha(a)$ is found algebraically from Equation (2.53) as

$$\alpha(a) = \frac{3a^6}{8\pi G} (H^2 + \frac{k}{L^2 a^2}).$$
(2.54)

As a result the potential energy is calculated as

$$V(a) = \frac{3}{8\pi G} \left[(H^2 + \frac{k}{L^2 a^2}) + \frac{a}{3} (HH' - \frac{k}{L^2 a^3}) \right].$$
(2.55)

The scalar field is found by substituting the potential into Equation (2.51) as

$$\phi(a) = \pm \int_{a_{in}}^{a} \sqrt{\frac{1}{4\pi G a H^2} (-HH' + \frac{k}{L^2 a'^3})} da' + \phi(a_{in}).$$
(2.56)

Therefore last equations can be used to construct the scalar field and the potential for a given Hubble function.

2.2.4. Solution for Given $\rho(a)$

When one starts the calculations with one of the following functions V(a), $\phi(a)$, H(a) one can end up with some unusual forms of the energy density. To avoid this possibility one should start the calculations for desired energy density. It is written in terms of our new independent variable "a" as

$$\rho(a) = \frac{1}{2}(\phi' a H)^2 + V(a).$$

One can recognize the first term on the right side of this equation as $\gamma(a)$. Hence we obtain

$$V(a) = \rho(a) - \gamma(a)$$
 and $\gamma(a) = \frac{(\phi' a H)^2}{2}$. (2.57)

Then we substitute this into the ϕ equation

$$\gamma^{'} + \frac{6}{a}\gamma = -V^{'},$$

and we obtain

$$\gamma(a) = -\frac{a}{6}\rho'. \tag{2.58}$$

By using the definition of $\gamma(a)$ we also obtain

$$\phi^{'2} = \frac{2\gamma(a)}{a^2 H^2}.$$
(2.59)

The Hubble function is known just by inserting the energy density into the original form of the first Einstein equation

$$H(a) = \sqrt{\frac{8\pi G}{3}\rho - \frac{k}{L^2 a^2}}.$$
(2.60)

Therefore the scalar field is formulated as

$$\phi(a) = \pm \int_{a_{in}}^{a} \sqrt{\frac{-\frac{a'}{3}\rho'}{a'^2[\frac{8\pi G}{3}\rho - \frac{k}{L^2a'}]}} da' + \phi(a_{in}).$$
(2.61)

The potential energy is written by substitution of (2.58) into (2.57) as

$$V(a) = \rho(a) + \frac{a}{6}\rho'.$$
 (2.62)

Exact solution of the field equations for a desired energy density are given by the formulas in (2.60-2.62).

2.3. Single Component Universes

We present some general solutions for a universe which has a single component. This purpose is easily achieved for a given $\rho(a)$ in Section 2.3.1 and for a given V(a) in Section 2.3.2. In addition we have performed calculations for both a curved universe and a spatially flat universe. Therefore one can see the effect of curvature term in dynamics of the universe.

2.3.1. General Solution for $\rho(a) = \frac{\rho_n}{a^n}$

To satisfy the weak energy condition $\rho \ge 0$ and $\rho + p \ge 0$ one should start calculations with a given energy density.

<u>2.3.1.1.</u> $k \neq 0$. We begin this subsection by taking the energy density as in the form of perfect fluid

$$\rho = \frac{\rho_n}{a^n}.\tag{2.63}$$

Then by applying the procedure which is explained in Section 2.2.4 we immediately obtain H(a), V(a), $\phi(a)$, p(a), q(a) as

$$H(a) = \sqrt{\frac{8\pi G\rho_n}{3a^n} - \frac{k}{a^2}},$$
(2.64)

$$V(a) = \frac{(6-n)}{6} \frac{\rho_n}{a^n},$$
(2.65)

$$\phi(a) = \pm \sqrt{\frac{n}{8\pi G}} \{ ln(\frac{a}{a_{in}}) + \frac{1}{1 - n/2} ln[\frac{1}{b}(1 + \sqrt{1 - \frac{3ka^{n-2}}{8\pi G\rho_n}})] \} + \phi(a_{in}), \qquad (2.66)$$

$$b = (1 + \sqrt{1 - \frac{3ka_{in}^{n-2}}{8\pi G\rho_n}}),$$

$$(2.67)$$

$$p(a) = \left(\frac{n}{3}\right)\frac{\rho_n}{a^n},$$
(2.67)
$$4(n-2)\pi G\rho_n$$
(2.68)

$$q(a) = \frac{4(n-2)\pi G\rho_n}{8\pi G\rho_n - 3ka^{n-2}}.$$
(2.68)

First, we interpret these formulas generally. When we apply the boundaries on equation of state $\nu = \frac{p}{\rho}$ we obtain

$$-1 \le \nu \le 1, \quad -1 \le \frac{n-3}{3} \le 1, \quad 0 \le n \le 6.$$
 (2.69)

Therefore all exotic fluids with energy density in the form of $\frac{\rho_n}{a^n}$ have $0 \le n \le 6$. This condition also makes the potential non-negative. Then special cases pops up immediately for n = 0, 2, 6. Furthermore we would like to add one more comment. The second term on the right side of (2.66) is always real. This is easily recognized when one writes the related components in terms of cosmological density parameters.

Case n = 0 corresponds to constant energy density $\rho = \rho_0$. One presents related functions for more comments,

$$H(a) = \sqrt{\frac{8\pi G\rho_0}{3} - \frac{k}{a^2}},$$
(2.70)

$$V(a) = \rho_0, \tag{2.71}$$

$$\phi(a) = 0, \tag{2.72}$$

$$p(a) = -\rho_0, \tag{2.73}$$

$$q(a) = -1 + \frac{3k}{3k - 8\pi G\rho_0 a^2}.$$
(2.74)

Furthermore a(t) can be formulated by the following steps

$$\int_{0}^{t} dt' = \int_{a_{in}}^{a} \frac{da'}{a' H(a')},$$
(2.75)

$$a(t) = \left(\frac{a_{in}}{2} + \frac{\sqrt{\mu^2 a_{in}^2 - k}}{2\mu}\right)e^{\mu t} + \left(\frac{a_{in}}{2} - \frac{\sqrt{\mu^2 a_{in}^2 - k}}{2\mu}\right)e^{-\mu t}, \quad \mu = \sqrt{\frac{8\pi G\rho_0}{3}}.$$
 (2.76)

Since the scalar field is zero in this case, our solutions reduce to solutions of standard cosmology with dark energy. According to results given in (2.70) and (2.76), to have a dynamic universe with real Hubble function and real scale factor cosmological constant must be big enough to overcome the smallness of the universe;

$$\frac{8\pi G\rho_0}{3} > \frac{k}{a_{in}^2}.$$
(2.77)

This is the mathematical reason which explains the big value of the cosmological constant in the early universe according to the standard model.

Case n = 2 creates a singularity in the scalar field as seen in (2.66). Thus we have calculated $\phi(a)$ separately and we have found it as

$$\phi(a) = \pm \sqrt{\frac{2\rho_2}{8\pi G\rho_2 - 3k}} ln(\frac{a}{a_{in}}) + \phi(a_{in}).$$
(2.78)

Furthermore a(t) is

$$a(t) = \tau t + a_{in}, \qquad \tau = \sqrt{\frac{8\pi G\rho_2}{3} - k},$$
 (2.79)

which is consistent with (2.68) which tells us that for n = 2, the universe expands with constant speed.

Case n = 6 requires special attention. Potential becomes V = 0 and equation of state becomes $\nu = \frac{p}{\rho} = 1$. This case corresponds to a massless scalar field.

<u>2.3.1.2.</u> k = 0. When we study the spatially flat universe, nature of the scalar field changes. As a result of this change we can formulate the potential as a function of the scalar field. The field is written as

$$\phi(a) = \pm \sqrt{\frac{n}{8\pi G}} ln(\frac{a}{a_{in}}) + \phi(a_{in}). \qquad (2.80)$$

Therefore one can formulate the scale factor and hence the potential as a function of the scalar field as

$$a = a_{in} exp[\pm \sqrt{\frac{8\pi G}{n}}(\phi - \phi(a_{in}))], \qquad (2.81)$$

$$V(a) = \left(\frac{6-n}{6}\right) \frac{\rho_n}{a_{in}^n} exp[\mp \sqrt{8\pi nG}(\phi - \phi(a_{in}))].$$
(2.82)

Furthermore we can find a(t) by

$$\int_{0}^{t} dt' = \int_{a_{in}}^{a} \frac{da'}{a'H(a')},$$
$$a(t) = (n\sqrt{\frac{2\pi G\rho_n}{3}}t + a_{in}^{n/2})^{2/n},$$
(2.83)

where there is a singularity in the case n = 0. This case corresponds to the standard model with cosmological constant. Hence

$$\int_{0}^{t} dt' = \int_{a_{in}}^{a} \frac{da'}{a'H(a')},$$

$$a(t) = a_{in}e^{\mu t}, \qquad \mu = \sqrt{\frac{8\pi G\rho_{0}}{3}}.$$
 (2.84)

As it is seen there is no constraint on this constant energy density. It can be big as well as it can be small.

2.3.2. General Solution for $V(a) = \frac{V_n}{a^n}$

To start calculations with given V(a) is more fundamental. After getting intuition about the form of the potential which is required for the perfect fluid, we continue our work by choosing the potential. Results are important because they are surprisingly different than Section 2.3.1.

<u>2.3.2.1.</u> $k \neq 0$. We have plugged in $V(a) = \frac{V_n}{a^n}$ in the formulas given by (2.26-2.28) and we have obtained

$$\gamma(a) = \frac{1}{a^6} \left[\frac{nV_n}{6-n} a^{6-n} + a^6_{in} \tilde{\gamma}(a_{in}) \right], \quad \tilde{\gamma}(a_{in}) = -\frac{nV_n}{6-n} a^{-n}_{in} + \gamma(a_{in}), \quad (2.85)$$

$$H(a) = \sqrt{\frac{16\pi G V_n}{(6-n)a^n} + \frac{8\pi G a_{in}^6 \tilde{\gamma}(a_{in})}{3a^6} - \frac{k}{L^2 a^2}},$$
(2.86)

$$\phi(a) = \pm \sqrt{6} \int_{a_{in}}^{a} \sqrt{\frac{nV_n a^{'6} + (6-n)a_{in}^6 \tilde{\gamma}(a_{in})a^{'n}}{48\pi G V_n a^{'8} + (n-6)(3ka^{'4} - 8\pi G a_{in}^6 \tilde{\gamma}(a_{in}))a^{'n+2}}} da^{'} + \phi(a_{in}),$$
(2.87)

$$\rho(a) = \frac{6V_n}{(6-n)a^n} + \frac{a_{in}^6 \tilde{\gamma}(a_{in})}{a^6}, \qquad (2.88)$$

$$p(a) = \frac{2(n-3)V_n}{(6-n)a^n} + \frac{a_{in}^6\tilde{\gamma}(a_{in})}{a^6},$$
(2.89)

$$q(a) = \frac{8\pi G[2(6-n)a_{in}^{6}\tilde{\gamma}(a_{in})a^{n} + 3(n-2)V_{n}a^{6}]}{48\pi GV_{n}a^{6} + (n-6)(3ka^{4} - 8\pi Ga_{in}^{6}\tilde{\gamma}(a_{in}))a^{n}}.$$
(2.90)

For all n there is the term proportional to a^{-6} in energy density. Therefore not only for zero potential but also for each potential, universe contains the stiff fluid.

For case n = 0 one should perform the calculations by starting from Equation (2.19),

$$\int_{\gamma(a_{in})}^{\gamma} \frac{d\gamma}{\gamma} = -\int_{a_{in}}^{a} \frac{6}{a'} da', \qquad (2.91)$$

$$\gamma(a) = \frac{\gamma(a_{in})a_{in}^6}{a^6}.$$
(2.92)

Thus we have

$$H(a) = \sqrt{\frac{8\pi G}{3} \left(\frac{\gamma(a_{in})a_{in}^6}{a^6} + V_0\right) - \frac{k}{L^2 a^2}},$$
(2.93)

$$\phi(a) = \pm \sqrt{6} \int_{a_{in}}^{a} \sqrt{\frac{\gamma(a_{in})a_{in}^{6}}{8\pi G(\gamma(a_{in})a_{in}^{6} + V_{0}a'^{6})a'^{2} - 3ka'^{6}}} da' + \phi(a_{in}), \qquad (2.94)$$

$$\rho(a) = \frac{\gamma(a_{in})a_{in}^6}{a^6} + V_0, \tag{2.95}$$

$$p(a) = \frac{\gamma(a_{in})a_{in}^{6}}{a^{6}} - V_{0},$$
(2.96)

$$q(a) = \frac{8\pi G[2\gamma(a_{in})a_{in}^{6} - V_0 a^{6}]}{8\pi G[\gamma(a_{in})a_{in}^{6} + V_0 a^{6}] - 3ka^4}.$$
(2.97)

In contrast to constant energy density case, constant potential differs from cosmological constant case.

As it is seen from formulas there is a singularity for n = 6. Thus we have investigated this case separately:

$$\gamma(a) = \frac{1}{a^6} [6V_6 ln(\frac{a}{a_{in}}) + a^6_{in} \gamma(a_{in})],$$
(2.98)

$$H(a) = \sqrt{\frac{8\pi G}{3} \left[\frac{V_6 + a_{in}^6 \gamma(a_{in})}{a^6} + \frac{6V_6 ln(\frac{a}{a_{in}})}{a^6}\right] - \frac{k}{L^2 a^2}},$$
(2.99)

$$\phi(a) = \pm \sqrt{6} \int_{a_{in}}^{a} \sqrt{\frac{a_{in}^{6} \gamma(a_{in}) + 6V_{6}ln(\frac{a'}{a_{in}})}{[48\pi GV_{6}ln(\frac{a'}{a_{in}}) + 8\pi G(V_{6} + a_{in}^{6}\gamma(a_{in})) - 3ka'^{4}]a'^{2}}} da' + \phi(a_{in}),$$
(2.100)

$$\rho(a) = \frac{a_{in}^6 \gamma(a_{in}) + V_6}{a^6} + \frac{6V_6 ln(\frac{a}{a_{in}})}{a^6}, \qquad (2.101)$$

$$p(a) = \frac{a_{in}^6 \gamma(a_{in}) - V_6}{a^6} + \frac{6V_6 ln(\frac{a}{a_{in}})}{a^6}, \qquad (2.102)$$

$$q(a) = \frac{8\pi G[2a_{in}^6\gamma(a_{in}) - V_6 + 12V_6ln(\frac{a}{a_{in}})]}{48\pi GV_6ln(\frac{a}{a_{in}}) + 8\pi G(a_{in}^6\gamma(a_{in}) + V_6) - 3ka^4}.$$
(2.103)

30

Energy density and pressure should be written in the following form

$$\rho(a) = \rho_1(a) + \rho_2(a), \qquad (2.104)$$

$$\rho_1(a) = \frac{a_{in}^6 \gamma(a_{in})}{a^6}, \qquad \rho_2(a) = \frac{V_6}{a^6} ln[e(\frac{a}{a_{in}})^6], \qquad (2.105)$$

$$p(a) = p_1(a) + p_2(a), \tag{2.106}$$

$$p_1(a) = \frac{a_{in}^6 \gamma(a_{in})}{a^6}, \qquad p_2(a) = \frac{V_6}{a^6} ln[\frac{1}{e}(\frac{a}{a_{in}})^6].$$
 (2.107)

Thus each component satisfies the continuity equation which is given by (2.9) according to perfect fluid theorem.

Furthermore investigation of equation of state for the second part of the fluid is important. First we write the pressure in the following form:

$$p_2(a) = \frac{V_6}{a^6} ln[\frac{1}{e}(\frac{a_{in}}{a})^6(\frac{a}{a_{in}})^{12}], \qquad (2.108)$$

$$p_2(a) = \frac{V_6}{a^6} \{ ln[\frac{1}{e}(\frac{a_{in}}{a})^6] + ln(\frac{a}{a_{in}})^{12} \}.$$
 (2.109)

Then equation of state turns into

$$\nu_2 = \frac{p_2}{\rho_2},\tag{2.110}$$

$$\nu_2 = \frac{\ln[\frac{1}{e}(\frac{a_{in}}{a})^6] + \ln(\frac{a}{a_{in}})^{12}}{\ln[e(\frac{a}{a_{in}})^6]},$$
(2.111)

$$\nu_2 = -1 + \frac{12ln(\frac{a}{a_{in}})}{1 + 6ln(\frac{a}{a_{in}})}.$$
(2.112)

In addition

$$\lim_{a \to a_{in}} \nu_2 = -1, \qquad \lim_{a \to \infty} \nu_2 = 1.$$
 (2.113)

This phenomenon says that at the beginning of the universe there was a negative pressure. This pressure was huge because it is proportional to $\frac{1}{a^6}$. As the universe expands this pressure and the related energy density becomes negligible since both of them proportional to $\frac{1}{a^6}$.

<u>2.3.2.2. k=0.</u> First simplification occurs in the relation between cosmological time and the scale factor of the universe as

$$t = \int_{a_{in}}^{a} \frac{da'}{a'H(a')},$$

$$t = \int_{a_{in}}^{a} \frac{da'}{a'\sqrt{\frac{8\pi G}{3}(\frac{a_{in}^{6}\tilde{\gamma}(a_{in})}{a'^{6}} + \frac{6v_{n}}{(6-n)a'^{n}})}},$$

$$t = \sqrt{\frac{1}{24\pi Ga_{in}^{6}\tilde{\gamma}(a_{in})}} \left[a'^{3}{}_{2}F_{1}(\frac{1}{2}, \frac{3}{6-n}; \frac{3}{6-n} + 1; -\frac{6v_{n}}{(6-n)a_{in}^{6}\tilde{\gamma}(a_{in})}a'^{(6-n)})\right]_{a_{in}}^{a}.$$
(2.115)

 $_2F_1$ is the hypergeometric function

$${}_{2}F_{1}(1/2,b;b+1;u) = \sum_{n=0}^{\infty} \frac{(1/2)_{n}(b)_{n}}{(b+1)_{n}} \frac{u^{n}}{n!},$$
(2.116)

where $(b)_n$ is the Pochhammer symbol which is defined as

$$(b)_n = \begin{cases} 1 & \text{if } n = 0, \\ b(b+1)(b+2)\dots(b+n-1), & \text{if } n = 1, 2\dots \end{cases}$$
(2.117)

Details of this calculation is given in Appendix A.6 Our condition, n < 6 for positivity of energy density avoids the singularity of the hypergeometric function given in (2.115). One can go further by choosing initial condition $a_{in}^6 \tilde{\gamma}(a_{in}) = 0$. Results are very similar to what we have obtained in Section 2.3.1.2. The scale factor is found as

$$a(t) = \left(2n\sqrt{\frac{\pi G v_n}{(6-n)}}t + a_{in}^{n/2}\right)^{2/n}.$$
(2.118)

Positivity condition n < 6 for energy density makes a(t) real. The scalar field is found as

$$\phi(a) = \pm \sqrt{\frac{n}{8\pi G}} ln(\frac{a}{a_{in}}) + \phi(a_{in}).$$
(2.119)

Then

$$a = a_{in}exp(\pm\mu\psi), \quad \psi = \phi(a) - \phi(a_{in}) \quad \text{and} \quad \mu = 2\sqrt{\frac{2\pi G}{n}}.$$
 (2.120)

Therefore

$$V(\psi) = V_n a_{in}^{-n} exp(\mp n\mu\psi) \quad \text{or } V(\phi) = V_n a_{in}^{-n} exp\{\mp n\mu[\phi(a) - \phi(a_{in})]\}.$$
 (2.121)

In constant potential case where n = 0 one can obtain explicit form of the scalar field as

$$\phi(a) = \mp \frac{1}{2} \sqrt{\frac{1}{3\pi G}} ln \left[\frac{1}{ba^3} (1 + \sqrt{1 + \frac{V_0 a^6}{\gamma(a_{in}) a_{in}^6}}) \right] + \phi(a_{in}), \qquad (2.122)$$
$$b = \frac{1}{a_{in}^3} (1 + \sqrt{1 + \frac{V_0}{\gamma(a_{in})}}).$$

Formulation of a(t) is also possible

$$t = \int_{a_{in}}^{a} \frac{da'}{a'\sqrt{\frac{8\pi G}{3}(V_0 + \frac{\gamma(a_{in})a_{in}^6}{a'^6})}},$$
(2.123)

$$a(t) = \left[\frac{b^2 V_0 e^{2\mu t} - \gamma(a_{in}) a_{in}^6 e^{-2\mu t}}{2bV_0}\right]^{1/3},$$

$$b = a_{in}^3 + \sqrt{a_{in}^6 + \frac{\gamma(a_{in}) a_{in}^6}{V_0}} \quad \text{and} \quad \mu = \sqrt{6\pi GV_0}.$$
(2.124)

This case will be investigated in more detail in Section 2.4.3.

The case n = 6 has two simplification for a spatially flat universe. The scalar field has been found as

$$\phi(a) = \pm \frac{1}{12} \sqrt{\frac{3}{\pi G}} \frac{1}{V_6} I(a) + \phi(a_{in}), \qquad (2.125)$$

$$I(a') = x(a')\sqrt{1 - \frac{V_6}{x(a')}} + \frac{V_6}{2}ln[-\frac{2x(a')}{V_6} + 1 + \frac{2x(a')}{V_6}\sqrt{1 - \frac{V_6}{x(a')}}], \qquad (2.126)$$

$$x(a') = a_{in}^{6} \gamma(a_{in}) + V_{6}[1 + 6ln(\frac{a'}{a_{in}})].$$
(2.127)

Expression of a(t) is

$$a(t) = a_{in} exp \left\{ \frac{1}{6} \left[2 \left(erfi^{-1}(\mu t + b) \right)^2 - 1 - \frac{a_{in}^6 \gamma(a_{in})}{V_6} \right] \right\},$$
(2.128)

$$\mu = \sqrt{\frac{8\pi G}{3}} \frac{1}{a_{in}^3} exp[\frac{1}{2} + \frac{a_{in}^6 \gamma(a_{in})}{2V_6}], \qquad b = erfi[\frac{1}{2} \left(1 + \frac{a_{in}^6 \gamma(a_{in})}{V_6}\right)], \qquad (2.129)$$

$$erfi(\theta) = \frac{2}{\sqrt{\pi}} \int_0^\theta exp(z^2)dz,$$
(2.130)

where $erfi(\theta)$ is the *imaginary error function*. Details of this calculation is given in Appendix A.7. When θ is real $erfi(\theta)$ is real [119]. Therefore it's inverse function $erfi^{-1}(\theta)$ becomes real for real θ .

2.4. Early Epoch of the Universe

There have been many studies which show that the early universe should expand with exponential expansion to be able to reach its size today. Thus our purpose in this section is to explore the mathematical turn on and turn of mechanism to start and to end up exponential expansion. For this reason we have studied three different combinations for curved and spatially flat universes.

2.4.1. Dark Energy

We will search for the universe with energy density in the following form

$$\rho(a) = \frac{\rho_n}{a^n} + \rho_0.$$
 (2.131)

<u>2.4.1.1.</u> $k \neq 0$. Firstly we take nonzero curvature. We have found H(a), V(a), $\phi(a)$, p(a) and q(a) as

$$H(a) = \sqrt{\frac{8\pi G}{3}(\frac{\rho_n}{a^n} + \rho_0) - \frac{k}{a^2}},$$
(2.132)

$$V(a) = \frac{(6-n)}{6} \frac{\rho_n}{a^n} + \rho_0, \qquad (2.133)$$

$$\phi(a) = \pm \int_{a_{in}}^{a} \sqrt{\frac{n\rho_n}{8\pi G(\rho_n a'^2 + \rho_0 a'^{2+n}) - 3ka'^n}} da' + \phi(a_{in}), \qquad (2.134)$$

$$p(a) = \frac{(n-3)}{3} \frac{\rho_n}{a^n} - \rho_0, \qquad (2.135)$$

$$q(a) = \frac{4\pi G[-2\rho_0 a^n + (n-2)\rho_n]a^2}{8\pi G[\rho_n a^2 + \rho_0 a^{2+n}] - 3ka^n}.$$
(2.136)

We continue to investigate the dynamics of the early universe for small a as

$$\lim_{a \to 0} q(a) = \frac{4\pi G[(n-2)\rho_n]a^2}{8\pi G\rho_n a^2 - 3ka^n},$$
(2.137)

$$\lim_{a \to 0} q(a) = \begin{cases} -1 + \frac{n}{2} & \text{if } 2 < n \le 6, \\ 0 & \text{if } 0 \le n \le 2. \end{cases}$$
(2.138)

One should check the roots of q(a). Since the denominator of the $q(a) \sim H^2(a)$ and $H^2(a) > 0$ we are only interested in numerator of q(a).

$$q(a) = 0 \qquad \Rightarrow \qquad a = [\frac{(n-2)\rho_n}{2\rho_0}]^{1/n},$$
 (2.139)

where there is no real and positive root for $n \leq 2$. Hence for $2 < n \leq 6$ the universe starts to expand with deceleration and then expansion turns to acceleration. For $0 < n \leq 2$ the universe starts with constant velocity and it immediately accelerates. In both cases although there are mathematical turn on mechanism to initiate acceleration there is no mathematical turn of mechanism to end acceleration in the this model.

<u>2.4.1.2.</u> k = 0. For k = 0, behaviour of the deceleration parameter changes as follows

$$\lim_{a \to 0} q(a) = -1 + \frac{n}{2} \quad \text{if} \quad 0 \le n \le 6.$$
 (2.140)

Root of q(a) is still given by (2.139). The difference between spatially flat and curved universe occurs just at the beginning of the universe for n < 2. The universe starts to expand with acceleration.

The scalar field can be simplified as

$$\phi(a) = \mp \frac{1}{\sqrt{2\pi Gn}} ln \left[\frac{a^{-n/2} + \sqrt{a^{-n} + \frac{\rho_0}{\rho_n}}}{a_{in}^{-n/2} + \sqrt{a_{in}^{-n} + \frac{\rho_0}{\rho_n}}} \right] + \phi(a_{in}).$$
(2.141)

Then one can derive $V(\phi)$ as follows

$$\frac{a^{-n/2} + \sqrt{a^{-n} + \frac{\rho_0}{\rho_n}}}{a_{in}^{-n/2} + \sqrt{a_{in}^{-n} + \frac{\rho_0}{\rho_n}}} = exp(-\sqrt{n}\psi), \quad \psi = \pm\sqrt{2\pi G}[\phi(a) - \phi(a_{in})].$$
(2.142)

Thus for some specific values of n one obtains

$$V(\psi) = \begin{cases} \rho_0 & \text{if} \quad n = 6, \\ \frac{1}{3} \frac{\rho_4}{a^4(\psi)} + \rho_0 & \text{if} \quad n = 4, \\ \frac{1}{2} \frac{\rho_3}{a^3(\psi)} + \rho_0 & \text{if} \quad n = 3, \end{cases}$$
(2.143)

where

$$a(\psi) = \begin{cases} \sqrt{2} \{ \frac{-a_{in}^2 e^{2\psi} [1 - \sqrt{1 + a_{in}^4 \frac{\rho_0}{\rho_4}} + (1 + \sqrt{1 + a_{in}^4 \frac{\rho_0}{\rho_4}}) e^{4\psi}]}{a_{in}^4 \frac{\rho_0}{\rho_4} (1 + e^{8\psi}) - 2(2 + a_{in}^4 \frac{\rho_0}{\rho_4}) e^{4\psi}}{p_4} \}^{1/2} & \text{if} \quad n = 4, \\ \frac{a_{in}^{3/2} e^{\sqrt{3}\psi} [-1 + \sqrt{1 + a_{in}^3 \frac{\rho_0}{\rho_3}} - (1 + \sqrt{1 + a_{in}^3 \frac{\rho_0}{\rho_3}}) e^{2\sqrt{3}\psi}]}{2^{2/3} \{ \frac{a_{in}^{3/2} \rho_0}{a_{in}^3 \frac{\rho_0}{\rho_3} (1 + e^{4\sqrt{3}\psi}) - 2(2 + a_{in}^3 \frac{\rho_0}{\rho_3}) e^{2\sqrt{3}\psi}} \}^{2/3} & \text{if} \quad n = 3. \end{cases}$$

$$(2.144)$$

Formulation of a(t) is possible for k = 0 as

$$t = \int_{a_{in}}^{a} \frac{da'}{a'H(a')},$$
(2.145)

$$t = \sqrt{\frac{3}{8\pi G\rho_0}} \frac{2}{n} ln \left[\frac{a^{n/2} + \sqrt{a^n + \frac{\rho_n}{\rho_0}}}{a_{in}^{n/2} + \sqrt{a_{in}^n + \frac{\rho_n}{\rho_0}}}\right].$$
 (2.146)

For some specific n

$$a(t) = \left[\frac{(a_{in}^3 + \sqrt{a_{in}^6 + \frac{\rho_n}{\rho_0}})e^{3\mu t} + (a_{in}^3 - \sqrt{a_{in}^6 + \frac{\rho_n}{\rho_0}})e^{-3\mu t}}{2}\right]^{1/3}, \quad \text{if} \quad n = 6, \quad (2.147)$$

$$a(t) = \left[\frac{(a_{in}^2 + \sqrt{a_{in}^4 + \frac{\rho_n}{\rho_0}})e^{2\mu t} + (a_{in}^2 - \sqrt{a_{in}^4 + \frac{\rho_n}{\rho_0}})e^{-2\mu t}}{2}\right]^{1/2}, \quad \text{if} \quad n = 4, \quad (2.148)$$

$$a(t) = \left[\frac{(a_{in}^{3/2} + \sqrt{a_{in}^3 + \frac{\rho_n}{\rho_0}})e^{3\mu t/2} + (a_{in}^{3/2} - \sqrt{a_{in}^3 + \frac{\rho_n}{\rho_0}})e^{-3\mu t/2}}{2}\right]^{2/3}, \quad \text{if} \quad n = 3, \quad (2.149)$$

where $\mu = \sqrt{\frac{8\pi G\rho_0}{3}}$. If one chooses $a_{in} = 0$, for $t \ll 1$ a(t) can be approximately written as

$$a(t) = \begin{cases} 3^{1/3} (\frac{\rho_6}{\rho_0})^{1/6} (\mu t)^{1/3} & \text{if} \quad n = 6, \\ 2^{1/2} (\frac{\rho_4}{\rho_0})^{1/4} (\mu t)^{1/2} & \text{if} \quad n = 4, \\ (\frac{3}{2})^{2/3} (\frac{\rho_3}{\rho_0})^{1/3} (\mu t)^{2/3} & \text{if} \quad n = 3. \end{cases}$$
(2.150)

2.4.2. Cosmic Domain Walls

Cosmic domain walls are known with their contribution to energy density with term $\rho \sim 1/a$. Their equation of state parameter is given by $\nu = -2/3$. Dynamics of the universe with two components where one of them is a domain wall are very similar to dynamics of universes with two components where one of them is dark energy. We have taken the energy density in the following form

$$\rho(a) = \frac{\rho_w}{a} + \frac{\rho_n}{a^n}.$$
(2.151)

<u>2.4.2.1.</u> $k \neq 0$. For a curved space results are found as

$$H(a) = \sqrt{\frac{8\pi G}{3}(\frac{\rho_w}{a} + \frac{\rho_n}{a^n}) - \frac{k}{L^2 a^2}},$$
(2.152)

$$V(a) = \frac{5\rho_w}{6a} + \frac{(6-n)}{6}\frac{\rho_n}{a^n},$$
(2.153)

$$\phi(a) = \pm \int_{a_{in}}^{a} \sqrt{\frac{n\rho_n a' + \rho_w a'^n}{8\pi G[\rho_w a'^{2+n} + \rho_n a'^3] - 3ka'^{1+n}}} da' + \phi(a_{in}), \qquad (2.154)$$

$$p(a) = \frac{-2\rho_w}{3a} + \frac{(n-3)}{3}\frac{\rho_n}{a^n},$$
(2.155)

$$q(a) = \frac{4\pi G[-\rho_w a^n + (n-2)\rho_n a]a}{8\pi G(\rho_w a^n + \rho_n a)a - 3ka^n}.$$
(2.156)

Behaviour of the deceleration parameter at the beginning is obtained as

$$\lim_{a \to 0} q(a) = \begin{cases} -1 + \frac{n}{2} & \text{if } 2 < n \le 6, \\ 0 & \text{if } 0 < n \le 2. \end{cases}$$
(2.157)

Furthermore since denominator of $q(a) \sim H^2(a)$ as stated before, numerator of q(a) determines dynamics of the universe. Roots of the deceleration parameter is found as

$$a = [(n-2)\frac{\rho_n}{\rho_w}]^{1/(n-1)}.$$
(2.158)

Therefore the universe start to expand with deceleration and then turns to accelerate for $2 < n \leq 6$. On the other hand for $0 \leq n \leq 2$ at the beginning of universe its velocity was constant and thus the universe starts to its expansion with acceleration.

<u>2.4.2.2.</u> k = 0. Behaviour of q(a) changes as

$$\lim_{a \to 0} q(a) = \begin{cases} -1 + \frac{n}{2} & \text{if } 1 < n \le 6, \\ -\frac{1}{2} & \text{if } 0 \le n \le 1. \end{cases}$$
(2.159)

Therefore the differences in dynamics of the universe when it is spatially flat is seen when $0 \le n < 1$. In this case at the beginning its velocity is not constant and the universe starts its expansion with acceleration. $\phi(a)$ has been simplified for three cases: for domain walls and stiff fluid as

$$\phi(a) = \pm \frac{1}{5\sqrt{2\pi G}} \left\{ -\sqrt{6} ln \left[\frac{(\sqrt{6}\sqrt{\rho_s + \rho_w a^5} + \sqrt{6\rho_s + \rho_w a^5}) a_{in}^{5/2}}{(\sqrt{6}\sqrt{\rho_s + \rho_w a^5} + \sqrt{6\rho_s + \rho_w a^5}) a^{5/2}} \right] \right\} + ln \left[\frac{\sqrt{\rho_s + \rho_w a^5} + \sqrt{6\rho_s + \rho_w a^5}}{\sqrt{\rho_s + \rho_w a^5} + \sqrt{6\rho_s + \rho_w a^5}} \right] + \phi(a_{in}) \quad \text{where} \quad n = 6, \qquad (2.160)$$

for domain walls and radiation as

$$\phi(a) = \pm \frac{1}{3\sqrt{2\pi G}} \{-2ln[\frac{(2\sqrt{\rho_r + \rho_w a^3} + \sqrt{4\rho_r + \rho_w a^3})a_{in}^{3/2}}{(2\sqrt{\rho_r + \rho_w a_{in}^3} + \sqrt{4\rho_r + \rho_w a_{in}^3})a^{3/2}}] + ln[\frac{\sqrt{\rho_r + \rho_w a^3} + \sqrt{4\rho_r + \rho_w a^3}}{\sqrt{\rho_r + \rho_w a_{in}^3} + \sqrt{4\rho_r + \rho_w a_{in}^3}}]\} + \phi(a_{in}) \quad \text{where} \quad n = 4, \quad (2.161)$$

for domain walls and matter as

$$\phi(a) = \pm \frac{1}{2\sqrt{2\pi G}} \left\{ -\sqrt{3}ln \left[\frac{(\sqrt{3}\sqrt{\rho_m + \rho_w a^2} + \sqrt{3\rho_m + \rho_w a^2})a_{in}}{(\sqrt{3}\sqrt{\rho_m + \rho_w a^2_{in}} + \sqrt{3\rho_m + \rho_w a^2_{in}})a} \right] + ln \left[\frac{\sqrt{\rho_m + \rho_w a^2} + \sqrt{3\rho_m + \rho_w a^2}}{\sqrt{\rho_m + \rho_w a^2_{in}} + \sqrt{3\rho_m + \rho_w a^2_{in}}} \right] \right\} + \phi(a_{in}) \quad \text{where} \quad n = 3.$$
(2.162)

It is possible to simply the relation between time and the scale factor as

$$t = \int_{a_{in}}^{a} \frac{da'}{a'H(a')},$$

$$t = \int_{a_{in}}^{a} \frac{da'}{a'\sqrt{\frac{8\pi G}{3}(\frac{\rho_w}{a'} + \frac{\rho_n}{a'^n})}},$$
(2.163)

$$t = \sqrt{\frac{3}{2\pi G\rho_w}} \left[\sqrt{a'_2} F_1(\frac{1}{2}, \frac{1}{2-2n}; \frac{1}{2-2n} + 1; -\frac{\rho_n}{\rho_w} a'^{(1-n)}) \right]_{a_{in}}^a,$$
(2.164)

where $_2F_1$ is the hypergeometric function which was introduced in Section 2.3.2.2. Details of this calculation is given in Appendix A.6. To avoid singularities in the hypergeometric function in (2.164) n should be chosen such that 3/2 < n.

2.4.3. Dark Energy Revisited

We have already examined the case $V = V_0$ in Section 2.3.2. For spatially flat universe we have obtained

$$a(t) = \left[\frac{b^2 V_0 e^{2\mu t} - \gamma(a_{in}) a_{in}^6 e^{-2\mu t}}{2bV_0}\right]^{1/3},$$

$$b = a_{in}^3 + \sqrt{a_{in}^6 + \frac{\gamma(a_{in}) a_{in}^6}{V_0}} \quad \text{and} \quad \mu = \sqrt{6\pi GV_0}.$$

Then we formulate the Hubble function and the deceleration parameter as a function of t as

$$H(t) = \frac{2\mu}{3} \left(1 - \frac{2f}{f - b^2 V_0 e^{4\mu t}}\right), \qquad f = \gamma(a_{in}) a_{in}^6, \tag{2.165}$$

$$q(t) = -1 + \frac{12b^2 f V_0 e^{4\mu t}}{(1 + b^2 V_0 e^{4\mu t})^2}.$$
(2.166)

First constraint on our parameters comes from positivity of the Hubble function

$$H(t) > 0 \qquad \Rightarrow \qquad f < b^2 V_0. \tag{2.167}$$

On the other hand q(t) has only one real root

$$q(t) = 0 \qquad \Rightarrow \qquad t = \frac{1}{4\mu} ln[\frac{(5+2\sqrt{6})f}{b^2 V_0}].$$
 (2.168)

We choose

$$t > 0 \qquad \Rightarrow \qquad f > \frac{b^2 V_0}{\tau}, \qquad \tau = (5 + 2\sqrt{6}).$$
 (2.169)

One can write

$$f = \left(\frac{1}{\tau} + \varepsilon\right)b^2 V_0, \quad 0 < \varepsilon < 1 - \frac{1}{\tau} \quad \Rightarrow \frac{b^2 V_0}{\tau} < f < b^2 V_0. \tag{2.170}$$

Now we will find the condition which results in acceleration at the beginning of the universe

$$\lim_{t \to 0} q(t) = -1 + \frac{12b^2 f V_0}{(f + b^2 V_0)^2} < 0.$$
(2.171)

Thus

$$5 - \frac{1}{\tau} - \sqrt{24} < \varepsilon < 1 - \frac{1}{\tau}, \tag{2.172}$$

$$4.3 \times 10^{-16} < \varepsilon < 0.90. \tag{2.173}$$

With these initial conditions universe starts to expand with acceleration and then turns into deceleration.

2.4.4. Combination Containing Exotic Matter

We have already explored the case of exotic matter in Section 2.3.2. Now we will study combination of this kind of matter and some ordinary matters in the early universe. We have taken the most general form of the potential as

$$V(a) = \frac{V_s}{a^6} + \frac{V_n}{a^n}.$$
 (2.174)

Related cosmological functions have been found as

$$\gamma(a) = \frac{1}{a^6} \left[\frac{nV_n}{6-n} a^{6-n} + 6V_s ln(\frac{a}{a_{in}}) + a^6_{in} \tilde{\gamma}(a_{in}) \right], \quad \tilde{\gamma}(a_{in}) = \gamma(a_{in}) - \frac{nV_n}{(6-n)a^n_{in}},$$
(2.175)

$$H(a) = \sqrt{\frac{8\pi G}{3} \left[\frac{6V_n}{(6-n)a^n} + \frac{a_{in}^6 \tilde{\gamma}(a_{in}) + V_s \left[1 + 6ln(\frac{a}{a_{in}})\right]}{a^6}\right] - \frac{k}{a^2}},$$
(2.176)

$$\phi(a) = \pm \sqrt{6} \int_{a_{in}}^{a} \left\{ \left[(6-n) [a_{in}^{6} \tilde{\gamma}(a_{in}) + 6V_{s} ln(\frac{a'}{a_{in}})] + nV_{n}a'^{6} \right] \right. \\ \left. \left. \left. \left[(6-n) \{ 8\pi G[a_{in}^{6} \tilde{\gamma}(a_{in}) + V_{s}(1 + 6ln(\frac{a'}{a_{in}}))] - 3ka'^{4} \} a'^{2+n} + 48\pi GV_{n}a'^{8} \right] \right\}^{1/2} da' \\ \left. + \phi(a_{in}), \right\}$$

$$(2.177)$$

$$\rho(a) = \frac{6V_n}{(6-n)a^n} + \frac{V_s}{a^6} ln[e(\frac{a}{a_{in}})^6] + \frac{a_{in}^6\tilde{\gamma}(a_{in})}{a^6}, \qquad (2.178)$$

$$p(a) = \frac{2(n-3)V_n}{(6-n)a^n} + \frac{V_s}{a^6} ln[\frac{1}{e}(\frac{a}{a_{in}})^6] + \frac{a_{in}^6\tilde{\gamma}(a_{in})}{a^6},$$
(2.179)

$$q(a) = \frac{8\pi G\{(6-n)[2a_{in}^{6}\tilde{\gamma}(a_{in}) + V_{s}(-1 + 12ln(\frac{a}{a_{in}}))]a^{n} + 3(n-2)V_{n}a^{6}\}}{(6-n)\{8\pi G[a_{in}^{6}\tilde{\gamma}(a_{in}) + V_{s}(1 + 6ln(\frac{a}{a_{in}}))] - 3ka^{4}\}a^{n} + 48\pi GV_{n}a^{6}}.$$
 (2.180)

<u>2.4.4.1.</u> Combination with Radiation. When the exotic matter is accompanied with radiation its dynamics in spatially flat universe is governed by the following deceleration parameter

$$q(a) = \frac{2a_{in}^6\tilde{\gamma}(a_{in}) + V_s[-1 + 12ln(\frac{a}{a_{in}})] + 3V_r a^2}{a_{in}^6\tilde{\gamma}(a_{in}) + V_s[1 + 6ln(\frac{a}{a_{in}})] + 3V_r a^2}, \quad \tilde{\gamma}(a_{in}) = \gamma(a_{in}) - \frac{2V_r}{a_{in}^4}.$$
 (2.181)

One can trace its behaviour back into time as

$$q(a_{in}) = \frac{2a_{in}^6\gamma(a_{in}) - V_r a_{in}^2 - V_s}{a_{in}^6\gamma(a_{in}) + V_r a_{in}^2 + V_s}.$$
(2.182)

The following condition

$$V_s + V_r a_{in}^2 > 2a_{in}^6 \gamma(a_{in}). \tag{2.183}$$

causes accelerated beginning for the universe. Moreover this choice also results in negative total pressure in the beginning as

$$p(a_{in}) = \frac{\gamma(a_{in})a_{in}^6 - V_r a_{in}^2 - V_s}{a_{in}^6} < 0, \qquad (2.184)$$

while energy density remains positive. Thus negative pressure results in accelerated motion for a while. Then this behaviour changes as pressure becomes positive and the universe decelerates. Therefore this exotic matter and radiation with initial condition which satisfies (2.183) also has mathematical turn on and turn off mechanism for accelerated motion in the early universe.

2.4.4.2. Combination with Domain Walls. In spatially flat universe the deceleration parameter becomes

$$q(a) = \frac{10a_{in}^6 \tilde{\gamma}(a_{in}) - 3v_w a^5 + 5v_s [-1 + 12ln(\frac{a}{a_{in}})]}{5a_{in}^6 \tilde{\gamma}(a_{in}) + 6v_w a^5 + 5v_s [1 + 6ln(\frac{a}{a_{in}})]}.$$
(2.185)

Although this universe may start to expand with acceleration or deceleration, after a while it will accelerate because the leading term a^5 in the numerator has a negative coefficient.

<u>2.4.4.3.</u> Combination with Dark Energy. In spatially flat universe the deceleration parameter becomes

$$q(a) = \frac{2a_{in}^{6}\tilde{\gamma}(a_{in}) - v_0 a^6 + v_s [-1 + 12ln(\frac{a}{a_{in}})]}{a_{in}^{6}\tilde{\gamma}(a_{in}) + v_0 a^6 + v_s [1 + 6ln(\frac{a}{a_{in}})]}.$$
(2.186)

Although this universe may start to expand with acceleration or deceleration, after a while it will accelerate because the leading term a^6 in the numerator has a negative coefficient.

2.5. Late Time Expansion of the Universe

We try to understand the present era which is usually described by dark matter and dark energy. However we consider different forms of energy components. Thus in the first subsection we model a universe with domain walls and matter. After obtaining formulas for functions we compare our results with supernova type Ia data just by curve fitting. In the second subsection we study dark energy dominated universe with the same steps.

2.5.1. Cosmic Walls

The hypothesis that the scalar field is the dark matter and the dark energy was investigated for flat universe in [120]. The results were compared with observations of type Ia supernovae which were available in 2000. In that study, matter part of the universe was neglected and it was found that $\rho_{\phi} \sim a^{-1.09}$ and $q_0 = -0.45$. Different from them we include matter component of the universe and we solve field equations analytically. Then we compare our results with the type Ia supernovae data released in 2018 [121]. Furthermore in this comparison we extract the value of Hubble constant H_0 and the value of absolute magnitude M with cosmological density parameters.

Today our universe is believed to be almost flat and contribution of radiation to the energy density is very tiny. For this reason we investigate the case in which

$$\rho(a) = \frac{\rho_m}{a^3} + \frac{\rho_w}{a}.$$
 (2.187)

Domain wall dominated universes have been already studied in Section 2.4.2. Just plugging n = 3 in Section 2.4.2.2 we obtain the Hubble function, the scalar field and the potential as

$$H(a) = \sqrt{\frac{8\pi G}{3}} \left(\frac{\rho_w}{a} + \frac{\rho_m}{a^3}\right),$$
(2.188)

$$\phi(a) = \pm \frac{1}{2\sqrt{2\pi G}} \left\{-\sqrt{3} ln \left[\frac{(\sqrt{3}\sqrt{\rho_m + \rho_w a^2} + \sqrt{3\rho_m + \rho_w a^2})a_{in}}{(\sqrt{3}\sqrt{\rho_m + \rho_w a^2_{in}} + \sqrt{3\rho_m + \rho_w a^2_{in}})a}\right] + ln \left[\frac{\sqrt{\rho_m + \rho_w a^2} + \sqrt{3\rho_m + \rho_w a^2_{in}}}{\sqrt{\rho_m + \rho_w a^2_{in}} + \sqrt{3\rho_m + \rho_w a^2_{in}}}\right] + \phi(a_{in}),$$
(2.189)

$$V(a) = \frac{5\rho_w}{6a} + \frac{\rho_m}{2a^3},$$
(2.190)

$$p(a) = -\frac{2\rho_w}{3a},$$
(2.191)

$$q(a) = -\frac{1}{2} + \frac{\rho_m}{\rho_m + \rho_w a^2},$$
(2.192)

$$t = \sqrt{\frac{3}{2\pi G\rho_w}} \left[\sqrt{a'_2} F_1(\frac{1}{2}, -\frac{1}{4}; \frac{3}{4}; -\frac{\rho_m}{\rho_w a'^2}) \right]_{a_{in}}^a.$$
 (2.193)

The last relation is obtained by substituting n = 3 in (2.164). Investigation of the deceleration parameter tells us

$$\lim_{a \to 0} q(a) = \frac{1}{2}, \qquad \lim_{a \to \infty} q(a) = -\frac{1}{2}, \qquad (2.194)$$

$$q(a) = 0 \qquad \Rightarrow \qquad a = \sqrt{\frac{\rho_m}{\rho_w}}.$$
 (2.195)

Hence if $\rho_m < \rho_w$ this model of the universe accelerates.

To test the reliability of the theoretical model we will use supernovae data. Luminosity distance-redshift (For definition see the Appendix A.9.) relation had been already constructed as [122]

$$d_L = (1+z)R_0 S(\chi(z)).$$
(2.196)

The comoving coordinate χ is given as

$$\chi(z) = \frac{c}{R_0} \int_0^z \frac{dz'}{H(z')}.$$
(2.197)

The function $r = S(\chi)$ is given by

$$S(\chi) = \begin{cases} Sin(\chi) & \text{if } k = 1, \\ \chi & \text{if } k = 0, \\ Sinh(\chi) & \text{if } k = -1 \end{cases}$$

Thus

$$R_0 S(\chi(z)) = \frac{c}{H_0} \begin{cases} \left|\Omega_{k,0}\right|^{-1/2} S(\sqrt{|\Omega_{k,0}|} E(z)) & \text{for } \Omega_k \neq 0, \\ E(z) & \text{for } \Omega_k = 0, \end{cases}$$

where $E(z) = \frac{R_0 H_0}{c} \chi(z)$.

For spatially flat universe with matter and domain walls it reduces to the following form

$$d_L = \frac{c(1+z)}{H_0} \int_0^z \frac{dz'}{[\Omega_m (1+z')^3 + \Omega_w (1+z')]^{1/2}}$$
(2.198)

where $\Omega_m = 1 - \Omega_w$. The relation between observational measurements and the theory are established as

$$m = 5log_{10}(\frac{d_L}{1Mpc}) + 25 + M, \qquad (2.199)$$

where m and M are the apparent and the absolute magnitudes respectively. Then the distance modulus is defined as $\mu = m - M$.

Before going further we would like to remind the Hubble tension and supernova absolute magnitude tension. Both of them are fundamental cosmological parameters. Their values must be presented precisely. To determine the value of H_0 different methods have been applied. According to the Planck measurement of the cosmic microwave background (CMB) anisotropies, assuming the base- Λ CDM model [51] $H_0 = 67.36 \pm 0.54 km s^{-1} Mpc^{-1}$. On the other hand Hubble Space Telescope (HST) observations of Cepheids have been used to calibrate the measurements using type Ia supernovae in [123] and it has been declared $H_0 = 74.13 \pm 1.42 km s^{-1} Mpc^{-1}$.

In last years it has been pointed out that the absolute peak magnitude M_B of Type Ia supernovae is converted into a value of H_0 [124–127]. It's value has been stated as $M_B = -19.401 \pm 0.027$ mag [128] in 2020 and $M_B = -19.244 \pm 0.037$ mag [125] in 2021 by application of different methods.

The most recent data set for type Ia supernova observation which is called as Pantheon dataset was released in [121]. 1048 data points are presented as (m, z) pairs where z < 2.3. Since there is a debate on values of H_0 and M we include their values as parameters which are to be derived from curve fitting. Therefore we have to extract values of ρ_w , H_0 and M from data. To be able to see effects of these numbers on each other separately we applied the curve fitting method for three combinations of two parameter sets.

Then we applied the χ^2 test to measure the goodness of these fits. χ^2 per degrees of freedom, χ^2_{ν} is calculated according to following formula

$$\chi^{2} = \sum_{i}^{N} \frac{(\mu_{i}^{data} - \mu_{i}^{model})^{2}}{\sigma_{i}^{2}},$$
(2.200)

$$\chi^2_{\nu} = \frac{\chi^2}{\nu}, \qquad \nu = N - k,$$
(2.201)

where k is the number of parameters that will be extracted from the N number of data points.

First, we assign trial number for Ω_w and results are shown in Table 2.1. The best fit which is obtained for $\Omega_w = 0.90$ gives $H_0 = 72.1563 \pm 0.0001$, $M = -19.269 \pm 0.004$ and $\chi^2/\nu = 1.009$.

Ω_w	H_0	M	χ^2/ u
0.70	72.0756 ± 0.0001	-19.213 ± 0.004	1.124
0.75	72.0945 ± 0.0001	-19.227 ± 0.004	1.075
0.80	72.1142 ± 0.0001	-19.240 ± 0.004	1.038
0.85	72.1347 ± 0.0001	-19.255 ± 0.004	1.014
0.90	72.1563 ± 0.0001	-19.269 ± 0.004	1.009
0.95	72.1789 ± 0.0001	-19.285 ± 0.004	1.026
0.9999	72.2028 ± 0.0001	-19.301 ± 0.004	1.073

Table 2.1. Values of H_0 and M for Given Ω_w .

Table 2.2. Values of M and Ω_w for Given H_0 .

H_0	M	Ω_w	χ^2/ u
74	-19.211 ± 0.006	0.888 ± 0.015	1.008
73	-19.240 ± 0.006	0.888 ± 0.015	1.008
72	-19.270 ± 0.006	0.888 ± 0.015	1.008
71	-19.301 ± 0.006	0.888 ± 0.015	1.008
70	-19.332 ± 0.006	0.888 ± 0.015	1.008
69	-19.363 ± 0.006	0.888 ± 0.015	1.008
68	-19.395 ± 0.006	0.888 ± 0.015	1.008
67	-19.427 ± 0.006	0.888 ± 0.015	1.008

After getting intuition about parameters we perform curve fitting for trial H_0 values. Results are given in Table 2.2. Numbers in Table 2.2 tell us that the hundredths digit of M is sensitively depended on the ones digit of H_0 . In addition, more accurate result for Ω_w is obtained. All of the results have the same χ^2/ν value. Thus we choose

 3^{rd} line: H = 72, $M = -19.270 \pm 0.006$, $\Omega_w = 0.888 \pm 0.015$ and $\chi^2/\nu = 1.008$. These numbers are compatible with best fit of the table 2.1

Effect of trial values of M on H_0 and Ω_w are presented in Table 2.3. It is apparent that the value of the ones digit of H_0 is sensitively depended on the hundredths digit of M. Our choice is the 5th line: M = 19.25, $H_0 = 72.68 \pm 0.21$, $\Omega_w = 0.889 \pm 0.015$ and $\chi^2/\nu = 1.008$. These numbers are in agreement with best fit of the table 2.1.

M	H_0	Ω_w	χ^2/ν
-19.45	66.29 ± 0.19	0.889 ± 0.015	1.008
-19.40	67.83 ± 0.20	0.889 ± 0.015	1.008
-19.35	69.41 ± 0.20	0.889 ± 0.015	1.008
-19.30	71.03 ± 0.20	0.889 ± 0.015	1.008
-19.25	72.68 ± 0.21	0.889 ± 0.015	1.008
-19.20	74.38 ± 0.22	0.889 ± 0.015	1.008
-19.15	76.11 ± 0.22	0.889 ± 0.015	1.008

Table 2.3. Values of H_0 and Ω_w for Given M.

Assuming base Λ CDM cosmology, late universe parameters were found as $H_0 = 67.27 \pm 0.60$, $\Omega_m = 0.3166 \pm 0.0084$ and $\Omega_{\Lambda} = 0.6834 \pm 0.0084$ in [51]. It is known that $\Omega_m = \Omega_{bm} + \Omega_{dm}$ where $\Omega_{bm} \simeq 0.05$ and $\Omega_{dm} \simeq 0.27$. However our results indicate that late universe parameters as $H_0 = 72.68 \pm 0.21$, $\Omega_w = 0.889 \pm 0.015$ and $\Omega_m = 0.111 \pm 0.015$. Since $\Omega_{bm} \simeq 0.05$, $\Omega_{dm} \simeq 0.06$. Therefore domain wall dominated universe is a candidate to explain 94 percentage of the structures in the present universe while still 6 percentage of the universe remains as unknown.

To compare our results with Pantheon-data graphically we draw distance modulus μ vs redshift z plot. In Figure 2.1 we use results given in 5th line of Table 2.3 where $M = -19.25, H_0 = 72.68 \pm 0.21, \rho_w = 0.889 \pm 0.015$ and $\rho_m = 0.111 \pm 0.015$.



Figure 2.1. Distance Modulus vs Redshift Plot for Domain Wall Dominated Universe. Dots represent observation of Pantheon data while green line represents domain wall dominated universe.

We obtain $q_0 = -0.389$ by using values of cosmological density parameters in (2.192).

2.5.2. Dark Energy

Now we will present exact solutions for energy density given as

$$\rho = \rho_0 + \frac{\rho_m}{a^3}.$$
 (2.202)

Actually this case corresponds to n = 3 in Section 2.4.1. For k = 0 we have already obtained the scalar factor as

$$a(t) = \left[\frac{(a_{in}^{3/2} + \sqrt{a_{in}^3 + \frac{\rho_m}{\rho_0}})e^{3\mu t/2} + (a_{in}^{3/2} - \sqrt{a_{in}^3 + \frac{\rho_m}{\rho_0}})e^{-3\mu t/2}}{2}\right]^{2/3}, \qquad (2.203)$$

where $\mu = \sqrt{\frac{8\pi G\rho_0}{3}}$. In addition potential is formulated as

$$V(\psi) = \frac{1}{2} \frac{\rho_m}{a^3(\psi)} + \rho_0, \tag{2.204}$$

$$a(\psi) = 2^{2/3} \left\{ \frac{a_{in}^{3/2} e^{\sqrt{3}\psi} \left[-1 + \sqrt{1 + a_{in}^3 \frac{\rho_0}{\rho_m}} - \left(1 + \sqrt{1 + a_{in}^3 \frac{\rho_0}{\rho_m}}\right) e^{2\sqrt{3}\psi}\right]}{a_{in}^3 \frac{\rho_0}{\rho_m} (1 + e^{4\sqrt{3}\psi}) - 2\left(2 + a_{in}^3 \frac{\rho_0}{\rho_m}\right) e^{2\sqrt{3}\psi}} \right\}^{2/3}, \quad (2.205)$$

$$\psi = \pm \sqrt{2\pi G} [\phi(a) - \phi(a_{in})].$$
(2.206)

Behaviour of the deceleration parameter is shown by

$$q(a) = \frac{-2\rho_0 a^3 + \rho_m}{2(\rho_0 a^3 + \rho_m)},$$
(2.207)

$$\lim_{a \to \infty} q(a) = -1. \tag{2.208}$$

To extract cosmological parameters from Pantheon data we apply the procedure as explained in the previous subsection with modification

$$d_L = \frac{c(1+z)}{H_0} \int_0^z \frac{dz'}{[\Omega_m (1+z')^3 + \Omega_0]^{1/2}},$$
(2.209)

where $\Omega_m = 1 - \Omega_0$.

Table 2.4. Values of H_0 and M for Given Ω_0 .

Ω_0	H_0	M	χ^2/ν
0.50	72.0674 ± 0.0001	-19.208 ± 0.004	1.199
0.60	72.1183 ± 0.0001	-19.243 ± 0.004	1.059
0.70	72.1749 ± 0.0001	-19.282 ± 0.004	0.992
0.80	72.2384 ± 0.0001	-19.324 ± 0.004	1.041
0.90	72.3116 ± 0.0001	-19.372 ± 0.005	1.296

First, we perform curve fitting for trial values of Ω_0 . Results are shown in Table 2.4. The best fit occurs for $\Omega_0 = 0.70$. Thus $H_0 = 72.1749 \pm 0.0001$, $M = -19.282 \pm 0.004$ and $\chi^2/\nu = 0.992$.

Then we perform curve fitting for trial H_0 values. Results are given in Table 2.5. Results in the third line are compatible with the best fit of Table 2.4. $M = -19.294 \pm 0.007$, $\Omega_0 = 0.715 \pm 0.012$ and $\chi^2/\nu = 0.990$ are obtained for a given $H_0 = 72$.

H_0	M	Ω_0	χ^2/ u
74	-19.234 ± 0.007	0.715 ± 0.012	0.990
73	-19.264 ± 0.007	0.715 ± 0.012	0.990
72	-19.294 ± 0.007	0.715 ± 0.012	0.990
71	-19.324 ± 0.007	0.715 ± 0.012	0.990
70	-19.355 ± 0.007	0.715 ± 0.012	0.990
69	-19.386 ± 0.007	0.715 ± 0.012	0.990
68	-19.418 ± 0.007	0.715 ± 0.012	0.990
67	-19.450 ± 0.007	0.715 ± 0.012	0.990

Table 2.5. Values of M and Ω_0 for Given H_0 .

Finally we test the effect of M on H_0 and Ω_0 . Results are presented in Table 2.6. Numbers in the fourth line are compatible with the best fit of Table 2.4. $H_0 = 71.80 \pm 0.22$, $\Omega_0 = 0.715 \pm 0.012$ and $\chi^2/\nu = 0.990$ are found for a given M = -19.30.

All the tables in this section have a common interpretation: The number in ones digit of the Hubble constant H_0 is sensitively depended on the number in the hundredths digit of the absolute magnitude M in both models. In addition as the value of H_0 increases, the value of M decreases. We will stop to dig more about this argument here since it is beyond the scope of our aims. Cosmologists will continue to reveal the relation between H_0 and M more clearly on further studies.

Now we would like to compare our results for a domain wall dominated universe and a dark energy dominated universe in the same plot. However this goal can not be achieved accurately, because best fit values of M are different for both models. For this reason we plot two figures.

M	H_0	Ω_0	χ^2/ν
-19.45	67.00 ± 0.21	0.715 ± 0.012	0.990
-19.40	68.57 ± 0.21	0.715 ± 0.012	0.990
-19.35	70.16 ± 0.22	0.715 ± 0.012	0.990
-19.30	71.80 ± 0.22	0.715 ± 0.012	0.990
-19.25	73.47 ± 0.23	0.715 ± 0.012	0.990
-19.20	75.18 ± 0.23	0.715 ± 0.012	0.990
-19.15	76.93 ± 0.24	0.715 ± 0.012	0.990

Table 2.6. Values of H_0 and Ω_0 for Given M.





Figure 2.2. Distance Modulus vs Redshift Figure 2.3. Distance Modulus vs Redshift Plot for M = -19.25.

Dots represent observations, green line represents domain wall dominated universe and red line represents dark energy dominated universe.

Plot for M = -19.30.

Dots represent observations, green line represents domain wall dominated universe and red line represents dark energy dominated universe.

We draw our Figure 2.2 by taking M = -19.25 for which one of the best fits of the domain wall dominated universe is obtained with parameters $H_0 = 72.68 \pm 0.21$, $\Omega_w = 0.889 \pm 0.015$ and $\Omega_m = 0.111 \pm 0.015$. On the other hand we have obtained $H_0 = 73.47 \pm 0.23$, $\Omega_0 = 0.715 \pm 0.012$, $\Omega_m = 0.285 \pm 0.12$ and $\chi^2/\nu = 0.990$ for M = -19.25.

In Figure 2.3 we choose one of the best fits of the dark energy dominated universe for which M = -19.30, $H_0 = 71.80 \pm 0.22$, $\Omega_0 = 0.715 \pm 0.012$ and $\Omega_m = 0.285 \pm 0.012$. In the first model we have obtained $H_0 = 71.03 \pm 0.20$, $\Omega_w = 0.889 \pm 0.015$, $\Omega_m = 0.111 \pm 0.015$ and $\chi^2/\nu = 0.990 = 1.008$ for M = -19.30

Two figures are almost the same because χ^2/ν for both models are very close the 1. We need more data for bigger redshift values to decide whether one of the models is superior to the other one. We obtain $q_0 = -0.572$ by using these values of cosmological parameters in (2.207).

2.6. Discussion

We studied FLRW cosmology with real scalar field which is minimally coupled to gravity. Our main motivation in this chapter has been to assign the effective scalar field a source of all components of energy density. We applied a change of variable twice which is a more powerful method in the set of differential equations which represent dynamics of the universe. In the first one, we have replaced the independent variable "t" with "a". In the second one, we have changed the dependent variable of the ϕ equation so that it became a first order linear differential equation. We presented exact solutions in four different forms; solutions for given V(a), solutions for given $\phi(a)'$, solutions for given H(a), and solutions for given $\rho(a)$.

Then we have examined single component universes and two component universes for a given energy density and for a given potential. In these solutions we have searched for mathematical mechanisms which create turn on and turn off for early inflationary expansion. We have explored mathematical structure of a new exotic matter. Equation of state for this component changes with the scale factor or equivalently changes with time. A universe which consists of radiation and this exotic matter, has mathematical machinery to turn on and to turn off inflationary expansion in early epoch.

We have investigated the present era of the universe. Domain wall dominated universe and dark energy dominated universe have been studied. We have extracted numerical values of cosmological parameters from the most recent type Ia supernova data by taking care of the Hubble tension and the absolute magnitude tension. For domain wall dominated universe we have found that $\Omega_w = 0.889 \pm 0.015$, $\Omega_m = 0.111 \pm 0.015$ and $H_0 = 72.68 \pm 0.21$ for M = -19.25. This universe accelerates with $q_0 = -0.389$.

On the other hand for dark energy dominated universe cosmological parameters have been found as $\Omega_0 = 0.715 \pm 0.012$, $\Omega_m = 0.285 \pm 0.012$ and $H_0 = 71.80 \pm 0.22$ for M = -19.30. Deceleration parameter of this universe is q = -0.572. Detailed analysis for the relation between distance modulus and redshift have shown that the number in ones digit of the Hubble constant H_0 is sensitively depended on the number in the hundredths digit of the absolute magnitude M in both models.

These two analyses indicate that both models equivalently explains dynamics of late time accelerated expansion of the universe. The difference between these models most probably will be seen when bigger redshift data are available. One can also reach our calculations which are presented in this chapter in [129].

3. ANALYTIC SOLUTIONS OF BRANS-DICKE COSMOLOGY

The major purpose of this chapter is to solve field equations of Brans-Dicke cosmology analytically. First, we present field equations in Section 3.1. Then we introduce our mathematical techniques in Section 3.2 and we rewrite the field equations with the scale size "a" as the new independent variable. This set of equations can be reduced to a constraint equation and a first order differential equation. Exact solutions are found in Section 3.3 for given $\phi(a)$ and $\rho(a)$, in Section 3.4 for given $\phi(a)$ and V(a), in Section 3.5 for given $\phi(a)$ and H(a). A universe consisting of single component energy-matter density is investigated in Section 3.6. The early epoch of the universe is studied in Section 3.7. We present a solution for a universe which contains dark energy and matter in Section 3.8.1 and we present a solution for a universe which consists of cosmic domain walls and matter in Section 3.8.2. Comparison of both models with observation is illustrated by a plot. Discussion and summary are given in Section 3.9. In this study we choose Brans-Dicke parameter $\omega > 4 \times 10^4$ to be compatible with results of Einstein telescope [130] and time delay experiments [131].

3.1. Field Equations

In this study we will use FLWR metric given by,

$$ds^{2} = dt^{2} - a^{2}(t)\left[\frac{dr^{2}}{1 - kr^{2}} + r^{2}(d\theta^{2} + \sin^{2}\theta d\varphi^{2})\right].$$
(3.1)

where k is the curvature parameter, $a(t) = \frac{R(t)}{R(t_0)}$ is the normalized scale factor of the universe. Here r has dimension of *lenght*, a(t) is dimensionless and k has diemsion of

 $lenght^{-2}$. In BDJT the Lagrange density can be written as [23, 132-135]

$$\mathcal{L} = \left[-\Phi R + \omega \frac{1}{\Phi} g^{\mu\nu} \partial_{\mu} \Phi \partial_{\nu} \Phi - V(\Phi) + L_M \right] \sqrt{-g}, \qquad (3.2)$$

$$= \left[-\frac{1}{8\omega} \phi^2 R + \frac{1}{2} g^{\mu\nu} \partial_\mu \phi \partial_\nu \phi - V(\phi) + L_M \right] \sqrt{-g}, \qquad (3.3)$$

where we will call ϕ as the Jordan scalar field and Φ as the Brans-Dicke scalar field, the two being related by

$$\Phi = \frac{1}{8\omega}\phi^2,\tag{3.4}$$

where ω is the dimensionless Brans-Dicke parameter, R is the Ricci scalar and L_M represents the contribution due to matter fields. We use the metric signature (+, -, -, -) and units with $\hbar = 1$, c = 1. We prefer to use the Jordan scalar for the scalar field since in flat spacetime the Lagrangian then becomes the standard Lagrangian for a scalar field

$$\mathcal{L} = \frac{1}{2} \partial_{\mu} \phi \partial^{\mu} \phi - V(\phi).$$
(3.5)

The homogeneous and isotropic cosmological field equations obtained from this Lagrangian density have already been calculated for a potential $V(\phi) = \frac{1}{2}m^2\phi^2$ by Arik, Calik and Sheftel in [136–138]. Thus their general form for a potential $V(\phi)$ are given by,

$$\frac{3}{4\omega}\phi^2\left(\frac{\dot{a}^2}{a^2} + \frac{k}{a^2}\right) - \frac{1}{2}\dot{\phi}^2 - V(\phi) + \frac{3}{2\omega}\frac{\dot{a}}{a}\dot{\phi}\phi = \rho_m, \qquad (3.6)$$

$$\frac{-1}{4\omega}\phi^2 \left(2\frac{\ddot{a}}{a} + \frac{\dot{a}^2}{a^2} + \frac{k}{a^2}\right) - \frac{1}{\omega}\frac{\dot{a}}{a}\dot{\phi}\phi - \frac{1}{2\omega}\ddot{\phi}\phi - \left(\frac{1}{2} + \frac{1}{2\omega}\right)\dot{\phi}^2 + V(\phi) = p_m, \quad (3.7)$$

$$\ddot{\phi} + 3\frac{\dot{a}}{a}\dot{\phi} + \frac{dV(\phi)}{d\phi} - \frac{3}{2\omega}\left(\frac{\ddot{a}}{a} + \frac{\dot{a}^2}{a^2} + \frac{k}{a^2}\right)\phi = 0.$$
(3.8)
3.2. Mathematical Techniques

It is apparent that the independent variable time t is not seen explicitly in our set of differential equations given by (3.6-3.8). For this reason one can choose a new variable in terms of old dependent variables. Hence the order of differential equation can be reduced by one [117]. Our choice is $\dot{a} = aH(a)$. Then a becomes new independent variable and field equations transform into the following form

$$\frac{3}{4\omega}\phi^{2}\left(H^{2}+\frac{k}{a^{2}}\right) - \frac{1}{2}\left(\phi'Ha\right)^{2} - V(a) + \frac{3}{2\omega}\phi\phi'H^{2}a = \rho_{m},$$
(3.9)
$$-\frac{1}{4\omega}\phi^{2}\left(3H^{2}+2H'Ha+\frac{k}{a^{2}}\right) - \frac{3}{2\omega}\phi'\phi H^{2}a - \frac{1}{2\omega}\left(\phi''H^{2}a^{2}+\phi'H'Ha^{2}\right)\phi$$

$$-\left(\frac{1}{2}+\frac{1}{2\omega}\right)\left(\phi'Ha\right)^{2} + V(a) = p_{m},$$
(3.10)
$$\phi''H^{2}a^{2}+\phi'H'Ha^{2}+4\phi'H^{2}a + \frac{V'}{\phi'} - \frac{3}{2\omega}\left(2H^{2}+H'Ha+\frac{k}{a^{2}}\right)\phi = 0,$$
(3.11)

where the prime denotes $\frac{d}{da}$ (For details of the transformation see Appendix A.3.). We will name (3.9) as the energy density equation, (3.10) as the pressure equation and (3.11) as the ϕ equation.

These three equations are connected by the perfect fluid continuity equation,

$$\dot{\rho}(a) + 3\frac{\dot{a}}{a}(\rho(a) + p(a)) = 0, \qquad (3.12)$$

$$[\rho'(a)a + 3(\rho(a) + p(a))]H(a) = 0.$$
(3.13)

In other words when we plug energy density and pressure which are given by (3.9-3.10) in (3.13) we end up with $-a\phi'(a)H(a) \times (3.11)$. This shows that we have two independent equations and four unknown functions: H(a), V(a), $\phi(a)$ and $\rho(a)$ (one can choose p(a) instead of $\rho(a)$). In our calculations we will use the energy density equation and the ϕ equation. When we plug solutions in to the pressure equation, $\rho(a)$ and the resulting p(a) will satisfy the perfect fluid equation of state.

It is seen that energy density equation is a first order variable coefficient and nonlinear differential equation for $\phi(a)$. In addition this is not a type of Bernoulli equation which is analytically solvable. In addition (3.11) is second order variable coefficient nonlinear differential equation for $\phi(a)$. Analytic solutions for both of them have not been established yet. To be able to solve them, $\phi(a)$ should be given. Then we will be left with $\rho(a)$, V(a) and H(a). Since (3.9) does not contain any derivative of last three functions it is just a constraint. On the other hand first derivatives of H(a)and V(a) are seen in (3.11). Thus our system consist of a constraint equation and a first order differential equation. One of the following functions; H(a), V(a) and $\phi(a)$ is still free. One of them should be accompanied with $\phi(a)$ as a given function. Three combinations exist; $\rho(a)$ and $\phi(a)$, V(a) and $\phi(a)$, H(a) and $\phi(a)$. We will solve our system for these three sets separately.

3.3. Solution for Given $\rho(a)$ and $\phi(a)$

One can solve the constraint Equation (3.9) and the first order differential Equation (3.11) by different methods:

- Method 1) First, find V(a) from the constraint equation then substitute it in the differential equation and solve it for H(a).
- Method 2) First, solve the differential equation for V(a) then substitute it in the constraint equation and solve it for H(a).
- Method 3) First, find H(a) from the constraint equation then substitute it in the differential equation and solve it for V(a).
- Mehod 4) First, solve the differential equation for H(a) then substitute it in the constraint equation and solve it for V(a).

Results are different representations of the same solution. However they may have different singular cases which require special attention. To reach complete set of solutions we perform all four methods. Moreover, most of the solutions include integrals of different functions and sometimes integration techniques can be inadequate . For this reason one of the form of solutions can be superior to others. We would like to remind that the ρ equation is equivalent to the constraint equation and the first order differential equation is equivalent to the ϕ equation. We will use both names.

3.3.1. Method 1

<u>3.3.1.1. Non-Singular Case.</u> Firstly we find V(a) from the constraint equation

$$V(a) = \frac{3}{4\omega}\phi^2(a)(H^2(a) + \frac{k}{a^2}) - \frac{1}{2}(\phi'(a)H(a)a)^2 + \frac{3}{2\omega}\phi\phi'H^2a + \rho(a).$$
(3.14)

Then the ϕ equation becomes

$$H'(a) + \frac{a[(1+2\omega)\phi^{'2}(a) + \phi(a)\phi^{''}(a)]}{\phi(a)(\phi(a) + a\phi^{'}(a))}H(a) = \frac{2\omega\rho^{'}a^{3} + 3k\phi^{2}(a)}{3a^{3}\phi(a)(\phi(a) + a\phi^{'}(a))H(a)},$$
 (3.15)

which is the Bernoulli equation. Thus we apply change of variable $\gamma(a) = H^2(a)$ and we obtain

$$\gamma'(a) + \frac{2a[(1+2\omega)\phi^{'2}(a) + \phi(a)\phi^{''}(a)]}{\phi(a)(\phi(a) + a\phi^{'}(a))}\gamma(a) = \frac{2(2\omega\rho'a^3 + 3k\phi^2(a))}{3a^3\phi(a)(\phi(a) + a\phi^{'}(a))}.$$
 (3.16)

Its solution is found as

$$\gamma(a) = \frac{exp\left[-\int_{a_{in}}^{a} P(a')da'\right]}{\left[\phi(\phi + a\phi')\right]^{2}} \Big\{\int_{a_{in}}^{a} exp\left[\int_{a_{in}}^{a'} P(a'')da''\right]Q(a')da' + \tilde{\gamma}(a_{in})\Big\}, \quad (3.17)$$

$$P(a) = \frac{2[2\omega a\phi'^{2}(a) - 3\phi(a)\phi'(a)]}{\phi(a)(\phi(a) + a\phi'(a))}, \quad Q(a) = \frac{2(2\omega\rho'a^{3} + 3k\phi^{2}(a))\left[\phi(a)(\phi(a) + a\phi'(a))\right]}{3a^{3}}.$$
(3.18)

$$\tilde{\gamma}(a_{in}) = \gamma(a_{in}) \Big[\phi(a_{in}) \Big(\phi(a_{in}) + a_{in} \phi'(a_{in}) \Big) \Big]^2.$$
(3.19)

Hence

$$H(a) = \sqrt{\gamma(a)},\tag{3.20}$$

where $\gamma(a)$ is given by (3.17) (Details of this calculation is given in the Appendix B.3.). One should choose $\phi(a)$ such that $\gamma(a)$ is positive. Thus H(a) will be a real function. However if one goes further by assigning specific form to the Hubble function and tries to solve (3.17) for $\phi(a)$, she will end up with a second order, variable coefficient, nonlinear differential equation for whose solution there is no known method. For this reason we will continue with trial $\phi(a)$ functions in our examples.

<u>3.3.1.2. Singular Case.</u> The denominator of the function P(a) which is used to formulate the Hubble function becomes singular when $\phi(a) = F/a$. Hence this case needs special attention. V(a) is obtained form the ρ equation,

$$V(a) = -\rho(a) + \frac{F^2 \Big[3k - (3 + 2\omega)a^2 H^2(a) \Big]}{4\omega a^4}.$$
(3.21)

Then the ϕ equation turns into the following form

$$\frac{3F^2 \left[k - (3 + 2\omega)a^2 H^2(a) \right] + 2\omega a^5 \rho'(a)}{2F\omega a^3} = 0.$$
(3.22)

The Hubble function is found as

$$H(a) = \sqrt{\frac{3F^2k + 2\omega a^5\rho'(a)}{3(3+2\omega)F^2a^2}}.$$
(3.23)

Hence the final form of potential is given by

$$V(a) = -\rho(a) - \frac{a}{6}\rho'(a) + \frac{kF^2}{2\omega a^4}.$$
(3.24)

If one inserts the energy density of an ordinary matters which is proportional to $1/a^n$ results are

$$H(a) = \sqrt{\frac{3F^2k - 2n\omega\rho_n a^{4-n}}{3(3+2\omega)F^2a^2}},$$
(3.25)

$$V(a) = \frac{F^2 k}{2\omega a^4} + \frac{(n-6)\rho_n}{a^n},$$
(3.26)

$$V(\phi) = \frac{F^2 k}{2\omega} \phi^4 + (n-6)\rho_n \phi^n.$$
 (3.27)

Except the radiation dominated universe and dark energy dominated universe all these matters results in an imaginary form of the Hubble function and negative potential values as easily recognized from (3.25-3.26). In addition radiation dominated universe corresponds to $V = \lambda \phi^4/4$ which will be studied in chapter 3 of this thesis.

3.3.2. Method 2

<u>3.3.2.1. Non-Singular Case.</u> In the second method we start with the ϕ equation. By using it V(a) is found as

$$V(a) = -\int_{a_{in}}^{a} \frac{\phi' \left\{ -3\phi \left[k + a'^{2}H(2H + a'H') \right] + 2a'^{3}\omega H \left[(4H + a'H')\phi' + a'H\phi'' \right] \right\}}{2a'^{2}\omega} da' + V(a_{in}).$$
(3.28)

Then we insert V(a) in ρ equation and we take the derivative of both sides with respect to a. We obtain the following equation

$$H'(a) + \frac{a[(1+2\omega)\phi'^{2}(a) + \phi(a)\phi''(a)]}{\phi(a)(\phi(a) + a\phi'(a))}H(a) = \frac{2\omega\rho'(a)a^{3} + 3k\phi^{2}(a)}{3a^{3}\phi(a)(\phi(a) + a\phi'(a))H(a)}.$$
 (3.29)

This equation is equivalent to (3.15). Its solution has been already presented in (3.17-3.20). Although forms of potentials which are given by (3.24) and (3.28) are different, one can show that they are equal to each other up to a constant.

We have just shown that results of method 1 and method 2 are the same, thus there is no need to repeat calculations for singular case. In this method we end up with two integration constants. One of them should be chosen such that when final forms of formulas are inserted in the constraint equation it must be satisfied.

3.3.3. Method 3

<u>3.3.3.1. Non-Singular Case.</u> First we find H(a) from the ρ equation

$$H(a) = \sqrt{\frac{-3k\phi^2(a) + 4\omega a^2 \left[\rho(a) + V(a)\right]}{a^2 \left[3\phi^2(a) + 6a\phi(a)\phi'(a) - 2\omega a^2 \phi'^2\right]}}.$$
(3.30)

Then the ϕ equation turns into the following form

$$V'(a) + P(a)V(a) = Q(a), (3.31)$$

where

$$P(a) = \left\{ 2\phi'(a) \left[-6\phi^{3}(a) + (3+14\omega)a^{2}\phi(a)\phi'^{2}(a) - 2\omega(1+2\omega)a^{3}\phi'^{3}(a) + a\phi^{2}(a)((-6+8\omega)\phi'(a) + (3+2\omega)a\phi''(a)) \right] \right\} / \left\{ \phi(a)(\phi(a) + a\phi'(a)) \left[3\phi^{2}(a) + 6a\phi(a)\phi'(a) - 2\omega a^{2}\phi'^{2}(a) \right] \right\},$$
(3.32)

and

$$Q(a) = -\left\{\phi'(a)\left[-36\omega^{2}a^{4}\phi(a)\rho'(a)\phi'^{2}(a) + 8\omega^{3}a^{5}\rho'(a)\phi'^{3}(a) - 12\omega a^{2}\phi^{2}(a)\phi'(a)((-3+\omega)a\rho'(a) + k(3+2\omega)\phi'^{2}(a)) + 9a\phi^{3}(a)(2\omega a\rho'(a) + 3k(3+2\omega)\phi'^{2}(a)) + 9k(3+2\omega)\phi^{4}(a)(3\phi'(a) + a\phi'') + 12\omega a\rho(a)(6\phi^{3}(a) - (3+14\omega)a^{2}\phi(a)\phi'^{2}(a) + 2\omega(1+2\omega)a^{3}\phi'^{3}(a) - a\phi^{2}(a)((-6+8\omega)\phi'(a) + (3+2\omega)a\phi''(a)))\right]\right\}$$

$$+ \left\{6\omega a\phi(a)(\phi(a) + a\phi'(a))\left[-3\phi^{2}(a) - 6a\phi(a)\phi'(a) + 2\omega a^{2}\phi'^{2}(a)\right]\right\}.$$
(3.33)

Hence potential is

tial become undefined.

$$V(a) = exp\Big(-\int_{a_{in}}^{a} P(a')da'\Big)\Big\{\int_{a_{in}}^{a} exp\Big(\int_{a_{in}}^{a'} P(a'')da''\Big)Q(a')da' + V(a_{in})\Big\}.$$
 (3.34)

<u>3.3.3.2. Singular Cases.</u> The solution for method 3 has three singular cases; I) $\phi(a) = 0$, II) $\phi(a) + a\phi'(a) = 0$, III) $-3\phi^2(a) - 6a\phi(a)\phi'(a) + 2\omega a^2 \phi'^2(a) = 0$. For all of them both functions, P(a) and Q(a) which are used to formulate the poten-

First case is not acceptable because Brans-Dicke theory collapses for $\phi(a) = 0$. Second case has been already investigated in Method 1. We are left with the last one. We write $\phi(a) = Fexp(\int \alpha(a)da)$. Then singularity III becomes

$$2\omega a^2 \alpha^2 - 6a\alpha - 3 = 0. (3.35)$$

Thus

$$\alpha(a) = \frac{3 \pm \sqrt{9 + 6\omega}}{2\omega a}, \qquad \phi(a) = Fa^{(3\pm\sqrt{9+6\omega})/(2\omega)}.$$
(3.36)

Then the potential is found from the ρ equation

$$V(a) = \frac{3F^2k}{4\omega} a^{(3-2\omega \pm \sqrt{9+6\omega})/\omega} - \rho(a).$$
(3.37)

After writing $H(a) = \sqrt{\gamma(a)}$, ϕ equation becomes

$$\gamma'(a) + P(a)\gamma(a) = Q(a).$$
 (3.38)

For $\alpha(a) = \frac{3 + \sqrt{9 + 6\omega}}{2\omega a}$,

$$P(a) = \frac{2(3 + 3\omega + \sqrt{9 + 6\omega})}{\omega a},$$
(3.39)

$$Q(a) = \frac{4\omega a^{-(3+3\omega+\sqrt{9+6\omega})/\omega}}{3F^2(3+2\omega+\sqrt{9+6\omega})} \Big[3F^2 k a^{(3+\sqrt{9+6\omega})/\omega} + 2\omega a^3 \rho'(a) \Big].$$
(3.40)

Then

$$\gamma(a) = \frac{1}{a^{2(3+3\omega+\sqrt{9+6\omega})/\omega}} \Big\{ b \int_{a_{in}}^{a} a^{\prime(3+3\omega+\sqrt{9+6\omega})/\omega} \Big[3F^2 k a^{\prime(3+\sqrt{9+6\omega})/\omega} + 2\omega a^{\prime 3} \rho^{\prime} \Big] da^{\prime} + \tilde{\gamma}(a_{in}) \Big\},$$
(3.41)

$$b = \frac{4\omega}{3F^2(3 + 2\omega + \sqrt{9 + 6\omega})}, \qquad \tilde{\gamma}(a_{in}) = a_{in}^{2(3 + 3\omega + \sqrt{9 + 6\omega})/\omega} \gamma(a_{in}). \tag{3.42}$$

For $\alpha(a) = \frac{3 - \sqrt{9 + 6\omega}}{2\omega a}$,

$$P(a) = \frac{2(3+3\omega - \sqrt{9+6\omega})}{\omega a},$$
(3.43)

$$Q(a) = \frac{4\omega a^{-3(1+\omega)/\omega}}{3F^2(3+2\omega-\sqrt{9+6\omega})} \Big[3F^2 k a^{3/\omega} + 2\omega a^{(3\omega+\sqrt{9+6\omega})/\omega} \rho'(a) \Big].$$
(3.44)

Then

$$\gamma(a) = \frac{1}{a^{2(3+3\omega-\sqrt{9+6\omega})/\omega}} \Big\{ b \int_{a_{in}}^{a} a^{'(3+3\omega-2\sqrt{9+6\omega})/\omega} \Big[3F^2 k a^{'3/\omega} + 2\omega a^{'(3\omega+\sqrt{9+6\omega})/\omega} \rho' \Big] da^{'} \Big\}$$

$$+ \tilde{\gamma}(a_{in}) \Big\}, \tag{3.45}$$

$$b = \frac{4\omega}{3F^2(3 + 2\omega - \sqrt{9 + 6\omega})}, \qquad \tilde{\gamma}(a_{in}) = a_{in}^{2(3 + 3\omega - \sqrt{9 + 6\omega})/\omega} \gamma(a_{in}). \tag{3.46}$$

At the next step we examine nature of the Hubble function. First, we take

$$\phi(a) = Fa^{(3+\sqrt{9+6\omega})/(2\omega)}, \qquad \rho = \frac{\rho_n}{a^n}.$$
 (3.47)

Hubble function and the potential are found as

$$H(a) = \frac{1}{a^{(3+3\omega+\sqrt{9+6\omega})/\omega}} \left\{ \frac{2\omega^2 a^{(3-(n-4)\omega+\sqrt{9+6\omega})/\omega}}{3F^2(3+2\omega+\sqrt{9+6\omega})^2[3+(6-n)\omega+\sqrt{9+6\omega}]} \times \left[-4(3+2\omega+\sqrt{9+6\omega})n\omega\rho_n a^2 + 3F^2k\left(3+(6-n)\omega+\sqrt{9+6\omega}\right)a^{(3+n\omega+\sqrt{9+6\omega})/\omega}\right] + \tilde{\gamma}(ain) \right\}^{1/2}, \quad (3.48)$$
$$\tilde{\gamma}(a_{in}) = a_{in}^{2(3+3\omega+\sqrt{9+6\omega})/\omega}\gamma(a_{in}) - \left\{ \frac{2\omega^2 a_{in}^{(3-(n-4)\omega+\sqrt{9+6\omega})/\omega}}{3F^2(3+2\omega+\sqrt{9+6\omega})^2[3+(6-n)\omega+\sqrt{9+6\omega}]} \right\}$$

$$\times \left[-4(3+2\omega+\sqrt{9+6\omega})n\omega\rho_n a_{in}^2 + 3F^2k\left(3+(6-n)\omega+\sqrt{9+6\omega}\right)a_{in}^{(3+n\omega+\sqrt{9+6\omega})/\omega} \right] \right\},$$
(3.49)

$$V(a) = -\frac{\rho_n}{a^n} + \frac{3F^2k}{4\omega a^{(2\omega-3-\sqrt{9+6\omega})/\omega}}.$$
(3.50)

The term containing energy density in H(a) has a negative coefficient for 0 < n < 6and $\omega \gg 1$.

Then we take

$$\phi(a) = Fa^{(3-\sqrt{9+6\omega})/(2\omega)}, \qquad \rho = \frac{\rho_n}{a^n}.$$
 (3.51)

Hubble function and the potential are found as

$$H(a) = \frac{1}{a^{(3+3\omega-\sqrt{9+6\omega})/\omega}} \left\{ \frac{2\omega^2 a^{(3-(n-4)\omega-2\sqrt{9+6\omega})/\omega}}{3F^2(-3-2\omega+\sqrt{9+6\omega})^2[-3+(n-6)\omega+\sqrt{9+6\omega}]} \times \left[4(3+2\omega-\sqrt{9+6\omega})n\omega\rho_n a^{(2\omega+\sqrt{9+6\omega})/\omega} + 3F^2k\left(-3+(n-6)\omega+\sqrt{9+6\omega}\right)a^{3/\omega+n}\right] + \tilde{\gamma}(ain) \right\}^{1/2},$$
(3.52)

$$\tilde{\gamma}(a_{in}) = a_{in}^{2(3+3\omega-\sqrt{9+6\omega})/\omega} \gamma(a_{in}) - \left\{ \frac{2\omega^2 a_{in}^{(3-(n-4)\omega-2\sqrt{9+6\omega})/\omega}}{3F^2(-3-2\omega+\sqrt{9+6\omega})^2[-3+(n-6)\omega+\sqrt{9+6\omega}]} \times \left[4(3+2\omega-\sqrt{9+6\omega})n\omega\rho_n a_{in}^{(2\omega+\sqrt{9+6\omega})/\omega} + 3F^2k \left(-3+(n-6)\omega+\sqrt{9+6\omega} \right) a_{in}^{3/\omega+n} \right] \right\},$$
(3.53)

$$V(a) = -\frac{\rho_n}{a^n} + \frac{3F^2\kappa}{4\omega a^{(2\omega-3+\sqrt{9+6\omega})/\omega}}.$$
(3.54)

The term containing energy density in H(a) has a negative coefficient for 0 < n < 6and $\omega \gg 1$.

3.3.4. Method 4

<u>3.3.4.1. Non-Singular Case.</u> Substitution of $H(a) = \sqrt{\gamma(a)}$ into ϕ equation results in the following form,

$$\gamma'(a) + \tilde{P}(a)\gamma = \tilde{Q}(a), \qquad (3.55)$$

$$\tilde{P}(a) = \frac{4\left[-3\phi(a) + \omega a(4\phi'(a) + a\phi''(a))\right]}{a(-3\phi(a) + 2\omega a\phi'(a))}, \qquad \tilde{Q}(a) = \frac{-4\omega a^2 V'(a) + 6k\phi(a)\phi'(a)}{a^3\phi'(a)(-3\phi(a) + 2\omega a\phi'(a))}. \qquad (3.56)$$

Hence

$$\gamma(a) = exp\Big(-\int_{a_{in}}^{a} \tilde{P}(a')da'\Big)\Big\{\int_{a_{in}}^{a} exp\Big(\int_{a_{in}}^{a'} \tilde{P}(a'')da''\Big)\tilde{Q}(a')da' + \gamma(a_{in})\Big\}, \quad (3.57)$$
$$H(a) = \sqrt{\gamma(a)}. \quad (3.58)$$

After plugging H(a) into ρ equation we multiply both sides by

$$\frac{exp\left(\int_{a_{in}}^{a}\tilde{P}(a')da'\right)}{3\phi^{2}(a)+6a\phi(a)\phi'(a)-2\omega a^{2}\phi'^{2}}.$$
(3.59)

Then we take derivative of both sides with respect a. The resulting equation is equivalent to

$$V'(a) + P(a)V(a) = Q(a), (3.60)$$

where coefficients P(a) and Q(a) are given by (3.32) and (3.33) respectively. Thus solution is given by (3.34). In this method we end up with two integration constants. One of them should be chosen such that when final forms of formulas are inserted in the constraint equation it must be satisfied.

<u>3.3.4.2.</u> Singular Cases. This method has five singularities in its formulation. Three of them are the same singularities given in Method 3. Other two of them are seen in functions $\tilde{P}(a)$ and $\tilde{Q}(a)$

$$\mathbf{I})\phi'(a) = 0,$$

 $\mathrm{II})-3\phi(a) + 2\omega a\phi'(a) = 0.$

First one makes Brans-Dicke cosmology equivalent to Einstein cosmology which have been studied in chapter 2. The second one implies

$$\phi(a) = Fa^{3/(2\omega)}.$$
 (3.61)

Then ϕ equation becomes

$$\frac{a^{-3/\omega-2}}{12F\omega^2} \left\{ -9F^2 a^{3/\omega} \left[2k\omega - (3+2\omega)a^2 H^2(a) \right] + 8\omega^3 a^3 V'(a) \right\} = 0.$$
(3.62)

The Hubble function is found easily

$$H(a) = \frac{\sqrt{18F^2k\omega a^{3/\omega} - 8\omega^3 a^3 V'(a)}}{3\sqrt{(3+2\omega)F^2 a^{3/\omega+2}}}.$$
(3.63)

Hence the ρ equation turns into the following form

$$V'(a) + P(a)V(a) = Q(a),$$
 (3.64)

$$P(a) = \frac{3}{\omega a},$$

$$Q(a) = \frac{3(3F^2ka^{3/\omega} - 2\omega a^2\rho(a))}{2\omega^2 a^3}.$$
(3.65)

Its solution is

$$V(a) = \frac{1}{a^{3/\omega}} \left\{ \int_{a_{in}}^{a} a^{'3/\omega} Q(a^{'}) da^{'} + \tilde{V}(a_{in}) \right\},$$

$$\tilde{V}(a_{in}) = a_{in}^{3/\omega} V(a_{in}).$$
(3.66)

To investigate the nature of the Hubble function and the potential for usual forms of energy-matters we plug $\rho = \rho_n/a^n$. Thus we have

$$H(a) = \omega \left\{ \frac{2}{3(3+2\omega)} \left[\frac{4a_{in}^{3/\omega} V(a_{in})}{F^2 a^{6/\omega}} - \frac{3k}{(\omega-3)a^2} + \frac{4n\omega\rho_n}{(n\omega-3)F^2 a^{3/\omega+n}} \right] \right\}^{1/2}, \quad (3.67)$$

$$V(a) = \frac{V(a_{in})}{a^{3/\omega}} + \frac{3\rho_n}{(n\omega - 3)a^n} + \frac{9F^2k}{4\omega(3 - \omega)a^{-3/\omega + 2}},$$
(3.68)

$$\tilde{V}(a_{in}) = a_{in}^{3/\omega} V(a_{in}) - \frac{3\rho_n}{(n\omega - 3)a_{in}^n} - \frac{9F^2k}{4\omega(3 - \omega)a_{in}^{-3/\omega + 2}}.$$
(3.69)

Energy density related term in H(a) has a positive coefficient therefore the Hubble function is always real.

3.4. Solution for Given V(a) and $\phi(a)$

Special case of potential in the scalar-tensor theory is important. For this reason formulation of this case is significant. We establish the solution for our set of the constraint equation and the first order differential equation by two following different methods:

- Method 1) First, solve the ϕ equation for H(a) then plug it into the constraint equation to obtain $\rho(a)$.
- Method 2) First, find H(a) from the constraint equation then plug it in to the ϕ equation and solve it for $\rho(a)$.

3.4.1. Method1

<u>3.4.1.1. Non-Singular Case.</u> We start our calculations by changing the dependent variable as $H(a) = \sqrt{\gamma(a)}$ in ϕ equation. Hence it turns into the following form

$$\gamma'(a) + P(a)\gamma(a) = Q(a),$$
(3.70)

$$P(a) = \frac{4\left[-3\phi(a) + \omega a(4\phi'(a) + a\phi''(a))\right]}{a(-3\phi(a) + 2\omega a\phi'(a))},$$
(3.71)

$$Q(a) = \frac{-4\omega a^2 V'(a) + 6k\phi(a)\phi'(a)}{a^3\phi'(a)(-3\phi(a) + 2\omega a\phi'(a))}.$$
(3.72)

Its solution is given by

$$\gamma(a) = exp\Big(-\int_{a_{in}}^{a} P(a')da'\Big)\Big\{\int_{a_{in}}^{a} exp\Big(\int_{a_{in}}^{a'} P(a'')da''\Big)Q(a')da' + \gamma(a_{in})\Big\}.$$
 (3.73)

Energy density is obtained by inserting $H(a) = \sqrt{\gamma(a)}$ and V(a) in the ρ equation.

<u>3.4.1.2. Singular Case.</u> When $-3\phi(a) + 2\omega a \phi'(a) = 0$, previous solution becomes singular. This is the singularity of Section 3.3.4. Thus $\phi(a) = F a^{3/(2\omega)}$ and ϕ equation is equal to (3.62) so H(a) is found in (3.63). Then energy density is found as

$$\rho(a) = \frac{3F^2ka^{3/\omega-2}}{2\omega} - V(a) - \frac{\omega aV'(a)}{3}.$$
(3.74)

When we inspect the results for $V = \frac{V_n}{a^n}$ we obtain

$$H(a) = \frac{1}{3\sqrt{3+2\omega}} \sqrt{\frac{18k\omega}{a^2} + \frac{8n\omega^3 V_n}{F^2 a^{3/\omega+n}}},$$
(3.75)

$$\rho(a) = \frac{3F^2ka^{3/\omega-2}}{2\omega} + \frac{(n\omega-3)V_n}{3a^n}.$$
(3.76)

3.4.2. Method 2

<u>3.4.2.1. Non-Singular Case.</u> First we find H(a) from the ρ equation

$$H(a) = \sqrt{\frac{-3k\phi^2(a) + 4\omega a^2 \left[\rho(a) + V(a)\right]}{a^2 \left[3\phi^2(a) + 6a\phi(a)\phi'(a) - 2\omega a^2 \phi'^2\right]}}.$$
(3.77)

Then ϕ equation turns into the following form

$$\rho'(a) + P(a)\rho(a) = Q(a), \qquad (3.78)$$

where

$$P(a) = \left\{ 6 \left[6\phi^{3}(a) - (3+14\omega)a^{2}\phi(a)\phi^{'2}(a) + 2(1+2\omega)\omega a^{3}\phi^{'3}(a) - a\phi^{2}(a)((-6+8\omega)\phi^{'}(a) + (3+2\omega)a\phi^{''}(a)) \right] \right\} / \left\{ a(-3\phi(a) + 2\omega a\phi^{'}(a)) \left[-3\phi^{2}(a) - 6a\phi(a)\phi^{'}(a) + 2\omega a^{2}\phi^{'2}(a) \right] \right\},$$
(3.79)

$$Q(a) = -\left\{3\left[4\omega a^{3}\phi(a)(-(3+14\omega)V(a)+\omega aV'(a))\phi'^{3}(a)+8\omega^{2}(1+2\omega)a^{4}V(a)\phi'^{4}(a) + 3a\phi^{3}(a)\phi'(a)(8\omega V(a)-6\omega aV'(a)+3k(3+2\omega)\phi'^{2}(a)) + 4\omega a^{2}\phi^{2}(a)\phi'(a)\left(\phi'(a)((6-8\omega)V(a)+(-3+\omega)aV'(a)-k(3+2\omega)\phi'^{2}(a)) - (3+2\omega)aV(a)\phi''(a)\right) + 3\phi^{4}(a)\left(-2\omega aV'(a)+k(3+2\omega)\phi'(a)(3\phi'(a)+a\phi''(a))\right)\right]\right\} + 3\phi^{4}(a)\left(-2\omega aV'(a)+k(3+2\omega)\phi'(a)(3\phi'(a)+a\phi''(a))\right)\right]\right\} / \left\{2\omega a^{2}\phi'(a)(-3\phi(a)+2\omega a\phi'(a))\left[-3\phi^{2}(a)-6a\phi(a)\phi'(a)+2\omega a^{2}\phi'^{2}(a)\right]\right\}.$$

$$(3.80)$$

Hence energy density is found as

$$\rho(a) = exp\Big(-\int_{a_{in}}^{a} P(a')da'\Big)\Big\{\int_{a_{in}}^{a} exp\Big(\int_{a_{in}}^{a'} P(a'')da''\Big)Q(a')da' + \rho(a_{in})\Big\}.$$
 (3.81)

<u>3.4.2.2. Singular Cases.</u> This method has two singular cases for non-constant scalar field.

I)
$$-3\phi(a) + 2\omega a\phi'(a) = 0,$$

II) $-3\phi^2(a) - 6a\phi(a)\phi'(a) + 2\omega a^2 \phi'^2(a) = 0.$

First case implies $\phi(a) = Fa^{3/(2\omega)}$. Thus the ρ equation becomes

$$\frac{3F^2 a^{3/\omega-2} \Big[2k\omega + (3+2\omega)a^2 H^2(a) \Big]}{8\omega^2} - V(a) = \rho(a).$$
(3.82)

The Hubble function is found as

$$H(a) = \sqrt{\frac{\omega}{3+2\omega}} \sqrt{-\frac{2k}{a^2} + \frac{8\omega}{3F^2 a^{3/\omega}}(\rho(a) + V(a))}.$$
 (3.83)

Then the ϕ equation becomes

$$-\frac{3Fk}{\omega}a^{3/(2\omega)-2} + \frac{2a^{-3/(2\omega)}}{3F} \Big[3(\rho(a) + V(a)) + \omega aV'(a)\Big],$$
(3.84)

and energy density is found as

$$\rho(a) = \frac{3F^2k}{2\omega}a^{3/\omega-2} - V(a) - \frac{\omega aV'(a)}{3}.$$
(3.85)

Examination of this singularity with $V(a) = \frac{V_n}{a^n}$ gives us

$$H(a) = \sqrt{\frac{\omega}{3+2\omega}} \sqrt{\frac{2k}{a^2} + \frac{8n\omega^2 V_n}{9F^2 a^{3/\omega+n}}},$$
(3.86)

$$\rho(a) = \frac{3F^2k}{2\omega}a^{3/\omega-2} + \frac{(n\omega-3)V_n}{3a^n}.$$
(3.87)

Thus we have the energy density in usual form.

Second singularity has been already seen in Section 3.3.3.2. It implies $\phi(a) = Fa^{(3\pm\sqrt{9+6\omega})/(2\omega)}$.

Then the ρ equation has the following form

$$\frac{3F^2k}{4\omega}a^{(3-2\omega\pm\sqrt{9+6\omega})/\omega} - V(a) = \rho(a).$$
(3.88)

After writing $H(a) = \sqrt{\gamma(a)}$, ϕ equation becomes

$$\gamma'(a) + P(a)\gamma(a) = Q(a).$$
 (3.89)

For $\phi(a) = a^{(3+\sqrt{9+6\omega})/(2\omega)}$,

$$P(a) = \frac{2(3 + 3\omega + \sqrt{9 + 6\omega})}{\omega a},$$
(3.90)

$$Q(a) = \frac{a^{-(3+3\omega+\sqrt{9+6\omega})/\omega}}{3\sqrt{3+2\omega}F^2} \Big[6\sqrt{3}F^2ka^{(3+\sqrt{9+6\omega})/\omega} + 4\omega(\sqrt{3}-\sqrt{3+2\omega})a^3V'(a) \Big].$$
(3.91)

Then

$$\gamma(a) = \frac{1}{a^{2(3+3\omega+\sqrt{9+6\omega})/\omega}} \Big\{ b \int_{a_{in}}^{a} a^{\prime(3+3\omega+\sqrt{9+6\omega})/\omega} \Big[6\sqrt{3}F^{2}ka^{\prime(3+\sqrt{9+6\omega})/\omega} + 4\omega(\sqrt{3}-\sqrt{3+2\omega})a^{\prime 3}V^{\prime}(a^{\prime}) \Big] da^{\prime} + \tilde{\gamma}(a_{in}) \Big\},$$
(3.92)

$$b = \frac{1}{3\sqrt{3+2\omega}F^2}, \qquad \tilde{\gamma}(a_{in}) = a_{in}^{2(3+3\omega+\sqrt{9+6\omega})/\omega}\gamma(a_{in}).$$
(3.93)

For $\phi(a) = a^{(3-\sqrt{9+6\omega})/(2\omega)}$,

$$P(a) = \frac{2(3 + 3\omega - \sqrt{9 + 6\omega})}{\omega a},$$
(3.94)

$$Q(a) = \frac{a^{-3(1+\omega)/\omega}}{3F^2(3+2\omega-\sqrt{9+6\omega})} \Big[6F^2k(3-\sqrt{9+6\omega})a^{3/\omega} - 8\omega^2 a^{(3\omega+\sqrt{9+6\omega})/\omega}V'(a) \Big].$$
(3.95)

Then

$$\gamma(a) = \frac{1}{a^{2(3+3\omega-\sqrt{9+6\omega})/\omega}} \Big\{ b \int_{a_{in}}^{a} a'^{(3+3\omega-2\sqrt{9+6\omega})/\omega} \Big[6F^2 k (3-\sqrt{9+6\omega}) a'^{3/\omega} + 8\omega^2 a'^{(3\omega+\sqrt{9+6\omega})/\omega} V'(a') \Big] da' + \tilde{\gamma}(a_{in}) \Big\},$$
(3.96)

$$b = \frac{1}{3(3+2\omega-\sqrt{9+6\omega})F^2}, \qquad \tilde{\gamma}(a_{in}) = a_{in}^{2(3+3\omega-\sqrt{9+6\omega})/\omega}\gamma(a_{in}). \tag{3.97}$$

At the next step we examine nature of the Hubble function. First, we take

$$\phi(a) = Fa^{(3+\sqrt{9+6\omega})/(2\omega)}, \qquad V(a) = \frac{V_n}{a^n}.$$
 (3.98)

The Hubble function and the energy density are found as

$$H(a) = \left\{ \frac{\sqrt{3}k(\sqrt{3+2\omega}-\sqrt{3})}{2(3+2\omega)a^2} + \frac{8n\omega^3 V_n a^{-(3+n\omega+\sqrt{9+6\omega})/\omega}}{3(3+2\omega+\sqrt{9+6\omega}) \left[3+(6-n)\omega+\sqrt{9+6\omega}\right] F^2} + \tilde{\gamma}(a_{in})a^{-2(3+3\omega+\sqrt{9+6\omega})/\omega} \right\}^{1/2},$$
(3.99)

$$\tilde{\gamma}(a_{in}) = \left\{ \gamma(a_{in}) - \frac{\sqrt{3}k(\sqrt{3+2\omega} - \sqrt{3})}{2(3+2\omega)a_{in}^2} - \frac{8n\omega^3 V_n a_{in}^{-(3+n\omega+\sqrt{9+6\omega})/\omega}}{3(3+2\omega+\sqrt{9+6\omega})\left[3+(6-n)\omega+\sqrt{9+6\omega}\right]F^2} \right\} a_{in}^{2(3+3\omega+\sqrt{9+6\omega})/\omega},$$
(3.100)

$$\rho(a) = \frac{3F^2k}{4\omega a^{(2\omega-3-\sqrt{9+6\omega})/\omega}} - \frac{V_n}{a^n}.$$
(3.101)

Although the Hubble function is real for 0 < n < 6 and $\omega \gg 1$, the energy density always has a negative component.

Then we take

$$\phi(a) = Fa^{(3-\sqrt{9+6\omega})/(2\omega)}, \qquad V(a) = \frac{V_n}{a^n}.$$
 (3.102)

The Hubble function and the energy density are found as

$$H(a) = \left\{ -\frac{\sqrt{3}k(\sqrt{3+2\omega}+\sqrt{3})}{2(3+2\omega)a^2} + \frac{8n\omega^3 V_n a^{-(3+n\omega-\sqrt{9+6\omega})/\omega}}{3(3+2\omega-\sqrt{9+6\omega}) \left[3+(6-n)\omega-\sqrt{9+6\omega}\right] F^2} + \tilde{\gamma}(a_{in})a^{-2(3+3\omega-\sqrt{9+6\omega})/\omega} \right\}^{1/2},$$
(3.103)

$$\tilde{\gamma}(a_{in}) = \left\{ \gamma(a_{in}) + \frac{\sqrt{3}k(\sqrt{3+2\omega}+\sqrt{3})}{2(3+2\omega)a_{in}^2} - \frac{8n\omega^3 V_n a_{in}^{-(3+n\omega-\sqrt{9+6\omega})/\omega}}{3(3+2\omega-\sqrt{9+6\omega}) \left[3+(6-n)\omega-\sqrt{9+6\omega}\right] F^2} \right\} a_{in}^{2(3+3\omega-\sqrt{9+6\omega})/\omega},$$
(3.104)

$$\rho(a) = \frac{3F^2k}{4\omega a^{(2\omega-3+\sqrt{9+6\omega})/\omega}} - \frac{V_n}{a^n}.$$
(3.105)

Although the Hubble function is real for 0 < n < 6 and $\omega \gg 1$, the energy density always has a negative component.

3.5. Solution for Given H(a) and $\phi(a)$

The solution for specific form of the Hubble function can be crucial. To construct the solution one can follow two different paths:

- Method 1) First, find V(a) from the constraint equation then insert it into the ϕ equation and solve it for $\rho(a)$.
- Method 2) First, solve the ϕ equation for the V(a) then insert it into the constraint equation and obtain $\rho(a)$.

3.5.1. Method1

We find V(a) from the ρ equation

$$V(a) = -\rho(a) + \frac{3k\phi^2(a)}{4\omega a^2} + \left[3\phi^2(a) + 6a\phi(a)\phi'(a) - 2\omega a^2\phi'^2(a)\right]\frac{H^2(a)}{4\omega}.$$
 (3.106)

Then the ϕ equation turns into the following form

$$\rho'(a) = Q(a), \tag{3.107}$$

$$Q(a) = \frac{-3\phi^2(k - a^3HH') + a^3[3(1 + 2\omega)aH^2\phi'^2] + 3a^4H\phi(H'\phi' + H\phi'')}{2\omega a^3}.$$
 (3.108)

Thus

$$\rho(a) = \int_{a_{in}}^{a} Q(a')da' + \rho(a_{in}),$$

$$V(a) = -\left[\int_{a_{in}}^{a} Q(a')da' + \rho(a_{in})\right] \\
+ \frac{3k\phi^{2}(a)}{4\omega a^{2}} + \left[3\phi^{2}(a) + 6a\phi(a)\phi'(a) - 2\omega a^{2}\phi'^{2}(a)\right]\frac{H^{2}(a)}{4\omega}.$$
(3.109)
(3.109)

3.5.2. Method2

We solve the ϕ equation for the potential,

$$V(a) = -\int_{a_{in}}^{a} \phi' \Big\{ -3\phi \Big[k + a'^{2}H(2H + a'H') \Big] + 2\omega a'^{3}H \Big[(4H + a'H')\phi' + a'H\phi'' \Big] \Big\} da' + V(a_{in}).$$
(3.111)

We obtain energy density by substituting V(a) into the ρ equation

$$\rho(a) = \int_{a_{in}}^{a} \phi' \left\{ -3\phi \left[k + a'^{2}H(2H + a'H') \right] + 2\omega a'^{3}H \left[(4H + a'H')\phi' + a'H\phi'' \right] \right\} da' - V(a_{in}) + \frac{3k\phi^{2}(a)}{4\omega a^{2}} + \left[3\phi^{2}(a) + 6a\phi(a)\phi'(a) - 2\omega a^{2}\phi'^{2}(a) \right] \frac{H^{2}(a)}{4\omega}.$$
(3.112)

3.6. Single Component Universe

We investigate the universe with given energy density and the scalar field which are in the following form

$$\rho(a) = \frac{\rho_n}{a^n}, \qquad \phi(a) = \frac{F}{a^d}.$$
(3.113)

We have obtained the Hubble function and the potential function by applying the procedure explained in Section 3.3.1, we obtain the following formulas

$$H(a) = \sqrt{\frac{k}{\mathcal{B}a^2} - \frac{4n\rho_n\omega a^{2d-n}}{3\mathcal{C}F^2}} + \tilde{\gamma}(a_{in})a^{\mu}, \qquad (3.114)$$

$$V(a) = \frac{d^2(3+2\omega)F^2k}{2\mathcal{B}\omega a^{2+2d}} + \frac{\mathcal{A}\rho_n a^{-n}}{3\mathcal{C}} + \frac{\mathcal{F}\tilde{\gamma}(a_{in})F^2 a^{-2d+\mu}}{4\omega}, \qquad (3.115)$$

$$\tilde{\gamma}(a_{in}) = a_{in}^{-\mu} H^2(a_{in}) - \frac{k a_{in}^{-2-\mu}}{\mathcal{B}} + \frac{4n \rho_n \omega a_{in}^{2d-n-\mu}}{3\mathcal{C}F^2}, \qquad (3.116)$$

$$\mu = \frac{2d[1+2d(1+\omega)]}{-1+d},\tag{3.117}$$

$$\mathcal{A} = d\{3(-4+n) + 2d[-3 + (-6+n)\omega]\},\tag{3.118}$$

$$\mathcal{B} = [-1 + 2d(1 + d + d\omega)], \qquad (3.119)$$

$$\mathcal{C} = \{-n + d[4 + n + d(2 + 4\omega)]\},\tag{3.120}$$

$$\mathcal{F} = [-3 + 2d(3 + d\omega)], \tag{3.121}$$

$$p(a) = \frac{(n-3)\rho_n}{3a^n}.$$
(3.122)

We should choose d such that $\mathcal{B} < 0$ and $\mathcal{C} < 0$. Then we have real Hubble function. In addition if we also satisfy $\mu = 0$ at the same time this will have important physical implications which will be discussed soon. Thus

$$\mu = 0 \qquad \Rightarrow \qquad d = -\frac{1}{2(1+\omega)},\tag{3.123}$$

$$\mathcal{B} = -1 - \frac{1}{2(1+\omega)}, \qquad \mathcal{C} = -\frac{(3+2\omega)(1+n+n\omega)}{2(1+\omega)^2}.$$
 (3.124)

Then the Hubble function and potential are simplified to the following forms

$$H(a) = \sqrt{-\frac{2(1+\omega)k}{(3+2\omega)a^2} + \frac{8n(1+\omega)^2\omega\rho_n a^{-1/(1+\omega)-n}}{3(3+2\omega)[1+n(1+\omega)]F^2}} + \tilde{\gamma}(ain),$$
(3.125)

$$V(a) = \frac{(-3+n)\rho_n a^{-n}}{3(1+n(1+\omega))} - \frac{kF^2 a^{1/(1+\omega)-2}}{4\omega(1+\omega)} + \frac{(3+2\omega)(4+3\omega)F^2}{8\omega(1+\omega)^2}\tilde{\gamma}(a_{in})a^{1/(1+\omega)},$$
(2.107)

$$\tilde{\gamma}(a_{in}) = H^2(a_{in}) + \frac{2(1+\omega)k}{(3+2\omega)a_{in}^2} - \frac{8n(1+\omega)^2\omega\rho_n a_{in}^{-1/(1+\omega)-n}}{3(3+2\omega)[1+n(1+\omega)]F^2}.$$
(3.127)

The deceleration parameter is found by applying the chain rule

$$q(t) = \frac{d}{dt}(\frac{1}{H(t)}) - 1, \qquad (3.128)$$

$$q(a) = \frac{d}{da} (\frac{1}{H(a)}) a H(a) - 1, \qquad (3.129)$$

$$q(a) = -1 + \{2(1+\omega)[1+n(1+\omega)][-3F^2ka^{1/(1+\omega)+n} + 2n\omega\rho_n a^2]\} \\ /\{8n\omega(1+\omega)^2\rho_n a^2 + 3[1+n(1+\omega)]F^2[-2k(1+\omega) + (3+2\omega)\tilde{\gamma}(a_{in})a^2]a^{1/(1+\omega)+n}\}$$

$$(3.130)$$

When finding $q(a_{in})$ one should plug $\tilde{\gamma}(a_{in})$ which is given by (3.127)

$$q(a_{in}) = -1 + \frac{2(1+\omega)[-3F^2ka_{in}^{1/(1+\omega)+n} + 2n\omega\rho_n a_{in}^2]}{3(3+2\omega)F^2H^2(a_{in})a_{in}^2}.$$
 (3.131)

Since $a_{in} \ll 1$, $q(a_{in})$ is simplified further

$$\lim_{a \to a_{in}} q(a) = \begin{cases} -1 + \frac{4(1+\omega)n\omega\rho_n}{3(3+2\omega)F^2H^2(a_{in})}, & if \quad n+\frac{1}{1+\omega} > 2, \\ -1 + \frac{2(1+\omega)[-3F^2k+2n\omega\rho_n]}{3(3+2\omega)F^2H^2(a_{in})} & if \quad n+\frac{1}{1+\omega} = 2, \\ -1 - \frac{2(1+\omega)k}{(3+2\omega)H^2(a_{in})a_{in}^{-1/(1+\omega)-n+2}}, & if \quad 0 < n+\frac{1}{1+\omega} < 2. \end{cases}$$

When we set today values of the scale factor a = 1, ρ_n and F become today's values of the energy density and the scalar field respectively. Thus dynamics of the early universe depends on not only initial value of the Hubble function and the scale factor but also today's values of the energy density and scalar field. In addition behaviour of the universe also depends on the curvature parameter k for $0 < n + \frac{1}{1+\omega} \le 2$.

The fate of the universe can be described by the following number

$$\lim_{a \to \infty} q(a) = -1. \tag{3.132}$$

Now, we formulate a(t) for a spatially flat universe as

$$t = \int_{a_{in}}^{a} \frac{da'}{a'H(a')},$$
(3.133)

$$H(a) = \sqrt{\delta a^{-1/(1+\omega)-n} + \alpha}, \qquad \delta = \frac{8n(1+\omega)^2 \omega \rho_n}{3(3+2\omega)[1+n(1+\omega)]F^2}, \qquad \alpha = \tilde{\gamma}(a_{in}).$$
(3.134)

Thus

$$t = \frac{1}{\kappa\sqrt{\alpha}} ln \Big[\frac{a^{\kappa} + \sqrt{a^{2\kappa} + \frac{\delta}{\alpha}}}{a_{in}^{\kappa} + \sqrt{a_{in}^{2\kappa} + \frac{\delta}{\alpha}}} \Big], \qquad \kappa = \frac{1}{2} \Big[\frac{1 + n(1 + \omega)}{1 + \omega} \Big]. \tag{3.135}$$

Hence

$$a(t) = \left[\frac{\left(a_{in}^{\kappa} + \sqrt{a_{in}^{2\kappa} + \frac{\delta}{\alpha}}\right)e^{\kappa\sqrt{\alpha}t} + \left(a_{in}^{\kappa} - \sqrt{a_{in}^{2\kappa} + \frac{\delta}{\alpha}}\right)e^{-\kappa\sqrt{\alpha}t}}{2}\right]^{1/\kappa},$$
(3.136)

where δ , α and κ are given in (3.134-3.135). This formula indicates a very significant interpretation. When there is no constant energy density, the universe still expands exponentially because of nonzero initial conditions. We will investigate this phenomena in depth by adding constant term to the energy density in Sections 3.7 and 3.8

As we said at the beginning of Section 3.1 we have obtained perfect fluid equation of state

$$\nu = \frac{p(a)}{\rho(a)} = \frac{(n-3)}{3}.$$
(3.137)

Potential formula as a function of the scalar field are written as

$$V(\phi) = \frac{(-3+n)\rho_n}{3(1+n(1+\omega))} (\frac{\phi}{F})^{-2n(1+\omega)} - \frac{kF^2}{4\omega(1+\omega)} (\frac{\phi}{F})^{-2-4\omega} + \frac{(3+2\omega)(4+3\omega)}{8\omega(1+\omega)^2} \tilde{\gamma}(a_{in})\phi^2.$$
(3.138)

Most probably ϕ^2 term in the potential is responsible for late time accelerated expansion. This formula is simpler than the corresponding formula $V(\phi)$ in our previous study [129].

3.7. Early Universe, Dark Energy and Radiation

In this section the scalar field and the energy density is taken as

$$\phi(a) = \frac{F}{a^d}, \qquad d = -\frac{1}{2(1+\omega)}, \qquad \rho(a) = \frac{\rho_r}{a^4} + \Lambda.$$
 (3.139)

To see whether Λ contributes to H(a) we have performed calculations in all four different methods. Results are found in the following form

$$H(a) = \sqrt{-\frac{2(1+\omega)k}{(3+2\omega)a^2} + \frac{32(1+\omega)^2\omega\rho_r a^{-1/(1+\omega)-4}}{3(3+2\omega)(5+4\omega)F^2} + \alpha},$$
 (3.140)

$$V(a) = -\frac{F^2 k a^{1/(1+\omega)-2}}{4\omega(1+\omega)} + \beta a^{1/(1+\omega)} + \frac{\rho_r}{3(5+4\omega)a^4} - \Lambda.$$
 (3.141)

In method-1 we find

$$\alpha_1 = H^2(a_{in}) + \frac{2(1+\omega)k}{(3+2\omega)a_{in}^2} - \frac{32(1+\omega)^2\omega\rho_r a_{in}^{-1/(1+\omega)-4}}{3(3+2\omega)(5+4\omega)F^2},$$
(3.142)

$$\beta_1 = \frac{(3+2\omega)(4+3\omega)F^2}{8\omega(1+\omega)^2}\alpha_1.$$
(3.143)

In method-2 we find

$$\alpha_2 = \alpha_1, \qquad \beta_2 = \beta_1. \tag{3.144}$$

In this method to be able to satisfy the constraint equation with final formulas of V(a)and H(a) initial conditions must satisfy the following equation

$$V(a_{in}) = -\frac{F^2 k a_{in}^{1/(1+\omega)-2}}{4\omega(1+\omega)} + \beta_2 a_{in}^{1/(1+\omega)} + \frac{\rho_r}{3(5+4\omega)a_{in}^4} - \Lambda, \qquad (3.145)$$

which is consistent with the formula of V(a).

In method-3 we find

$$\alpha_3 = \frac{8\omega(1+\omega)^2 \tilde{V}(a_{in})}{(3+2\omega)(4+3\omega)F^2},$$
(3.146)

$$\tilde{V}(a_{in}) = V(a_{in})a_{in}^{-1/(1+\omega)} - \frac{\rho_r a_{in}^{-1/(1+\omega)-4}}{3(5+4\omega)} + \frac{F^2 k}{4\omega(1+\omega)a_{in}^2} + \Lambda a_{in}^{-1/(1+\omega)}, \quad (3.147)$$

$$\beta_3 = \frac{(3+2\omega)(4+3\omega)F^2}{8\omega(1+\omega)^2}\alpha_3.$$
(3.148)

Although at first sight it seems that $\alpha_1 \neq \alpha_3$, one can show that they are equal to each other by substituting a_{in} in H(a) and solving the equation for $V(a_{in})$.

In method-4 we find

$$\alpha_4 = \alpha_3, \qquad \beta_4 = \beta_3. \tag{3.149}$$

By the same reasoning in method-2, constraint equation implies the following condition on initial values

$$a_{in}^{1/(1+\omega)+3}H^{2}(a_{in}) - \frac{32\omega(1+\omega)^{2}\rho_{r}}{3(3+2\omega)(5+4\omega)F^{2}a_{in}} - \frac{8\omega(1+\omega)^{2}\tilde{V}(a_{in})a_{in}^{1/(1+\omega)+3}}{(3+2\omega)(4+3\omega)F^{2}} + \frac{2(1+\omega)ka_{in}^{1/(1+\omega)+1}}{3+2\omega} = 0.$$
(3.150)

 $V(a_{in})$ which is obtained by solving this equation is consistent with the formula of V(a).

The deceleration parameter is found as

$$q(a) = -1 + \frac{2(1+\omega)(5+4\omega)[-3F^2ka^{1/(1+\omega)+2}+8\omega\rho_r]}{32\omega(1+\omega)^2\rho_r + 3(5+4\omega)F^2[-2k(1+\omega)+(3+2\omega)\alpha_1a^2]a^{1/(1+\omega)+2}},$$
(3.151)

$$\alpha_1 = H^2(a_{in}) + \frac{2(1+\omega)k}{(3+2\omega)a_{in}^2} - \frac{32(1+\omega)^2\omega\rho_r a_{in}^{-1/(1+\omega)-4}}{3(3+2\omega)(5+4\omega)F^2}.$$
(3.152)

Thus initial value of the deceleration parameter is found

$$q(a_{in}) = -1 + \frac{2(1+\omega)[-3F^2ka_{in}^{1/(1+\omega)+2} + 8\omega\rho_r]}{3(3+2\omega)F^2H^2(a_{in})}.$$
(3.153)

The deceleration parameter which is given in (3.151) is same as the deceleration parameter found in single component universe without constant term, since α_1 given in (3.152) is equal to $\tilde{\gamma}(a_{in})$ given in (3.127) with n = 4.

The scale factor as a function of time is written as

$$a(t) = \left[\frac{\left(a_{in}^{\kappa} + \sqrt{a_{in}^{2\kappa} + \frac{\delta}{\alpha}}\right)e^{\kappa\sqrt{\alpha}t} + \left(a_{in}^{\kappa} - \sqrt{a_{in}^{2\kappa} + \frac{\delta}{\alpha}}\right)e^{-\kappa\sqrt{\alpha}t}}{2}\right]^{1/\kappa}, \qquad (3.154)$$

$$\kappa = \frac{1}{2} \left(\frac{5+4\omega}{1+\omega} \right), \qquad \delta = \frac{32(1+\omega)^2 \omega \rho_r}{3(3+2\omega)(5+4\omega)F^2}, \tag{3.155}$$

where $\alpha = \alpha_1$ and α_1 is given by (3.142).

Formula of the potential as function of the scalar field is found as

$$V(\phi) = -\frac{F^2 k}{4\omega(1+\omega)} (\frac{\phi}{F})^{-2-4\omega} + \beta (\frac{\phi}{F})^2 + \frac{\rho_r}{3(5+4\omega)} (\frac{\phi}{F})^{-8(1+4\omega)} - \Lambda, \qquad (3.156)$$

where $\beta = \beta_1$ which is given by (3.143). Pressure is found as in expected form

$$p(a) = \frac{\rho_r}{3a^4} - \Lambda. \tag{3.157}$$

3.8. Late Time Expansion of the Universe

3.8.1. Dark Energy Dominated Universe

We study in spatially flat universe where

$$\phi(a) = \frac{F}{a^d}, \qquad d = -\frac{1}{2(1+\omega)}, \qquad \rho(a) = \frac{\rho_m}{a^3} + \Lambda.$$
 (3.158)

Then related function is formulated as

$$H(a) = \sqrt{\frac{8\omega(1+\omega)^2 \rho_m}{(3+2\omega)(4+3\omega)F^2 a^{1/(1+\omega)+3}}} + \tilde{\gamma}(a_{in}), \qquad (3.159)$$

$$V(a) = \frac{(3+2\omega)(4+3\omega)}{8\omega(1+\omega)^2} F^2 \tilde{\gamma}(a_{in}) a^{1/(1+\omega)} - \Lambda, \qquad (3.160)$$

$$q(a) = -1 + \frac{4\omega(1+\omega)(4+3\omega)\rho_m}{8\omega(1+\omega)^2\rho_m + (3+2\omega)(4+3\omega)F^2\tilde{\gamma}(a_{in})a^{1/(1+\omega)+3}},$$
 (3.161)

$$\tilde{\gamma}(a_{in}) = H^2(a_{in}) - \frac{8(1+\omega)^2 \omega \rho_m a_{in}^{-1/(1+\omega)-3}}{(3+2\omega)(4+3\omega)F^2},$$
(3.162)

$$p(a) = -\Lambda. \tag{3.163}$$

The scale factor as function of time is found as

$$a(t) = \left[\frac{(a_{in}^{\kappa} + \sqrt{a_{in}^{2\kappa} + \frac{\delta}{\alpha}})e^{\kappa\sqrt{\alpha}t} + (a_{in}^{\kappa} - \sqrt{a_{in}^{2\kappa} + \frac{\delta}{\alpha}})e^{-\kappa\sqrt{\alpha}t}}{2}\right]^{1/\kappa},$$
(3.164)

$$\kappa = \frac{1}{2} \left(\frac{4+3\omega}{1+\omega} \right), \qquad \delta = \frac{8(1+\omega)^2 \omega \rho_m}{(3+2\omega)(4+3\omega)F^2}.$$
(3.165)

where $\alpha = \tilde{\gamma}(a_{in})$ and $\tilde{\gamma}(a_{in})$ is given by (3.162).

In Brans-Dicke theory effective gravitational constant is defined as $G_{eff} = \frac{\omega}{2\pi\phi^2}$. Thus it's present value becomes $G_0 = \frac{\omega}{2\pi F^2}$ where $F = \phi(t_0)$. In addition when we take observational value of Brans-Dicke parameter $\omega \gg 1$, the Hubble function becomes

$$H(a) = \sqrt{\frac{8\pi G_0 \rho_m}{3a^3} + \tilde{\gamma}(a_{in})}.$$
 (3.166)

By using following definitions of cosmological parameters

$$\Omega_{m,0} = \frac{8\pi G}{3H_0^2}\rho_m, \qquad \Omega_{\tilde{\gamma},0} = \frac{8\pi G}{3H_0^2}\rho_{\tilde{\gamma}}, \qquad \rho_{\tilde{\gamma}} = \frac{3\tilde{\gamma}(a_{in})}{8\pi G}, \qquad (3.167)$$

the Hubble function is written as

$$H(a) = H_0 \sqrt{\frac{\Omega_{m,0}}{a^3} + \Omega_{\tilde{\gamma},0}}.$$
 (3.168)

When one replace $\Omega_{\tilde{\gamma},0}$ with $\Omega_{\Lambda,0}$ in (3.166) we will have the Hubble function for the late epoch of the universe in standard cosmology where dark energy dominates.

We have already compared this model with observation of type Ia supernovae data in chapter 2. In our previous study we have obtained $H_0 = 71.80 \pm 0.22$, $\Omega_{\Lambda,0} = 0.715 \pm 0.012$, $\Omega_m = 0.285 \pm 0.012$ and $\chi^2/\nu = 0.990$ for absolute magnitude M = -19.30. Present value of deceleration parameter has been found as $q_0 = -0.572$ with this cosmological parameters.

On the other hand potential can be rewritten as

$$V(\phi) = \frac{(3+2\omega)(4+3\omega)}{8\omega(1+\omega)^2}\tilde{\gamma}(a_{in})\phi^2 - \Lambda.$$
(3.169)

Most probably ϕ^2 potential is responsible from the accelerated expansion.

3.8.2. Domain Wall Dominated Universe

In this section we study in spatially flat universe with scalar field and the energy density is given as

$$\phi(a) = \frac{F}{a^d}, \qquad d = -\frac{1}{2(1+\omega)}, \qquad \rho(a) = \frac{\rho_m}{a^3} + \frac{\rho_w}{a}.$$
 (3.170)

The Hubble function, potential, deceleration parameter and the pressure is formulated as

$$H(a) = \sqrt{\frac{8\omega(1+\omega)^2}{3(3+2\omega)F^2}} \left[\frac{3\rho_m}{(4+3\omega)a^3} + \frac{\rho_w}{(2+\omega)a}\right] a^{-1/(1+\omega)} + \tilde{\gamma}(a_{in}), \qquad (3.171)$$

$$V(a) = \frac{(3+2\omega)(4+3\omega)}{8\omega(1+\omega)^2} \tilde{\gamma}(a_{in}) F^2 a^{1/(1+\omega)} - \frac{2\rho_w}{3(2+\omega)a},$$

$$q(a) = -1 + \{4\omega(1+\omega)(2+\omega)(4+3\omega)(3\rho_m+\rho_w a^2)\}$$

$$/\{3(2+\omega)(3+2\omega)(4+3\omega)\tilde{\gamma}(a_{in})F^2 a^{1/(1+\omega)+3} + 8\omega(1+\omega)^2[3(2+\omega)\rho_m$$
(3.172)

$$+ (4 + 3\omega)\rho_w a^2]\},$$
 (3.173)

$$\tilde{\gamma}(a_{in}) = H^2(a_{in}) - \frac{8\omega(1+\omega)^2}{3(3+2\omega)F^2} \Big[\frac{3\rho_m}{(4+3\omega)a_{in}^3} + \frac{\rho_w}{(2+\omega)a_{in}} \Big] a_{in}^{-1/(1+\omega)}, \qquad (3.174)$$

$$p(a) = -\frac{2\rho_w}{3a}.$$
(3.175)

Now we take $\tilde{\gamma}(a_{in}) = 0$, to investigate whether cosmic domain walls cause accelerated expansion of the universe or not. We again apply the fundamental idea of the Brans-Dicke theory as we did in Section 3.7. Thus we take $G_0 = \frac{\omega}{2\pi F^2}$ and we take $\omega \gg 1$. Then the Hubble function is simplified to

$$H(a) = \sqrt{\frac{8\pi G_0}{3} (\frac{\rho_m}{a^3} + \frac{\rho_w}{a})}.$$
(3.176)

The relation between the scale factor and time is found as

$$t = \int_{a_{in}}^{a} \frac{da'}{a'H(a')},$$

$$t = \int_{a_{in}}^{a} \frac{da'}{a'\sqrt{\frac{8\pi G}{3}(\frac{\rho_m}{a'^3} + \frac{\rho_w}{a'})}},$$
(3.177)

$$t = \sqrt{\frac{3}{2\pi G\rho_w}} \left[\sqrt{a'_2} F_1(\frac{1}{2}, -\frac{1}{4}; \frac{3}{4}; -\frac{\rho_m}{\rho_w a'^2}) \right]_{a_{in}}^a.$$
 (3.178)

 $_2F_1$ is the hypergeometric function

$${}_{2}F_{1}(1/2,b;b+1;u) = \sum_{n=0}^{\infty} \frac{(1/2)_{n}(b)_{n}}{(b+1)_{n}} \frac{u^{n}}{n!},$$
(3.179)

where $(b)_n$ is the Pochhammer symbol which is defined as

$$(b)_n = \begin{cases} 1 & \text{if } n = 0, \\ b(b+1)(b+2)...(b+n-1), & \text{if } n = 1, 2.... \end{cases}$$
(3.180)

The result presented in (3.178) corresponds to the result given in (2.164) with n = 3.

Type Ia supernovaes are known as standard candles since their measurements makes comparison of theory and observations is possible. This is done by the following relation

$$m = 5log_{10}(\frac{d_L}{1Mpc}) + 25 + M, (3.181)$$

where m and M are the apparent and the absolute magnitudes respectively. Then the distance modulus is defined as $\mu = m - M$.

Now we borrow one of our plots from chapter 2. In our previous study we have already performed curve fitting with type Ia supernovae data. We have found $H_0 = 71.03 \pm 0.20$, $\Omega_{w,0} = 0.889 \pm 0.015$, $\Omega_{m,0} = 0.111 \pm 0.015$ and $\chi^2/\nu = 1.008$ for M = -19.30 in domain wall dominated universe. We obtain $q_0 = -0.389$ with these values of cosmological density parameters. On the other hand for dark energy dominated universe we have found $H_0 = 71.80 \pm 0.22$, $\Omega_{\Lambda,0} = 0.715 \pm 0.012$, $\Omega_{m,0} = 0.285 \pm 0.012$ and $\chi^2/\nu = 0.990$ for M = -19.30. We obtain $q_0 = -0.572$. All these results are shown in Figure 3.1.



Figure 3.1. Distance Modulus vs Redshift Plot for M = -19.30 Reproduced Dots represent observations of Pantheon data while green line represents domain wall dominated universe and red line represents dark energy dominated universe.

In addition

$$\lim_{a \to \infty} q(a) = -0.5. \tag{3.182}$$

 $V(\phi)$ is formulated as

$$V(\phi) = -\frac{2\rho_w}{3(2+\omega)} (\frac{\phi}{F})^{-2(1+\omega)}.$$
(3.183)

Although power of ϕ in the potential is different from 2 it causes accelerated expansion of the universe.

3.9. Discussion

We have revealed three major points of cosmology in this study. These are the answers for the following questions:

- (i) Is there any corresponding energy for the cosmological constant?
- (ii) Is it possible to have accelerated universe without cosmological constant?
- (iii) Is ratio of dark matter to baryonic mater smaller than $\frac{0.27}{0.05}$?

It is known that constant energy density contributes to the Hubble function. We have examined this phenomena in Section 3.7 and in Section 3.8.1. Firstly we have solved our system of equations for early universe by taking

$$\phi(a) = \frac{F}{a^d}, \qquad d = -\frac{1}{2(1+\omega)}, \qquad \rho(a) = \frac{\rho_r}{a^4} + \Lambda.$$
 (3.184)

Then H(a) is found in the form of (3.140) where there is a constant term α . At first sight there is no contribution of Λ in α_1 given by (3.142). However one can also claim that Λ contributes in α_3 by substituting (3.147) in (3.146). This dilemma can be explained as follows. One can write

$$V(a_{in}) = \frac{3F^2 k a_{in}^{1/(1+\omega)-2}}{4\omega} + \frac{(3+2\omega)(4+3\omega)}{8\omega(1+\omega)^2} F^2 H^2(a_{in}) a^{1/(1+\omega)}(a_{in}) - \frac{\rho_r}{a_{in}^4} - \Lambda.$$
(3.185)

Then one can rewrite $H(a_{in})$ in terms of $V(a_{in})$ and Λ .

We have also examined the spatially flat late universe in Section 3.8.1 by supplying the following functions

$$\phi(a) = \frac{F}{a^d}, \qquad d = -\frac{1}{2(1+\omega)}, \qquad \rho(a) = \frac{\rho_m}{a^3} + \Lambda.$$
 (3.186)

The Hubble function has been found in the form given in (3.159) which has a constant term. At first sight this constant term which is given by (3.162) does not contain Λ . One can apply the same trick and one can write

$$V(a_{in}) = \frac{(3+2\omega)(4+3\omega)}{8\omega(1+\omega)^2} F^2 H^2(a_{in}) a_{in}^{1/(1+\omega)} - \frac{\rho_m}{a_{in}^3} - \Lambda.$$
 (3.187)

Again, one can rewrite $H(a_{in})$ in terms of $V(a_{in})$ and Λ and one can claim that H(a) contains Λ .

There are two possible interpretations:

- (i) Constant energy density does not contribute to the Hubble function, it only modifies value of potential at the beginning of the universe.
- (ii) Initial value of the Hubble function $H(a_{in})$, can be rewritten in terms of $V(a_{in})$ and Λ so constant energy density contributes to the Hubble function.

Both of them can be reasonable according to one's perspective. Although if one chooses second explanation which is in agreement with common trend, one can still ask the question in another way: Is it possible to have a constant term in the Hubble function when there is no constant energy density and no constant potential in the universe? To answer this question we have investigated single component universe in Section 3.6 by giving following functions

$$\phi(a) = \frac{F}{a^d}, \qquad d = -\frac{1}{2(1+\omega)}, \qquad \rho(a) = \frac{\rho_n}{a^n}.$$
 (3.188)

Then H(a) is found in the form given by (3.125). The Hubble function has a constant term although there is no constant energy density. This constant is given in (3.127). Now $V(a_{in})$ does not contain Λ . In addition, expected exponential expansion of a(t)is given in (3.136). Thus there is one possible explanation. When there is no constant energy density, the Hubble function still has a constant term which causes exponential expansion.

We have shown that universe has an accelerated expansion when one introduces cosmic domain walls with matter in the energy density in Section 3.8.2. We had studied this case with Friedmann cosmology in chapter 2, and we have compared our model with the latest supernovae data. It is seen that both dark energy dominated and domain wall dominated universe have perfect fit with data. The differences between these two models most probably will be seen when bigger redshift data are available.

While comparison of dark energy dominated universe with observation has resulted in $\Omega_{\Lambda} = 0.715 \pm 0.012$ and $\Omega_m = 0.285 \pm 0.012$, comparison of domain wall dominated universe with observation has resulted in $\Omega_w = 0.889 \pm 0.015$ and $\Omega_m = 0.111 \pm 0.015$. Hence a new question appears; is the ratio of dark matter to baryonic matter less than $\frac{0.27}{0.05}$?

In 1980s Modified theories of Newtonian Dynamics or MOND, were proposed [58, 139]. It mainly claims that observational aspects of galaxies can be understood without dark matter. Recent observational evidence for the external field effect in MOND which was proposed as an alternative to dark matter is presented in [60]. A new relativistic MOND theory [59] successfully reproduces cosmic microwave background power spectra. These theories may explain non-baryonic part of domain wall dominated universe which covers six percent of the energy-matter content of the universe.

In this chapter we have applied change of variable $a = \dot{a}H(a)$ to field equations of Brans-Dicke theory and have written all equations in terms of independent variable "a". Then we have ended up with a constraint equation and a Bernoulli type differential equation which can be linearized. We have presented analytic solutions for supplied pairs of functions; ($\phi(a)$ and $\rho(a)$), ($\phi(a)$ and V(a)), or ($\phi(a)$ and H(a)).

Investigation of single component universe has shown that one will have a constant term in the Hubble function although there is no constant energy density. Early epoch of the universe with dark energy and radiation have been studied, and exponential expansion is seen in a(t). Late-time acceleration is obtained for both dark energy dominated universe and domain wall dominated universe. When Brans-Dicke parameter $\omega \gg 1$, the Hubble function reduces to H(a) of the Einstein cosmology. In all cases potential is found as combination of power law potentials.

We would like to emphasize that so-called dark energy term may be just a number without corresponding constant energy-matter density. Details of this subject has been presented in the discussion. Hopefully astronomers and cosmologists will pay more attention for searches of cosmic domain walls. One can also reach our calculations which are presented in this chapter in [140].

4. INFLATION AND LINEAR EXPANSION IN THE RADIATION DOMINATED ERA IN BRANS-DICKE COSMOLOGY

In this chapter we study ϕ^4 potential in Brans-Dicke cosmology. This potential has been already recognized in Section 3.3.1.2. It results in real Hubble function and positive energy density. Our systematic studies have shown that choice of other power law potentials can not satisfy positivity of energy density together with real Hubble function. Here instead of presenting details of these parts, we just focus on ϕ^4 potential. Now we will pick time as a independent variable different than our choices in chapter 2 and chapter 3.

In this chapter we will use FLWR metric given by,

$$ds^{2} = dt^{2} - a^{2}(t)\left[\frac{dr^{2}}{1 - kr^{2}} + r^{2}(d\theta^{2} + \sin^{2}\theta d\varphi^{2})\right].$$
(4.1)

where k is the curvature parameter with k = -1, 0, 1 corresponding to open, flat, closed universes respectively, a(t) is the scale factor of the universe. Here r is dimensionless and a(t) has the dimension of lenght. We use the Lagrangian which has been introduced in chapter 3. We choose

$$V(\phi) = \frac{1}{4}\lambda\phi^4,\tag{4.2}$$

which in flat spacetime leads to a renormalizable quantum field theory. Due to the choice of scale-invariant potential, the coupling constant λ is dimensionless, so that there are no dimensional parameters in the BDJT part of the Lagrangian density in (3.2) and (3.3).
We would like to remind the reader the field equations which have been derived in previous chapter,

$$\frac{3}{4\omega}\phi^2\left(\frac{\dot{a}^2}{a^2} + \frac{k}{a^2}\right) - \frac{1}{2}\dot{\phi}^2 - V(\phi) + \frac{3}{2\omega}\frac{\dot{a}}{a}\dot{\phi}\phi = \rho_m, \qquad (4.3)$$

$$\frac{-1}{4\omega}\phi^2\left(2\frac{\ddot{a}}{a} + \frac{\dot{a}^2}{a^2} + \frac{k}{a^2}\right) - \frac{1}{\omega}\frac{\dot{a}}{a}\dot{\phi}\phi - \frac{1}{2\omega}\ddot{\phi}\phi - \left(\frac{1}{2} + \frac{1}{2\omega}\right)\dot{\phi}^2 + V(\phi) = p_m, \quad (4.4)$$

$$\ddot{\phi} + 3\frac{\dot{a}}{a}\dot{\phi} + \frac{dV(\phi)}{\phi} - \frac{3}{2\omega}\left(\frac{\ddot{a}}{a} + \frac{\dot{a}^2}{a^2} + \frac{k}{a^2}\right)\phi = 0.$$
(4.5)

We will name (4.3) as the energy density equation, (4.4) as the pressure equation and (4.5) as the ϕ equation.

Sen and Seshadri [141] have investigated the nature of a potential relevant to the power law expansion in BD cosmology. In [136–138] a perturbation technique was applied to the above equations with $V(\phi) = \frac{1}{2}m^2\phi^2$. In addition, exact solutions of modified BD cosmological equations have been found by using symmetry analysis [142]. On the other hand the quartic potential, $V(\phi) = \frac{1}{4}\lambda\phi^4$ has been studied in [143] where solutions are comparable with the observed cosmological data only for small negative values of ω for spatially flat FRW geometry. Chubaryan *et al.* have studied the quartic potential with barotropic equation of state in [144]. This potential has also been studied in [145] in the presence of a generalized Brans-Dicke parameter $\omega_{GBD}(\phi)$. Santos and Gregory have found linearly and exponentially expanding solutions for vacuum cosmologies [146].

Although to investigate role of stiff matter in cosmology was not the main purpose of this work, it appears in our results. An exocit fluid with an equation of state $\nu = p/\rho = 1$ was first introduced by Zeldovich [147]. This fluid is also called as the stiff fluid or Zeldovich fluid and gives energy density proportional to $1/a^6$. Many scientists have produced cosmological models with stiff matter [148–156]. In addition there exist studies on stiff matter in Brans-Dicke Theory [157–159]. In an another work a complex scalar field description of Bose-Einstein condensate dark matter was studied [160]. It has been found that the early universe evolves from stiff $(p = \rho)$ to radiationlike $(p = \rho/3)$. We have obtained a similar result in our calculations.

In this chapter we present exact solutions of (4.3-4.5) with $V(\phi) = \frac{1}{4}\lambda\phi^4$ in a k = 1 closed universe. We choose $\omega > 4 \times 10^4$ to be comfortable compatible with results of Einstein telescope [130] and time delay experiments [131]. Our starting point is to introduce a scale invariant solution in Section 4.1. Once we find the Jordan scalar field as a function of time, we calculate the scale-factor of the universe for different eras by tracking the behaviour of the field forward and backward in time. Therefore we will have three different $\phi(t)$, we will solve the ϕ equation for each case and we will obtain the scale factor as a function of time for each era. These solutions result in inflation in the early radiation dominated era, linear expansion in the late radiation dominated era and scale invariant solution with stiff fluid between these two eras.

Here we should take attention of the reader to the important remark. In classical cosmology the basic information obtained from observation is the value of the Hubble function. It is related with the theory by one of the Friedmann-Lemaitre equation

$$\dot{a}^{2}(t) = \frac{8\pi G}{3}\rho a^{2}(t) + \frac{1}{3}\Lambda c^{2}a^{2}(t) - c^{2}k.$$
(4.6)

In addition the followings are definitions of the Hubble function, total energy density, density parameters and the curvature density parameter

$$H \equiv \frac{\dot{a}}{a}, \qquad \rho(t) \equiv \rho_{m,0} [\frac{a_0}{a(t)}]^3 + \rho_{r,0} [\frac{a_0}{a(t)}]^4 + \rho_{\Lambda,0}, \tag{4.7}$$

$$\Omega_i(t) \equiv \frac{8\pi G}{3H^2(t)} \rho_i(t), \qquad \Omega_k(t) = -\frac{c^2 k}{H^2(t)a^2(t)}, \tag{4.8}$$

where the label i includes matter, radiation, and the vacuum. Comparison with observation is usually made by writing last equation in the form

$$H^{2} = H_{0}^{2} \{\Omega_{m,0} [\frac{a_{0}}{a(t)}]^{3} + \Omega_{r,0} [\frac{a_{0}}{a(t)}]^{4} + \Omega_{\Lambda,0} + \Omega_{k,0} [\frac{a_{0}}{a(t)}]^{2} \},$$
(4.9)

where 0 denotes present values of the related function. However in our model right hand side of the equation is different,

$$H^{2} = \frac{\rho + \frac{1}{2}\dot{\phi}^{2} + V(\phi) - \frac{3}{2\omega}\frac{\dot{a}}{a}\dot{\phi}\phi}{\frac{3}{4\omega}\phi^{2}} - \frac{k}{a^{2}}.$$
(4.10)

Thus our model should be compared with (4.9) after expressing ϕ and $\dot{\phi}$ as functions of *a*. The only mathematical constraint on the above equation is positivity of right side. For this reason one can obtain interesting negative energy density contributions by providing right hand side is positive as a whole. At first sight, negative energy density seems unreasonable. However in the literature there are many physical cases which contain negative energy. It was firstly discovered by Casimir in quantum field theory [161]. In 1975, Hawking found the negative energy density across the event horizon [162]. In the following years the negtaive energy fluxes in radiation from moving mirrors were reported [163, 164]. In different fields there are many studies which give rise to comments on negative energy [165–170].

We also show that introducing matter in the linearly expanding radiation era can give a decelerating universe with positive energy density perturbation or an accelerating universe with negative energy density perturbation depending on the choice of one of the constants in our model in Section 4.1. Then early inflation is investigated in Section 4.2. We find the Hubble function and the deceleration parameter for each era. Then we calculate temperature-time relations for each era we investigate in Section 4.3. We study the passage from big bang to the scale invariant solution and the passage from the scale invariant solution to the linearly expanding solution in Section 4.4. Finally we extrapolate our results to the present day and show that this model gives acceleration for the present universe, albeit with a deceleration parameter which is not as big as observed in Section 4.5.

4.1. Radiation Dominated Era

4.1.1. The Scale Invariant Solution

One important property of potential in (4.2) is that it does not introduce any dimensional parameters into the Lagrangian density so that the action and the resulting equations are scale invariant. In this part we have assumed that the relation between $\phi(t)$ and a(t) preserves the scale invariance and is given by

$$\phi(t) = \frac{A}{a(t)},\tag{4.11}$$

where A is a positive dimensionless constant. With this constraint, the ϕ equation becomes a second order nonlinear differential equation.

$$\frac{A[-3+2A^2\omega\lambda - (3+2\omega)\dot{a}^2 - (3+2\omega)a\ddot{a}]}{2\omega a^3} = 0,$$
(4.12)

where dot denotes derivative with respect to time. To be able to solve this equation we introduce new variable $\theta(t) = a^2(t)$. Then the differential equation reduces to

$$\ddot{\theta} = \frac{4\omega\lambda A^2 - 6}{3 + 2\omega}.\tag{4.13}$$

One can easily find θ and with appropriate choice of integration constants b_1 and b_2 the solution for the scale factor can be written as

$$a(t) = \sqrt{\left(\frac{2A^2\omega\lambda - 3}{3 + 2\omega}\right)t^2 + b_1t + b_2}.$$
(4.14)

Behaviour of energy density and pressure are found to be

$$\rho = A^2 \left[\frac{6 - 3A^2 \omega \lambda}{4a^4 \omega} + \frac{-(3 + 2\omega)b_1^2 + 4(-3 + 2A^2 \omega \lambda)b_2}{16a^6 \omega} \right],$$
(4.15)

$$p = A^2 \left[\frac{2 - A^2 \omega \lambda}{4a^4 \omega} + \frac{-(3 + 2\omega)b_1^2 + 4(-3 + 2A^2 \omega \lambda)b_2}{16a^6 \omega} \right].$$
 (4.16)

The term proportional to $1/a^6$ is the stiff matter term [147]. By using the usual continuity equation which is satisfied by (4.3-4.5) the terms proportional to a^{-4} are readily recognized as radiation whereas the a^{-6} stiff fluid are related to maximal pressure $p = \rho$ without violation of positivity of energy. Positivity of both terms and real scale factor requires $3/2 < A^2 \omega \lambda < 2$ and $b_2 > (3 + 2\omega)b_1^2/4(2A^2\omega\lambda - 3)$. We should note that constants b_1 and b_2 are important and they must not be chosen zero. Solutions before and after this era will be matched by adjusting b_1 and b_2 . We will call this phase of the universe as the scale invariant phase.

As the universe expands the second term becomes negligible Then the equation of state becomes,

$$\nu = \frac{p}{\rho} = \frac{1}{3},\tag{4.17}$$

as it should be in the radiation dominated era.

There are two more important cosmological parameters we should calculate are the Hubble function and the deceleration parameter;

$$H(t) = \frac{\dot{a}}{a}, \qquad q(t) = \frac{d}{dt}(\frac{1}{H}) - 1 = -\frac{\ddot{a}\ddot{a}}{\dot{a}^2}.$$
(4.18)

For this era these parameters are found as

$$H(t) = \frac{2(\frac{2A^2\omega\lambda - 3}{3 + 2\omega})t + b_1}{2[(\frac{2A^2\omega\lambda - 3}{2 + 2\omega})t^2 + b_1t + b_2]},$$
(4.19)

$$q(t) = \frac{(3+2\omega)[(3+2\omega)b_1^2 + 4(3-2A^2\omega\lambda)b_2]}{[2+(2A^2\omega\lambda-3) + (3+2\omega)b_1]^2}.$$
(4.20)

This era takes place after early inflation. Thus time is always greater than the value of t_1 which is found in Section 4.4 by (4.78). Therefore one can easily conclude that the Hubble parameter is always positive. Our condition on b_1 and b_2 which makes energy density positive, makes deceleration parameter negative. Thus we have an accelerated era.

4.1.2. Linearly Expanding Radiation Dominated Universe

As time increases its effect in (4.14) becomes larger so we can make the assumption

$$\phi(t) = \frac{B}{t}.\tag{4.21}$$

Then ϕ equation becomes

$$\frac{4\omega Ba^2 - 6\omega Bta\dot{a} + 2\omega\lambda B^3 a^2 - 3Bt^2(a\ddot{a} + \dot{a}^2 + 1)}{2\omega t^3 a^2} = 0.$$
(4.22)

Firstly we set $a^2 = \theta$ in the above equation and we obtain

$$t^2\ddot{\theta} + 2\omega t\dot{\theta} - \frac{4\omega}{3}(2+\lambda B^2)\theta = -2t^2.$$

$$(4.23)$$

Last equation is easily recognized as non-homogeneous Cauchy-Euler equation and its solution is found as

$$\theta(t) = c_1 t^{m_+} + c_2 t^{m_-} + \frac{3t^2}{2\omega(\lambda B^2 - 1) - 3},$$
(4.24)

$$m_{\pm} = \frac{1}{2} - \omega \pm \sqrt{\omega^2 - \omega + \frac{1}{4} + \frac{4\omega}{3}(2 + \lambda B^2)}, \qquad (4.25)$$

where c_1 and c_2 are integration constants (Details of this calculation is given in Appendix C). Unless one choose $c_1 = c_2 = 0$, it is impossible to obtain pressure and the density in the form c/a^n with constant c and rational n. Therefore we obtain the scale factor and the Jordan field as

$$a(t) = \sqrt{\frac{3}{2B^2\omega\lambda - (2\omega + 3)}}t, \quad \phi(t) = \sqrt{\frac{3}{2B^2\omega\lambda - (2\omega + 3)}}\frac{B}{a(t)}.$$
 (4.26)

This choice satisfies the ϕ equation which is always equal to zero. By using the gravitational field equations one can easily calculate energy, pressure and equation of state as

$$\rho = \frac{9B^2[B^2\omega\lambda - 2(2\omega+3)]}{[2B^2\omega\lambda - (2\omega+3)]^2} \frac{1}{4\omega a^4},$$
(4.27)

$$p = \frac{9B^2[B^2\omega\lambda - 2(2\omega+3)]}{[2B^2\omega\lambda - (2\omega+3)]^2} \frac{1}{12\omega a^4},$$
(4.28)

$$\nu = \frac{p}{\rho} = \frac{1}{3}.$$
(4.29)

Note that in this era although we have not imposed the ansatz $\phi(t) = A'/a(t)$, we have ended up with it.

After this point we will continue with results of Section 4.4 where we match solutions for each era. Continuity of $\phi(t)$ and a(t) at the passage from scale invariant phase to linearly expanding era gives

$$a(t) = \sqrt{\frac{2A^2\omega\lambda - 3}{2\omega + 3}}t, \quad B = A\sqrt{\frac{2\omega + 3}{2A^2\omega\lambda - 3}}.$$
 (4.30)

Now let us calculate the Hubble function and the deceleration parameter for this era,

$$H(t) = \frac{1}{t}, \qquad q(t) = 0.$$
 (4.31)

Thus universe expands with constant velocity.

4.1.3. Creation of Matter in the Late Radiation Dominated Era

When the temperature is much smaller than $m_e = 0.51$ MeV electrons and baryons can be considered non-relativistic. Pressure as compared to matter energy density can be neglected for a gas of non-relativistic particles. For this reason it can be accepted that at this stage of the universe non-relativistic matter starts to be created and eventually dominates radiation. We assume that creation of matter in radiation dominated era causes small changes in the Jordan field and in the scale factor. Thus we start the ansatz,

$$\tilde{\phi}(t) = \frac{A}{\mathcal{B}t} + \psi(t) \quad \text{with} \quad \psi(t) = ut^m,$$
(4.32)

$$\tilde{a}(t) = \mathcal{B}t + \alpha(t) \quad \text{with} \quad \alpha(t) = vt^n,$$
(4.33)

$$\mathcal{B} = \sqrt{\frac{2A^2\omega\lambda - 3}{2\omega + 3}},\tag{4.34}$$

and ψ and α are small. First we write all three equations in terms of new functions. In addition we neglect second and higher order terms $(\alpha^2, \psi^2, \alpha \psi, ...)$ in the corrections in the perturbations of α and ψ . We easily obtain the constant $v \sim t^{2+m-n}$ from the ϕ equation. It follows that we must choose m = n - 2 to keep v constant. Then energy density becomes

$$\rho = \frac{C_1}{t^4} + \frac{C_2}{t^{5-n}},\tag{4.35}$$

with C_1 and C_2 are constants. Thus we choose n = 2 to have two component energy density which is composed of a radiation part and a matter part

$$\rho = \frac{C_r}{a^4} + \frac{C_m}{a^3},$$
(4.36)

where

$$C_r = -\frac{3A^2(-2+A^2\omega\lambda)}{4\omega},\tag{4.37}$$

$$C_m = -\frac{Au[27 + 2\omega(6 + A^2\lambda(-12 - 5\omega + 2A^2\omega\lambda(2 + \omega)))]}{2\omega[-9 - 5\omega + 2A^2\omega\lambda(2 + \omega)]}.$$
 (4.38)

Then pressure is found as,

$$p = \frac{C_r/3}{a^4} + \frac{P_m}{a^3},\tag{4.39}$$

$$P_m = \frac{Au(-2 + A^2w\lambda)[-w + 2A^2w\lambda(1+w)]}{w[-9 - 5\omega + 2A^2\omega\lambda(2+\omega)]}.$$
(4.40)

The constant v which is part of the scale factor, is found as

$$v = \frac{u(-3+2A^2\omega\lambda)[-\omega+2A^2\omega\lambda(1+\omega)]}{A(3+2\omega)[-9-5\omega+2A^2\omega\lambda(2+\omega)]}.$$
(4.41)

We have already found $3/2 < A^2 \omega \lambda < 2$ at the end of the discussion of Section 4.1.1. In addition A, λ, ω are all positive parameters. In this scope one can easily find the following results by inspection,

$$sign(C_m) \equiv sign(P_m) \equiv sign(u) \quad and \quad sign(v) \equiv -sign(u).$$
 (4.42)

In addition it is also obvious that

$$\lim_{A^2\omega\lambda\to 2^-} \frac{p_{perturbation}}{\rho_{perturbation}} = \lim_{A^2\omega\lambda\to 2^-} \frac{P_m}{C_m} \to 0^+ \ll 1,$$
(4.43)

as it should be in the matter dominated era. Thus we can neglect the perturbation in pressure. In addition this result shows that we should take the value of $A^2\omega\lambda$ as 2⁻ that is a little less than 2. Other effects of this result on our model will be explained in the following sections.

The Hubble function and the deceleration parameter for this era are found as

$$H(t) = \frac{\sqrt{\frac{2A^2\omega\lambda - 3}{3+2\omega}} + 2vt}{\sqrt{\frac{2A^2\omega\lambda - 3}{3+2\omega}}t + vt^2}, \quad q(t) = -\frac{2tv(tv + \sqrt{\frac{2A^2\omega\lambda - 3}{3+2\omega}})}{(\sqrt{\frac{2A^2\omega\lambda - 3}{3+2\omega}} + 2tv)^2}.$$
 (4.44)

By using $v \ll 1$ which is the condition to write the perturbation in the field equations, we obtain

$$H(t) \simeq \frac{1}{t} + \frac{v}{\sqrt{\frac{2A^2\omega\lambda - 3}{3+2\omega}}}, \quad q(t) \simeq \frac{-2vt}{\sqrt{\frac{2A^2\omega\lambda - 3}{3+2\omega}}}.$$
(4.45)

Now one should develop a solid argument to determine the sign of the constant v. Firstly let us start with positivity of energy density. In this scope the parameter u must be chosen as a positive number to ensure positivity of perturbation term in energy density. Thus v is found to be negative. This tell us that the Hubble function can be negative and the universe can decelerate. This means creation of matter can cause a big crunch depending on the values of the constants in our model.

But since the universe has expanded during this era it has the positive Hubble function. This makes the constant v positive and thus the constant u negative. At this moment one can immediately say that matter energy density is negative. However baryonic matter has positive energy density. Therefore one should write the matter perturbation to radiation energy density with two components where one of them is positive and the other one is negative for example,

$$C_m = -\frac{Au[2\omega A^2\lambda(-12-5\omega)]}{2\omega[-9-5\omega+2A^2\omega\lambda(2+\omega)]} - \frac{Au[27+2\omega(6+A^2\lambda(2A^2\omega\lambda(2+\omega)))]}{2\omega[-9-5\omega+2A^2\omega\lambda(2+\omega)]}.$$
(4.46)

Thus we conclude that universe can accelerate in this era by using positivity of v. This result can be interpreted as baryonic matter being created together with matter of negative energy density. This causes acceleration which in the standard model would be described as dark matter. At this point we should once again mention that because of the dependence of Jordan scalar field $\phi(t)$ on the scale size a(t) the contribution of this negative energy density term to observation will be different from Einstein's gravity as indicated in (4.9) and (4.10).

4.2. Early Inflation in the Radiation Dominated Era

Here we relax our constraint $\phi(t) = \frac{A}{a(t)}$. However we will keep matching our solution for the Jordan field which was found to be

$$\phi(t) = \frac{A}{\sqrt{\left(\frac{2A^2\omega\lambda - 3}{3+2\omega}\right)t^2 + b_1t + b_2}}.$$
(4.47)

We see that as t goes to zero, ϕ becomes constant and to investigate this behaviour we look for a solution

$$\phi(t) = F. \tag{4.48}$$

 ϕ equation becomes

$$F^{3}\lambda - \frac{3F}{2wa^{2}}(1 + \dot{a}^{2} + a\ddot{a}) = 0.$$
(4.49)

$$\ddot{\theta} - \frac{4\omega\lambda F^2\theta}{3} = -2,\tag{4.50}$$

and the positive solution for the scale factor of the universe is

$$a(t) = \sqrt{c_1 e^{\alpha t} + c_2 e^{-\alpha t} + \frac{3}{2\omega\lambda F^2}}, \qquad \alpha = 2F\sqrt{\frac{\omega\lambda}{3}}.$$
(4.51)

For this solution $\phi(t)$ can be interpreted as an effective cosmological constant. Energy density and pressure are easily calculated by substitution of a(t) and $\phi(t)$ in Equations (4.3) and (4.4) as

$$p = \frac{9 - 16c_1c_2F^4\omega^2\lambda^2}{48\omega^2\lambda a^4},$$
(4.52)

$$\rho = \frac{9 - 16c_1c_2F^4\omega^2\lambda^2}{16\omega^2\lambda a^4},$$
(4.53)

$$\nu = \frac{p}{\rho} = \frac{1}{3}.$$
 (4.54)

To have $\rho > 0$ we must satisfy the condition $9 - 16c_1c_2F^4\omega^2\lambda^2 > 0$.

To simplify our result let us choose $c_1 = d_1/\alpha^2$ and $c_2 = d_2/\alpha^2$ where $d_1 > 0$. Then we have

$$a(t) = \frac{1}{\alpha}\sqrt{d_1 e^{\alpha t} + d_2 e^{-\alpha t} + 2}.$$
(4.55)

Choosing the integration constant d_1 and d_2 such that a(t) is minimum at t = 0 gives us $d_1 = d_2$ and a(t) must approach a positive non-zero value at t = 0. Therefore our results become

$$a(t) = \frac{\sqrt{2}}{\alpha} \sqrt{d_1 \cosh(\alpha t) + 1},\tag{4.56}$$

$$\rho(t) = \frac{9(1 - d_1^2)}{16\omega^2 \lambda a^4}, \quad p = \frac{\rho}{3}, \tag{4.57}$$

where $0 < d_1 < 1$.

For this solution the Hubble function and the deceleration parameter are found as

$$H(t) = \frac{d_1 \alpha \sinh(\alpha t)}{2(d_1 \cosh(\alpha t) + 1)},\tag{4.58}$$

$$q(t) = 1 - 2\coth^2(\alpha t) - \frac{2}{d_1}\coth(\alpha t)\operatorname{csch}(\alpha t).$$
(4.59)

It is obvious that the Hubble function takes positive values. Let us make a change of variable $x = \cosh(\alpha t)$ to see the behaviour of the deceleration parameter

$$q(x) = 1 - 2\left(\frac{x^2 + x/d_1}{x^2 - 1}\right), \qquad x \ge 1.$$
(4.60)

Since $x \ge 1$, the value of q(x) is always smaller than zero. Hence we have an accelerated expansion. In addition t = 0 corresponds x = 1 which creates singularity in the function q(x). This negative infinite value of the deceleration parameter can explain the big bang.

We can also choose a(t) = 0 when t = 0 with $d_2 = -2 - d_1$. Then we obtain

$$a(t) = \frac{\sqrt{2}}{\alpha} \sqrt{d_1 \sinh(\alpha t) + 1 - e^{-\alpha t}},\tag{4.61}$$

$$\rho(t) = \frac{9(1+d_1)^2}{16\omega^2\lambda a^4}, \quad p = \frac{\rho}{3}.$$
(4.62)

with $0 < d_1$.

For this solution the Hubble function and the deceleration parameter are found as

$$H(t) = \frac{\alpha [1 + d_1 e^{\alpha t} \cosh(\alpha t)]}{(e^{\alpha t} - 1)(2 + d_1 + d_1 e^{\alpha t})},$$
(4.63)

$$q(t) = -1 + \frac{8e^{\alpha t}[1 + d_1 e^{\alpha t}(2 + d_1 - \sinh(\alpha t))]}{(2 + d_1 + d_1 e^{2\alpha t})^2}.$$
(4.64)

Again let us make a change of variable $x = e^{\alpha t}$ to understand the behaviour of the last functions. Thus we have

$$H(x) = \frac{\alpha [1 + d_1 (x^2 + 1)/2]}{(x - 1)(2 + d_1 + d_1 x)},$$
(4.65)

$$q(x) = -1 + \frac{8x[1 + d_1x(2 + d_1 - (x - x^{-1})/2)]}{(2 + d_1 + d_1x^2)^2}, \qquad x \ge 1.$$
(4.66)

Because $x \ge 1$ the Hubble function is always positive. In this kind of big bang the Hubble function becomes infinite when x = 1 (t = 0). Besides, at this moment q(1) = 1 and as x increases (as time increases) q(x) becomes negative. Therefore expansion always starts with deceleration then acceleration occurs.

These results may indicate that early inflation took place in the radiation dominated era under the effect of the Brans-Dicke-Jordan field.

4.3. Temperature Calculations

The relation between energy density, pressure and temperature had already been derived as

$$\frac{dp(T)}{dT} = \frac{1}{T} [\rho(T) + p(T)].$$
(4.67)

In our calculations pressure and energy density is a function of time. Thus we apply chain rule and obtain the temperature as

$$T = T_0 exp\left[\int_{t_0}^t \frac{dp(t')}{\rho(t') + p(t')} dt'\right],$$
(4.68)

where T_0 is the reference temperature of the universe when it evolves with the scale factor $a(t_0)$. For the early universe, both choices of the scale factor gives the same temperature-time relation which is found as

$$T = T_0 \frac{a(t_0)}{a(t)}.$$
(4.69)

For the scale-invariant phase we have

$$T = T_0 \frac{\tilde{a}^2(t) a^3(t_0)}{\tilde{a}^2(t_0) a^3(t)},\tag{4.70}$$

$$\tilde{a}(t) = \sqrt{\left(\frac{2A^2\omega\lambda - 3}{3 + 2\omega}\right)t^2 + b_1t + \frac{(3 + 2\omega)b_1^2 - 4b_2}{8(2 - A^2\omega\lambda)}}.$$
(4.71)

In the Section 4.1.3 we found that $A^2 \omega \lambda$ is a little less than 2. Therefore temperature becomes $T_0 \frac{a^3(t_0)}{a^3(t)}$.

For the linearly expanding radiation dominated era we have obtained

$$T = T_0 \frac{a(t_0)}{a(t)}$$
 or $T = \frac{\tilde{T}_0}{t}$, (4.72)

which is the standard time-temperature relation in the radiation dominated era.

For the era where we introduce matter into radiation temperature is more complicated than the other cases and we find the temperature to be given by

$$T = \frac{(c_1 + uc_2 t)^r}{t} \quad \text{with} \quad \frac{1}{4} < r < 1,$$
(4.73)

where c_1 and c_2 are constant. Note that the constant perturbation term $u \ll 1$. Hence by using a series expansion the temperature can be written as

$$T = T_{\infty} \left(1 + \frac{t_1}{t} \right), \tag{4.74}$$

where T_{∞} and t_1 are constant. Physically T_{∞} will denote the approximate temperature at the end of the radiation dominated era provided that t_1 can be chosen small compared to the time elapsed from big bang to the end of the radiation dominated era.

4.4. Matching the Solutions

In this section we will try to match solutions of the scale factor, the Jordan field and the energy density for different eras which follow each other in time. We will not be interested in pressure and temperature because microscopic events can affect them. Let us call the scale factor of the universe $a_1(t)$ for the early inflation era, $a_2(t)$ for the scale invariant era, and $a_3(t)$ for the late radiation dominated era. Similarly we name the Jordan field solutions as $\phi_1(t)$, $\phi_2(t)$ and $\phi_3(t)$ and the energy densities as $\rho_1(t)$, $\rho_2(t)$, $\rho_3(t)$ respectively. Initially we have tried to match $\phi_1(t)$ with $\phi_2(t)$ smoothly at a certain time. Then we try to match $a_1(t)$ with $a_2(t)$ smoothly at the same certain time. Thus we obtain four equations for continuity of $\phi(t)$, $\dot{\phi}(t)$, a(t) and $\dot{a}(t)$. However it is only possible to have three equations satisfied at the same boundary. We eliminate continuity of $\dot{a}(t)$. Consequently we end up with the following equations

$$\phi_1(t) = \phi_2(t) \quad \text{at } t = t_1,$$
(4.75)

$$\dot{\phi}_1(t) = \dot{\phi}_2(t) \quad \text{at } t = t_1,$$
(4.76)

$$a_1(t) = a_2(t)$$
 at $t = t_1$. (4.77)

The passage from early inflation to scale invariant phase occurs at time t_1 where

$$t_1 = -\frac{3+2\omega}{2(2A^2\omega\lambda - 3)}b_1,$$
(4.78)

thus $b_1 < 0$. When we use $a_1(t) = \frac{\sqrt{2}}{\alpha} \sqrt{d_1 \cosh(\alpha t) + 1}$ we obtain

$$d_1 = \frac{2A^2\omega\lambda - 3}{3\cosh(\alpha t_1)},\tag{4.79}$$

so that the condition $0 < d_1 < 1$ satisfied. When we have $a_1(t) = \frac{\sqrt{2}}{\alpha} \sqrt{d_1 \sinh(\alpha t) + 1 - e^{-\alpha t}}$

$$d_1 = \frac{2A^2\omega\lambda - 3}{3\sinh(\alpha t_1)} + \frac{e^{-\alpha t_1}}{\sinh(\alpha t_1)},\tag{4.80}$$

so that the condition $0 < d_1$ satisfied.

By using the information obtained above and $3/2 < A^2 \omega \lambda < 2$ one can easily compare energy densities for the early universe and the scale-invariant phase. It is seen that there is a loss in energy density at the passage. In Section 4.1.3 we have concluded that the value of $3/2 < A^2 \omega \lambda < 2$ is a little less than 2. This makes scale invariant era stiff fluid dominated and in this limit there is still a loss in energy density at the passage.

Now we will study matching the scale-invariant phase with the linearly expanding radiation dominated era. We can satisfy only continuity of a(t) and $\phi(t)$. This gives the following results

$$t_2 = \frac{-b_2}{b_1}, \quad B = A\sqrt{\frac{2\omega+3}{2A^2\omega\lambda-3}}.$$
 (4.81)

Therefore when we combine the outcomes of the two matching procedure, scale factors for second and third eras can be written as

$$a_2(t) = \sqrt{\left(\frac{2A^2\omega\lambda - 3}{2\omega + 3}\right)\left[t^2 + 2t_1(t_2 - t)\right]},$$
(4.82)

$$a_3(t) = \sqrt{\frac{2A^2\omega\lambda - 3}{2\omega + 3}}t.$$
 (4.83)

Again it is found that there is a loss in the energy density at the passage.

4.5. Discussion

There are three gravitational field equations given by (4.3-4.5). A standard mathematical procedure to solve such a set of differential equations which has nonlinear terms with variable coefficients has not been invented yet. Usually one chooses an appropriate energy density to find the solution for the desired era. However our approach to the problem is different. We have used the scale invariant ansatz $\phi(t) = A/a(t)$ and obtained an exact solution for the Jordan field and the scale factor which evolves from radiation and stiff fluid combined phase to a radiation phase. Similar results have been found in the standard model [160]. We have named this combined phase as the scale invariant phase and by investigating the behaviour of the Jordan field backward in time, we have found a universe which starts to expand exponentially at big bang with pure radiation. Similarly, investigating the behaviour of the Jordan field by extrapolating forward in time we again obtain a pure radiation dominated phase which expands linearly. We have found that introducing matter in this linearly expanding late radiation era may cause deceleration or acceleration. Furthermore we have presented the time-temperature relations for each era. As a result we have not only found the scale factor and the Jordan field for each era but also we have found the order of the relevant eras in time. In the last part of the calculations we have matched solutions for $\phi(t)$ and a(t) at the boundaries of the eras.

Importantly, we have formulated two different scale factors which cause inflation at the beginning of time. One of them always has an accelerated expansion and the other one starts with decelerated expansion and then turns into accelerated expansion. Acceleration of the universe continues in stiff fluid dominated phase. This acceleration decreases and the universe expands linearly. Introducing matter in this era may cause big crunch or may cause acceleration of the universe depending on our choice of one of the parameters in the perturbation. Thus the three important features of closed space-like section, radiation domination and primordial inflation can be explained by the JBDT model.

In our model we have not formulated a solution for behaviour of the present universe. However by using field equations one can approximately find the Jordan scalar field $\phi(t)$ near the present era. For this purpose effective gravitational constant is defined as $G_{eff}^{-1} = \frac{2\pi}{\omega}\phi^2$. We take the Friedmann equation for our model given by (4.10) and the definitions given by (4.7) and (4.8) which are used in standard cosmology. Then we set $G_{present} = \frac{\omega}{2\pi\phi^2(t_0)}$ and put time $t = t_0$ to both sides of the equation where 0 denotes today. Here we should remind that the cosmological constant Λ in our model is zero. We obtain the following equation

$$\Omega_{m,0} + \Omega_{r,0} + \Omega_{k,0} + \frac{4\omega}{3H_0^2} [\frac{\dot{\phi}(t_0)}{\phi(t_0)}]^2 + \frac{\lambda\omega}{3H_0^2} \phi^2(t_0) - \frac{2}{H_0} [\frac{\dot{\phi}(t_0)}{\phi(t_0)}] = 1.$$
(4.84)

We borrow the following equation

$$\Omega_{m,0} + \Omega_{r,0} + \Omega_{k,0} + \Omega_{\Lambda,0} = 1, \qquad (4.85)$$

from the standard model of the cosmology. In addition we make a change of variable $\theta = \frac{\dot{\phi}(t_0)}{\phi(t_0)}$ and rearrange the terms for simplifications,

$$\theta^2 - \frac{3H_0}{2\omega}\theta + \frac{\lambda\omega}{8\pi G_{present}} - \frac{3H_0^2}{4\omega}\Omega_{\Lambda,0} = 0.$$
(4.86)

One can produce another useful formula by using today's value of pressure which is zero in the field equations. First we find $(-1/3)(\text{LHS of } (4.3))+(\text{LHS of } (4.4))=(-1/3)\rho$. Then we use the ϕ equation to get

$$\left(\frac{1}{3} + \frac{1}{2\omega}\right)\left[\ddot{\phi}(t)\phi(t) + \dot{\phi}^2(t)\right] + \left(1 + \frac{3}{2\omega}\right)\frac{\dot{a}(t)}{a(t)}\dot{\phi}(t)\phi(t) = -\frac{\rho}{3}.$$
(4.87)

We use the same tricks which have been used to find (4.86) with change of variable $\beta = \phi^2$. The result is

$$\ddot{\beta}(t_0) = \frac{-H_0^2}{4\pi G_{present}} \frac{(\Omega_{m,0} + \Omega_{r,0})}{(\frac{1}{3} + \frac{1}{2\omega})} - 3H_0\dot{\beta}(t_0).$$
(4.88)

First one should solve (4.86) and obtain θ . Then $\dot{\phi}(t_0) = \phi(t_0)\theta$ with $\phi(t_0) = \sqrt{\frac{\omega}{2\pi G_{present}}}$. Secondly one should use this information to find $\dot{\beta}(t_0) = 2\dot{\phi}(t_0)\phi(t_0)$. Then one should solve (4.88) and obtain $\ddot{\beta}$. Finally one should find $\ddot{\phi}(t_0) = \frac{\ddot{\beta} - 2\dot{\phi}^2(t_0)}{2\phi(t_0)}$. In conclusion one can approximately formulate the Jordan scalar field as

$$\phi(t) = \phi(t_0) + \dot{\phi}(t_0)(t - t_0) + \frac{1}{2}\ddot{\phi}(t_0)(t - t_0)^2, \qquad (4.89)$$

by using the Taylor expansion up to second order.

For clarification one can substitute numerical values. We take $H_0 = 73.52(km/s)/Mpc$, $\Omega_{\Lambda,0} = 0.68$, $\Omega_k \simeq 0$, $\Omega_{m,0} + \Omega_{r,0} = 0.32$, $G = 6.67 \times 10^{-11} m^3 k g^{-1} s^{-2}$ and $\omega = 4 \times 10^4$. After writing all components in terms of eV we have found that λ must be at most 1.3199×10^{-130} to have real solutions. At this value of λ ,

$$\dot{\phi}(t_0) = 2.86 \times 10^{-8} eV^2$$
 and $\ddot{\phi}(t_0) = -1.48 \times 10^{-40} eV^3$. (4.90)

At the value of $\lambda = 10^{-131}$ (4.86) gives two roots and our results are

$$\dot{\phi}(t_0) = -5.20 \times 10^{-6} eV^2$$
 and $\ddot{\phi}(t_0) = 2.44 \times 10^{-38} eV^3$, (4.91)

or

$$\dot{\phi}(t_0) = 5.26 \times 10^{-6} eV^2$$
 and $\ddot{\phi}(t_0) = -2.48 \times 10^{-38} eV^3$. (4.92)

One can calculate the value of G_{eff} 1.37×10^{10} years ago. When we use the highest allowed value of λ we find that G_{eff} was then $6.7243 \times 10^{-57} eV^{-2}$ and now it is $6.7236 \times 10^{-57} eV^{-2}$. We thus see that the expansion (4.89) can be extended to almost the end of the radiation dominated era.

Moreover one can calculate the deceleration parameter $q(t_0) = -\frac{a(t_0)\ddot{a}(t_0)}{\dot{a}^2(t_0)}$ after finding $\dot{\phi}(t_0)$ and $\ddot{\phi}(t_0)$. First we find 2(LHS of (4.3))+(LHS of (4.5))= 2\rho. Secondly we write $\frac{\ddot{a}(t)}{a(t)} = -q(t)H^2(t)$. Then we set $G_{present} = \frac{\omega}{2\pi\phi^2(t_0)}$ and put time $t = t_0$ to both sides of the equation. As a result we obtain

$$q(t_0) = \Omega_{m,0} + \Omega_{r,0} - \frac{2\omega}{3H_0^2} [\frac{\ddot{\phi}(t_0)}{\phi(t_0)} - (\frac{\dot{\phi}(t_0)}{\phi(t_0)})^2 + 3(1 + \frac{1}{\omega})H_0\frac{\dot{\phi}(t_0)}{\phi(t_0)} + \lambda\frac{\phi^2(t_0)}{2}].$$
 (4.93)

Numeric results are as follows. When λ has the highest allowed value $q(t_0) = -0.2$ which indicates present-universe is in a phase of accelerated expansion. As the value of λ decreases, the value of $q(t_0)$ approaches zero. On the other hand observational results were found as $q(t_0) = -1.0 \pm 0.4$ [17]. One can also reach our calculations which are presented in this chapter in [171, 172].

5. QUANTUM MECHANICS IN A SPACE WITH A FINITE NUMBER OF POINTS

In this chapter we would like to investigate what happens when the position space or the momentum space consists of a finite number of points so that the Hilbert space associated with the quantum mechanics is finite dimensional. Schwinger [104] has given a most instructive example where the position space and the momentum space each consist of d points and are both periodic. The starting point for this consideration is the well-known simple quantum mechanics on the circle S^1 . In this case the position space is continuous whereas the momenta are given by $p_n = \frac{\hbar}{r}n$, $(n = 0, \pm 1, \pm 2, ...)$ where r is the radius of the circle. Now restricting the position space S^1 to integer multiples of angle $2\pi/d$, the position eigenvectors can be denoted by $|n\rangle$, n = 0, 1, 2, ..., d - 1 such that

$$X\left|n\right\rangle = \frac{2\pi r}{d}n\left|n\right\rangle.$$
(5.1)

Note that this equation is not well-defined because the operator X is defined modulo $2\pi r$. A well defined operator is obtained by putting $V = e^{iX/r}$, which satisfies

$$V|n\rangle = e^{\frac{i2\pi}{d}n}|n\rangle = q^n|n\rangle, \quad \text{where} \quad q = e^{\frac{2\pi i}{d}}.$$
 (5.2)

By the standard interpretation of quantum mechanics, V can be regarded as the unitary translation operator in momentum space. On the other hand, the translation operator in position space should be defined by

$$U|n\rangle = |n+1\rangle, \ n = 0, 1, 2, ..., d-2,$$
(5.3)

$$U|d-1\rangle = |0\rangle.$$
(5.4)

It follows that U and V satisfy [104]

$$U^d = V^d = 1, (5.5)$$

$$VU = qUV$$
, where $q = e^{\frac{2\pi i}{d}}$, (5.6)

$$U^{\dagger} = U^{-1}, \ V^{\dagger} = V^{-1}, \tag{5.7}$$

which can be taken as the defining relations of quantum mechanics with d points in periodic position space and periodic momentum space. Equation (5.6) is usually taken as the starting point of quantum mechanics in a space with a finite number of points. It can be shown that by taking the limit where the number of points is infinite, the Heisenberg commutation relation for P and X is obtained [173]. Thus one can intuitively think that the correct choice of the Hamiltonian at the corresponding continuum limit will result in our calculations in this chapter. However, until now, the formulation of the Hamiltonian has not been considered.

Taking Equation (5.6) as the starting point necessarily leads to periodicity in position space and momentum space. In this chapter we will show that it is also possible to have quantum mechanics in a space with a finite number of points where the space is not periodic.

In standard quantum mechanics both the translation operator in position space denoted by U and the translation operator in momentum space denoted by V are unitary. In terms of the momentum operator P and the position operator X,

$$U(a) = e^{\frac{-iaP}{\hbar}},\tag{5.8}$$

$$V(b) = e^{\frac{ibX}{\hbar}},\tag{5.9}$$

$$V(b)U(a) = e^{\frac{iab}{\hbar}}U(a)V(b), \qquad (5.10)$$

where P and X are well-defined Hermitian operators. However for the discrete finite case only U and V are well-defined.

Bonatsos et al. [174] have considered the position and momentum operators for the q-deformed oscillator with q being a root of unity. They have shown that the phase space of this oscillator has a lattice structure, which is a non-uniformly distributed grid. In contrast, in this chapter we assume that the grid is uniform.

We define a deformed momentum operator

$$\tilde{P} = \frac{2\pi\hbar}{ad} \frac{U^{-1/2} - U^{1/2}}{q^{1/2} - q^{-1/2}}, \quad \text{where} \quad q = e^{\frac{2\pi i}{d}}, \quad U = e^{\frac{-iaP}{\hbar}}, \quad (5.11)$$

where in the last equation P is not well-defined since momentum is periodic. As $a \to 0$ and $d \to \infty$ so that ad = finite

$$\tilde{P} = P + \mathcal{O}(P^3). \tag{5.12}$$

Although the half integer power in the exponent may look problematic, for the choice of the free particle Hamiltonian the half integer power is irrelevant. Thus our choice of the Hamiltonian is given by

$$H = \frac{1}{2M}\tilde{P}^2 = \frac{\pi^2\hbar^2}{(ad)^2M} \frac{[2 - (U + U^{\dagger})]}{2sin^2(\pi/d)},$$
(5.13)

where M is the mass of the particle.

For a Hamiltonian to be acceptable it should reduce to $\frac{p^2}{2m}$ in the continuum limit. Our choice of the Hamiltonian is not unique. However, it is the simplest Hamiltonian which gives $\frac{p^2}{2m}$ in the continuum limit because it contains the first power of U and U^{\dagger} .

5.1. Nonunitary Translation Operators

We would also like to address the quantum mechanics when the position space is not periodic. We consider a position space of d points where a particle located at position X = na is described by the ket vector $|n\rangle$, n = 0, 1, ..., d - 1 where

$$X\left|n\right\rangle = na\left|n\right\rangle.\tag{5.14}$$

We regard the points x = 0 and x = (d-1)a as the end points of this discrete position space. We have to decide how the translation operator acts at the end points. For the right translation we define an operator u_+ whose action on the position eigenstates $|n\rangle$ is given by

$$u_{+} |n\rangle = |n+1\rangle, \qquad n = 0, 1, ..., d-2,$$
 (5.15)

$$u_+ |d-1\rangle = 0. (5.16)$$

Similarly, the left translation operator will be denoted u_{-} which satisfies

$$u_{-}|n\rangle = |n-1\rangle, \qquad n = 1, ..., d-1,$$
(5.17)

$$u_{-}\left|0\right\rangle = 0.\tag{5.18}$$

Note that this is the simplest kind of generalized oscillator which has a finite number of states. The most well-known of these generalized oscillators is the Biedenharn-Macfarlane oscillator [175, 176] with the spectrum

$$a^{\dagger}a = \frac{q^N - q^{-N}}{q - q^{-1}},\tag{5.19}$$

with real q. The spectrum also becomes well defined when q is a root of unity. The representations of the q-deformed oscillator algebra [177] with q a root of unity are discussed in [178].

$$u_{+}^{d} = u_{-}^{d} = 0, (5.20)$$

$$u_{+}^{\dagger} = u_{-}, \tag{5.21}$$

$$u_{-}u_{+} = 1 - u_{+}^{d-1}u_{-}^{d-1}, (5.22)$$

$$u_{+}u_{-} = 1 - u_{-}^{d-1}u_{+}^{d-1}.$$
(5.23)

Because our space has a finite number of points with $0 \le n \le d-1$,

$$u_{+}^{d} |n\rangle = u_{-}^{d} |n\rangle = 0.$$
(5.24)

Thus Equation (5.20) is satisfied. One can easily define u_+ and u_- in terms of position eigenstates and can arrive at Equation (5.21) because these states are orthonormal.

Then we check the unitarity of the operators

$$u_{+}u_{+}^{\dagger}|n\rangle = u_{+}u_{-}|n\rangle \neq |n\rangle, \quad \text{only when} \quad |n\rangle = |0\rangle, \quad (5.25)$$

$$u_{-}u_{-}^{\dagger}|n\rangle = u_{-}u_{+}|n\rangle \neq |n\rangle$$
, only when $|n\rangle = |d-1\rangle$, (5.26)

otherwise

$$u_{+}u_{+}^{\dagger}|n\rangle = u_{+}u_{-}|n\rangle = |n\rangle, \qquad n = 1, 2, ...d - 1,$$
(5.27)

$$u_{-}u_{-}^{\dagger}|n\rangle = u_{-}u_{+}|n\rangle = |n\rangle, \qquad n = 0, 1, ...d - 2.$$
 (5.28)

This means that we should have an additional term that breaks unitarity. This is the second term on the right hand side of Equations (5.22) and (5.23). One can easily show that Equations (5.22) and (5.23) are not only satisfied at the end states but also satisfied at intermediate states. The minimal set of relations which define the algebra

generated by u and u^{\dagger} is given by

$$uu^{\dagger} = 1 - u^{\dagger d - 1} u^{d - 1}$$
, and $u^{d} = 0$,

where $u_+ = u^{\dagger}$ and $u_- = u$.

Now we will show that these relations imply a unique representation with the desired properties and lead to a set of relations satisfied by u and u^{\dagger} . We define two useful operators

$$P_0 = 1,$$
 (5.29)

$$P_n = u^{\dagger n} u^n. \tag{5.30}$$

If we can show that

$$P_n P_m = P_m, \qquad m \ge n,\tag{5.31}$$

we will have the inclusion relation

$$P_n^2 = P_n. (5.32)$$

Thus we can consider them as the projection operators. We will prove Equations (5.31) by induction. For n = 1, we write

$$P_1 P_m = u^{\dagger} u u^{\dagger m} u^m,$$

= $u^{\dagger} (u u^{\dagger}) u^{\dagger m - 1} u^m,$ (5.33)

using (5.22)

$$P_1 P_m = u^{\dagger} (1 - P_{d-1}) u^{\dagger m - 1} u^m, \qquad (5.34)$$

and using (5.20)

$$P_1 P_m = u^{\dagger m} u^m = P_m. (5.35)$$

For n = l + 1, we have

$$P_{l+1}P_m = u^{\dagger}P_l(uu^{\dagger})P_{m-1}u,$$

= $u^{\dagger}P_l(1 - P_{d-1})P_{m-1}u,$
= $u^{\dagger}P_lP_{m-1}u - u^{\dagger}P_lP_{d-1}P_{m-1}u,$ where $l+1 \le m.$ (5.36)

Assuming (5.31) holds for n = l and remembering $l + 1 \le m$ implies $l \le m - 1$

$$P_{l+1}P_m = u^{\dagger}P_{m-1}u - u^{\dagger}P_{d-1}P_{m-1}u, \qquad (5.37)$$

using $u^{\dagger}P_{d-1} = 0$,

$$P_{l+1}P_m = u^{\dagger}P_{m-1}u = P_m. (5.38)$$

Thus the proof is complete.

Before going further, we will supply two more useful equations

$$P_{m}u^{\dagger} = u^{\dagger m}u^{m-1}(uu^{\dagger}),$$

= $u^{\dagger}P_{m-1}(1 - P_{d-1}),$
= $u^{\dagger}P_{m-1} - u^{\dagger}P_{d-1},$
= $u^{\dagger}P_{m-1},$ (5.39)

and

$$uP_{m} = (uu^{\dagger})P_{m-1}u,$$

= $(1 - P_{d-1})P_{m-1}u,$
= $P_{m-1}u - P_{d-1}u,$
= $P_{m-1}u.$ (5.40)

By replacing m with m + 1, we obtain

$$P_m u = u P_{m+1}.$$
 (5.41)

Now we will prove

$$u^n u^{\dagger n} = 1 - P_{d-n}, (5.42)$$

which implies the last part of our algebra (5.23) when n = d - 1

$$u^{d-1}u^{\dagger d-1} = 1 - u^{\dagger}u.$$

Our method is proof by induction. For n = 1

$$uu^{\dagger} = 1 - P_{d-1}, \tag{5.43}$$

which is (5.22).

For n = k + 1, we have

$$u^{k+1}u^{\dagger k+1} = u(u^k u^{\dagger k})u^{\dagger}.$$
 (5.44)

For n = k we assume $u^k u^{\dagger k} = 1 - P_{d-k}$. By substituting this into the last equation, we obtain

$$u^{k+1}u^{\dagger k+1} = u(1 - P_{d-k})u^{\dagger},$$

= $uu^{\dagger} - uP_{d-k}u^{\dagger}.$ (5.45)

Using (5.22) for the first term on the RHS and (5.41) for the second term on the RHS, we get

$$u^{k+1}u^{\dagger k+1} = 1 - P_{d-1} - P_{d-k-1}uu^{\dagger}, (5.46)$$

using (5.22) once more for the term uu^{\dagger} ,

$$u^{k+1}u^{\dagger k+1} = 1 - P_{d-1} - P_{d-k-1}(1 - P_{d-1}),$$
(5.47)

remembering $P_n P_m = P_m$ for $m \ge n$, which was our first proved equation

$$u^{k+1}u^{\dagger k+1} = 1 - P_{d-1} - P_{d-k-1} + P_{d-1},$$

$$u^{k+1}u^{\dagger k+1} = 1 - P_{d-(k+1)} \quad \text{Q.E.D.}.$$
 (5.48)

5.2. Eigenvalues and Eigenfunctions of the Hamiltonian for the Nonperiodic Space

In a manner similar to (5.13), we choose the Hamiltonian for the non-periodic case

$$H = \frac{\pi^2 \hbar^2}{(ad)^2 M} \frac{[2 - (u + u^{\dagger})]}{2sin^2(\pi/d)}.$$
(5.49)

We now calculate the eigenvalues and eigenvectors of $(u + u^{\dagger})$

$$\left|\lambda\right\rangle = \sum_{n=0}^{d-1} \lambda_n \left|n\right\rangle,\tag{5.50}$$

$$(u+u^{\dagger})|\lambda\rangle = \sum_{n=0}^{d-1} (\lambda_n |n-1\rangle + \lambda_n |n+1\rangle) = \sum_{n=0}^{d-1} \lambda \lambda_n |n\rangle, \qquad (5.51)$$

 \mathbf{SO}

$$\lambda_{n+1} + \lambda_{n-1} = \lambda \lambda_n$$
, with $\lambda_{-1} = 0$, and $\lambda_d = 0$. (5.52)

We obtain the general solution $\lambda_n = Ae^{in\theta} + Be^{-in\theta}$, where $\cos\theta = \lambda/2$. Using the boundary conditions $\lambda_{-1} = 0$ and $\lambda_d = 0$, we get $B = -Ae^{-2i\theta}$ and $\theta = \frac{\pi m}{d+1}$ where m = 1, 2, ..., d. m = 0 is excluded because it does not yield a non-zero eigenvector. Thus for each value of n, we obtain m different eigenvalues, given by

$$\lambda_n^{(m)} = A_m(e^{i\alpha nm} - e^{-2i\alpha m}e^{-i\alpha nm}) \quad \text{where} \quad \alpha = \pi/(d+1), \tag{5.53}$$

$$=A_m 2ie^{\frac{-im\pi}{1+d}} sin[\frac{m(1+n)\pi}{1+d}].$$
(5.54)

If we normalize $|\lambda^{(m)}\rangle$, we obtain

$$A_m = \sqrt{\frac{1}{2(d+1)}}.$$
 (5.55)

As a result the eigenvectors are written as

$$\left|\lambda^{(m)}\right\rangle = \frac{2ie^{\frac{-im\pi}{1+d}}}{\sqrt{2(d+1)}} \sum_{n=0}^{d-1} \sin\left[\frac{m(1+n)\pi}{1+d}\right] \left|n\right\rangle,\tag{5.56}$$

and the corresponding eigenvalues are given by

$$\lambda^{(m)} = 2\cos\theta = 2\cos(\frac{\pi m}{d+1})$$
 with $m = 1, 2, ...d.$ (5.57)

Now, it is time to compare what we have found with the usual quantum particle in an infinite one-dimensional square well. We calculate the eigenvalues of the Hamiltonian, by applying (5.49) to eigenvectors given by (5.56). We find

$$E_m = \frac{2\pi^2 \hbar^2}{(ad)^2 M} \frac{\sin^2[\frac{\pi m}{2(d+1)}]}{\sin^2(\frac{\pi}{d})} \quad \text{where} \quad m = 1, 2, ..., d.$$
(5.58)

In the continuum limit $a \to 0$, $d \to \infty$ and a(d-1) = L, these energy eigenvalues become

$$E_m = \frac{\pi^2 \hbar^2 m^2}{2ML^2},$$
(5.59)

which are the same energy eigenvalues in the case of a particle confined in an infinite one dimensional square well of width L. The wave functions are given by

$$\psi_m(n) = C_n^{(m)} \left\langle n \right| \lambda^{(m)} \right\rangle = C_n^{(m)} \sqrt{\frac{2}{d+1}} \sin[\frac{m(1+n)\pi}{d+1}].$$
(5.60)

We can easily find $C_n^{(m)}$ by normalization of the wave function,

$$\sum_{n=0}^{d-1} |\psi_m(n)|^2 a = 1, \qquad C_n^{(m)} = \frac{1}{\sqrt{a}}.$$
(5.61)

Thus the wave functions can be written as

$$\psi_m(n) = \sqrt{\frac{2}{(d+1)a}} \sin[\frac{m(1+n)\pi}{d+1}] \quad \text{where} \quad m = 1, 2, ..., d.$$
(5.62)

In the continuum limit the wave function becomes

$$\psi_m(x) = \sqrt{\frac{2}{L}} sin[\frac{m\pi x}{L}], \qquad (5.63)$$

which agrees with the wave functions of a square well in the interval $0 \leq x \leq L$.

5.3. Eigenvalues and Eigenfunctions of the Hamiltonian for the Periodic Space

For the periodic case, we need to find eigenvalues of $(U + U^{\dagger})$ using the position basis. Our choice is the same as (5.50). If we apply $(U + U^{\dagger})$ to this eigenvector, we will have

$$\lambda_{n+1} + \lambda_{n-1} = \lambda \lambda_n \quad \text{with} \quad \lambda_0 = \lambda_d.$$
 (5.64)

Since the equation is linear, we can superpose linearly independent solutions to find the general solution. Since we desire the linearly independent solutions to be real in order to be able to plot the wave function, we choose λ_n as

$$\lambda_n = C\cos(n\theta) + D\sin(n\theta), \tag{5.65}$$

which satisfies (5.64) with $\lambda = 2\cos\theta$. If we impose the boundary condition $\lambda_n = \lambda_{d+n}$, we obtain $\theta = \frac{2\pi m}{d}$ with m = 0, 1, ..., d-1 and

$$\left|\lambda^{(m)}\right\rangle = \sum_{n=0}^{d-1} \left[C\cos\left(\frac{2\pi nm}{d}\right) + D\sin\left(\frac{2\pi nm}{d}\right)\right] \left|n\right\rangle.$$
(5.66)

At this point there are options regarding the choice of C and D. By introducing a phase angle we can write this equation as

$$\left|\lambda^{(m)}\right\rangle = \sum_{n=0}^{d-1} C_m \cos\left(\frac{2\pi nm}{d} - \gamma_m\right) \left|n\right\rangle \quad \text{where} \quad \gamma_m = \frac{2\pi n_0 m}{d}.$$
 (5.67)

By selecting n_0 and $n = -\frac{d-1}{2}, -\frac{d-1}{2}+1, ..., \frac{d-1}{2}$, the solutions become parity eigenstates. Thus parity corresponds to $n \to -n$. As a result, $n_0 = \frac{d-1}{2}$ and for normalized eigenvectors we get

$$C_m = \begin{cases} \sqrt{\frac{1}{d}} & \text{if } m = 0, \\ \sqrt{\frac{2}{d}} & \text{if } m \neq 0. \end{cases}$$
(5.68)

The final form of eigenvectors is given by

$$\left|\lambda^{(m)}\right\rangle = C_m \sum_{n=0}^{d-1} \cos\left[\frac{2\pi nm}{d} - \frac{2\pi (d-1)m}{2d}\right] \left|n\right\rangle,$$
 (5.69)

and the eigenvalues are given by

$$\lambda^{(m)} = 2\cos\theta,$$

$$= 2\cos(\frac{2\pi m}{d}) \quad \text{where} \quad m = \begin{cases} 0, 1, ..., int(\frac{d-1}{2}) & \text{for positive parity states,} \\ 1, 2, ..., int(\frac{d}{2}) & \text{for negative parity states.} \end{cases}$$
(5.70)

Hence we find the energy eigenvalues as

$$E_m = \frac{2\pi^2 \hbar^2}{(ad)^2 M} \frac{\sin^2[\frac{m\pi}{d}]}{\sin^2(\frac{\pi}{d})}.$$
 (5.71)

In the continuum limit, we set ad = L and find

$$E_m = \frac{2\pi^2 \hbar^2 m^2}{ML^2},$$
 (5.72)

which are the same as energy eigenvalues in the case of particle in a box with periodic boundary conditions.

We continue by obtaining the wave functions. Applying the same method that we used for the nonperiodic case, we find that the ground state is unique and given by

$$\psi_0(n) = \sqrt{\frac{1}{da}}.\tag{5.73}$$

For odd d the excited states are doubly degenerate, whereas for even d the excited states are doubly degenerate except the highest energy state. The eigenvalues of the position operator are given by

$$X|n\rangle = an|n\rangle, \qquad n = -\frac{d-1}{2}, -\frac{d-1}{2} + 1, ..., \frac{d-1}{2}.$$
 (5.74)

Then the positive parity excited states are given by the wave functions

$$\psi_m^+(n) = \sqrt{\frac{2}{da}}\cos(\frac{2\pi mn}{d}), \qquad m = 1, 2, ..., int(\frac{d-1}{2}),$$
(5.75)

and the negative parity excited states are given by the wave functions

$$\psi_m^-(n) = \sqrt{\frac{2}{da}} sin(\frac{2\pi mn}{d}), \qquad m = 1, 2, ..., int(\frac{d}{2}).$$
 (5.76)

If one plots the wave functions, it is observed that the wave functions of the omitted m values are the same as the wave functions of the indicated m values. The total number of states, including the ground state, is d.

At the continuum limit, the ground state wave function, the even parity wave functions and the odd parity wave functions become, respectively,

$$\psi_0^+(x) = \sqrt{\frac{1}{L}},\tag{5.77}$$

$$\psi_m^+(x) = \sqrt{\frac{2}{L}} \cos[\frac{2m\pi x}{L}],$$
(5.78)

$$\psi_m^-(x) = \sqrt{\frac{2}{L}} sin[\frac{2m\pi x}{L}],$$
 (5.79)

which are the even parity solutions and the odd parity solutions for a particle in a box in an interval $-L/2 \le x \le L/2$ with periodic boundary conditions.

5.4. Discussion

The d = 2 case gives the fermionic oscillator. For this case u^{\dagger} and u correspond to the creation and annihilation operators for a single fermionic degree of freedoom. For the periodic case Schwinger [179] has shown that for d = 2, U and V in fact generate the Pauli algebra. Our work shows that even for the non-periodic case, d = 2 leads to Pauli-matrices through the relations,

$$\sigma_1 = u^{\dagger} + u, \tag{5.80}$$

$$\sigma_2 = i(u^{\dagger} - u), \tag{5.81}$$

$$\sigma_3 = uu^{\dagger} - u^{\dagger}u, \tag{5.82}$$

(For details of these equations see Appendix D).

We should also note that a universe with a finite number of points may be a physical reality. The size of the visible universe is 10^{27} m, whereas the smallest classical length is the Planck length, 10^{-35} m. This means that the physical space continuum can be regarded as consisting of $d_{universe}/l_{planck} = 10^{62}$ points lying along one dimension. Thus, it may be that what we call the space continuum can be described by quantum mechanics with $d = 10^{62}$ points along one dimension. Another possibility is that in a Klauza-Klein like theory [180] the internal space can consist of a finite number of points. One can also reach our calculations which are presented in this chapter in [181].
6. QUANTUM MECHANICS ON PERIODIC AND NON-PERIODIC LATTICES AND ALMOST UNITARY SCHWINGER OPERATORS

Schwinger considered a periodic lattice on which the translation operator U is unitary due to the periodicity of the lattice. On such a lattice the position can be again expressed by a unitary operator V such that

$$VU = qUV \quad \text{where} \quad q = e^{\frac{2\pi i}{d}}, \tag{6.1}$$
$$V^{d} = U^{d} = 1 \quad \text{and} \quad VV^{\dagger} = UU^{\dagger} = 1,$$

where d is the number of points on the periodic lattice. Schwinger chose the integer d to be prime and in this case the relation $q = e^{\frac{2\pi i}{d}}$ can be omitted since it is already implied by the other equations.

In the previous chapter and in [181] it has been shown that a finite lattice has an almost unitary quasi-translation operator a which satisfies

$$aa^{\dagger} = 1 - a^{\dagger d - 1} a^{d - 1}, \quad \text{and} \quad a^{\dagger d} = 0,$$
(6.2)

$$a^{\dagger}a = 1 - a^{d-1}a^{\dagger d-1}, \quad \text{and} \quad a^{d} = 0.$$
 (6.3)

The operators a^{\dagger} and a in the above relations can be respectively regarded as the right quasi-translation operator and the left quasi-translation operator since an end point can be translated only in one direction. A point which lies at the right end of the finite lattice can only be translated left end vice versa. Equation (6.2) gives the minimal set of relations that define the algebra generated by a and a^{\dagger} . The second set of the relations written in equation (6.3) can be derived using (6.2). Although the algebras defined by Equation (6.1) and by equation (6.2) look very different, physically they accomplish basically the same concept. Therefore the exact mathematical relation between them should be unveiled.

It is obvious that the translation operators U and U^{\dagger} of the Schwinger algebra commute. On the other hand quasi-translation operators a and a^{\dagger} do not commute as can be seen from equations (6.2) and (6.3). In this sense discrete non-periodic space is a deformation of discrete cyclic space. Deformation emerges as a result of having the end points.

6.1. Mathematical Structure of the Almost Unitary Translation Operators

In the previous chapter [181] projection operators P_n were defined as,

$$P_n = a^{\dagger n} a^n, \quad \text{where} \quad P_0 = 1, \tag{6.4}$$

and it was shown that

$$a^n a^{\dagger n} = 1 - P_{d-n}. \tag{6.5}$$

Thus we can also define another projection operator such as,

$$R_n = a^n a^{\dagger n}, \quad \text{where} \quad R_0 = 1. \tag{6.6}$$

Therefore one can easily see the following relations between the projection operators P_n and R_n

$$P_n = 1 - R_{d-n}$$
 and $R_n = 1 - P_{d-n}$. (6.7)

Their properties which are calculated in the Appendix E are summarized as,

$$P_n = a^{\dagger n} a^n, \quad P_0 = 1, \quad P_m a^{\dagger} = a^{\dagger} P_{m-1}, \quad a P_m = P_{m-1} a, \tag{6.8}$$
$$P_n P_m = P_j \quad \text{where} \quad j = \max(n, m),$$
$$P_m a^n = a^{\dagger n} P_m = 0 \quad \text{for} \quad n+m \ge d,$$

and

$$R_n = a^n a^{\dagger n}, \quad R_0 = 1, \quad R_m a^{\dagger} = a^{\dagger} R_{m+1}, \quad a R_m = R_{m+1} a, \tag{6.9}$$
$$R_n R_m = R_j \quad \text{where} \quad j = \max(n, m),$$
$$a^n R_m = R_m a^{\dagger n} = 0 \quad \text{for} \quad n+m \ge d.$$

In the previous chapter and also in [181] we have already considered a position space of d points where a particle located at position $X = \beta n$ is described by the ket vector $|n\rangle$, n = 0, 1, ..., d - 1 where

$$X\left|n\right\rangle = \beta n\left|n\right\rangle,\tag{6.10}$$

with β as the grid spacing. We can define the position operator as

$$X = \beta \sum_{m=1}^{d-1} P_m,$$

$$X = \beta \{ a^{\dagger}a + \dots + a^{\dagger n}a^n + \dots + a^{\dagger d-1}a^{d-1} \}.$$
(6.11)

Applying this to $|n\rangle$ one gets the desired result.

6.2. The Schwinger Algebra in Terms of the Almost Unitary Translation Operators

We define unitary operators U and V that cyclically permute the vectors of a given system in terms of almost unitary operators a and a^{\dagger}

$$U = a^{\dagger} + a^{d-1}, \tag{6.12}$$

$$V = \sum_{n=0}^{d-1} q^n (P_n - P_{n+1}).$$
(6.13)

Then we will show that these definitions satisfy the Schwinger algebra given by Equation (6.1). The first relation we will prove is

$$VV^{k} = \sum_{n=0}^{d-1} q^{(k+1)n} (P_n - P_{n+1}), \qquad (6.14)$$

which has the inclusion relation

$$V^{d} = VV^{d-1} = \sum_{n=0}^{d-1} q^{dn} (P_n - P_{n+1})$$

$$= \sum_{n=0}^{d-1} (P_n - P_{n+1})$$

$$= (1 + a^{\dagger}a + a^{\dagger 2}a^2 + \dots + a^{\dagger d-1}a^{d-1}) - (a^{\dagger}a + a^{\dagger 2}a^2 + \dots + a^{\dagger d-1}a^{d-1} + a^{\dagger d}a^d)$$

$$= 1,$$

$$(6.15)$$

where we used the algebra relation $a^{\dagger d} = 0$.

We will prove the equation (6.14) using the method of proof by induction.

For k = 1 we have

$$VV = \sum_{\substack{n,m=0\\n,m=0}}^{d-1} q^{n+m} (P_n - P_{n+1}) (P_m - P_{m+1}), \qquad (6.16)$$

$$= \sum_{\substack{n,m=0\\n\geq m+1}}^{d-1} q^{n+m} (P_n P_m - P_n P_{m+1} - P_{n+1} P_m + P_{n+1} P_{m+1}),$$

$$= \sum_{\substack{n,m=0\\n\geq m+1}}^{d-1} q^{n+m} (P_n P_m - P_n P_{m+1} - P_{n+1} P_m + P_{n+1} P_{m+1}),$$

$$+ \sum_{\substack{n,m=0\\n< m}}^{d-1} q^{n+m} (P_n P_m - P_n P_{m+1} - P_{n+1} P_m + P_{n+1} P_{m+1}).$$

Using the property $P_n P_m = P_j$ where $j = \max(n, m)$, we obtain

$$VV = 0 + \sum_{\substack{n,m=0\\n=m}}^{d-1} q^{n+m} (P_n - P_{m+1}) + 0, \qquad (6.17)$$
$$= \sum_{n=0}^{d-1} q^{2n} (P_n - P_{n+1}).$$

We assume that for k = l

$$VV^{l} = \sum_{n=0}^{d-1} q^{(l+1)n} (P_n - P_{n+1}).$$
(6.18)

For k = l + 1,

$$VV^{l+1} = (VV^l)V,$$

$$= \sum_{n=0}^{d-1} q^{(l+1)n} (P_n - P_{n+1}) \sum_{m=0}^{d-1} q^m (P_m - P_{m+1}),$$
(6.19)

$$=\sum_{\substack{n,m=0\\n\geq m+1}}^{d-1} q^{(l+1)n+m} (P_n P_m - P_n P_{m+1} - P_{n+1} P_m + P_{n+1} P_{m+1}),$$

$$=\sum_{\substack{n,m=0\\n\geq m+1}}^{d-1} q^{(l+1)n+m} (P_n P_m - P_n P_{m+1} - P_{n+1} P_m + P_{n+1} P_{m+1}),$$

$$+\sum_{\substack{n,m=0\\n=m}}^{d-1} q^{(l+1)n+m} (P_n P_m - P_n P_{m+1} - P_{n+1} P_m + P_{n+1} P_{m+1}),$$

$$+\sum_{\substack{n,m=0\\n=m}}^{d-1} q^{(l+1)n+m} (P_n P_m - P_n P_{m+1} - P_{n+1} P_m + P_{n+1} P_{m+1}).$$

Using the same property $P_n P_m = P_j$ where $j = \max(n, m)$, we obtain

$$VV^{l+1} = 0 + \sum_{\substack{n,m=0\\n=m}}^{d-1} q^{(l+1)n+m} (P_n - P_{m+1}) + 0, \qquad (6.20)$$
$$= \sum_{n=0}^{d-1} q^{(l+2)n} (P_n - P_{n+1}) \quad \text{Q.E.D.}.$$

Then we will show that $VV^{\dagger} = 1$ by using definition of V.

Since $P_n = P_n^{\dagger}$ and $q^{\dagger n} = q^{d-n}$, we have

$$VV^{\dagger} = \sum_{m=0}^{d-1} \sum_{n=0}^{d-1} q^{m} q^{d-n} (P_{m} - P_{m+1}) (P_{n} - P_{n+1}), \qquad (6.21)$$
$$= \sum_{m,n=0}^{d-1} q^{m} q^{d-n} (P_{m} P_{n} - P_{m} P_{n+1} - P_{m+1} P_{n} + P_{m+1} P_{n+1}),$$
$$= \sum_{\substack{m,n=0\\m \ge n+1}}^{d-1} q^{m+d-n} (P_{m} P_{n} - P_{m} P_{n+1} - P_{m+1} P_{n} + P_{m+1} P_{n+1}),$$
$$+ \sum_{\substack{m,n=0\\m = n}}^{d-1} q^{m+d-n} (P_{m} P_{n} - P_{m} P_{n+1} - P_{m+1} P_{n} + P_{m+1} P_{n+1}),$$

$$+ \sum_{\substack{m,n=0\\m
= $0 + \sum_{m=0}^{d-1} q^d (P_m - P_{m+1}) + 0,$
= $(1 + a^{\dagger}a + a^{\dagger 2}a^2 + \dots + a^{\dagger d-1}a^{d-1}) - (a^{\dagger}a + a^{\dagger 2}a^2 + \dots + a^{\dagger d-1}a^{d-1} + a^{\dagger d}a^d),$
= 1 Q.E.D..$$

Next, we will show that

$$UU^n = a^{\dagger n+1} + a^{d-(n+1)}$$
 with $n = 0, 1, \cdots, d-1,$ (6.22)

which implies

$$U^{d} = UU^{d-1},$$

$$= a^{\dagger d} + a^{d-d},$$

$$= 1.$$
(6.23)

Our method is proof by induction. For n = 1, we have

$$UU = (a^{\dagger} + a^{d-1})(a^{\dagger} + a^{d-1}),$$

$$= a^{\dagger}a^{\dagger} + a^{\dagger}a^{d-1} + a^{d-1}a^{\dagger} + a^{d-1}a^{d-1},$$

$$= a^{\dagger 2} + (a^{\dagger}a)a^{d-2} + a^{d-2}(aa^{\dagger}),$$

$$= a^{\dagger 2} + P_{1}a^{d-2} + a^{d-2}R_{1},$$

$$= a^{\dagger 2} + a^{d-2}P_{1+d-2} + a^{d-2}(1 - P_{d-1}),$$

$$= a^{\dagger 2} + a^{d-2}P_{d-1} + a^{d-2} - a^{d-2}P_{d-1},$$

$$= a^{\dagger 2} + a^{d-2}.$$
(6.24)

For n = l, we assume that

$$UU^{l} = a^{\dagger l+1} + a^{d-(l+1)}.$$
(6.25)

For n = l + 1, we obtain

$$UU^{l+1} = (UU^{l})U, \qquad (6.26)$$

$$= (a^{\dagger l+1} + a^{d-(l+1)})(a^{\dagger} + a^{d-1}), \qquad (6.26)$$

$$= a^{\dagger l+2} + a^{\dagger l+1}a^{d-1} + a^{d-(l+1)}a^{\dagger} + a^{2d-(l+2)}, \qquad (a^{\dagger}), \qquad (a^{\dagger})^{1/2} + (a^{\dagger l+1}a^{l+1})a^{d-1-(l+1)} + a^{d-(l+2)}(aa^{\dagger}), \qquad (a^{\dagger})^{1/2} + P_{l+1}a^{d-1-(l+1)} + a^{d-(l+2)}R_{1} \quad \text{use } (8), \qquad (a^{\dagger l+2} + a^{d-(l+2)}P_{l+1+d-l-2} + a^{d-(l+2)}(1 - P_{d-1}), \qquad (a^{\dagger l+2} + a^{d-(l+2)}P_{d-1} + a^{d-(l+2)} - a^{d-(l+2)}P_{d-1}, \qquad (a^{\dagger l+2} + a^{d-(l+2)}P_{d-1} + a^{d-(l+2)} - a^{d-(l+2)}P_{d-1}, \qquad (a^{\dagger l+2} + a^{d-(l+2)}Q.E.D..$$

The last term in the third line is zero because at most l = d - 2 according to equation (6.22).

We will obtain $UU^{\dagger} = 1$ just by substitution

$$UU^{\dagger} = (a^{\dagger} + a^{d-1})(a + a^{\dagger d-1}), \qquad (6.27)$$
$$= a^{\dagger}a + a^{\dagger d} + a^{d} + a^{d-1}a^{\dagger d-1},$$
$$= P_1 + R_{d-1},$$
$$= 1 \quad \text{Q.E.D.}.$$

where we have used (6.2), (6.3) and (6.7).

We have the formula for U and V, so we will show that VU = qUV just by substitution. Thus left hand side of the formula is equal to

$$VU = \sum_{n=0}^{d-1} q^n (P_n - P_{n+1})(a^{\dagger} + a^{d-1}),$$

$$= \sum_{n=0}^{d-1} q^n (P_n a^{\dagger} + P_n a^{d-1} - P_{n+1} a^{\dagger} - P_{n+1} a^{d-1}).$$
(6.28)

Since $P_n = a^{\dagger}a$ and $P_0 = 1$, $P_n a^{d-1} = 0$ except for n = 0 and $P_{n+1}a^{d-1} = 0$ for all n. Therefore we obtain

$$VU = \sum_{n=0}^{d-1} q^n (P_n a^{\dagger} - P_{n+1} a^{\dagger}) + q^0 a^{d-1}, \qquad (6.29)$$
$$= P_0 a^{\dagger} - P_1 a^{\dagger} + \sum_{n=1}^{d-1} q^n (a^{\dagger} P_{n-1} - a^{\dagger} P_n) + a^{d-1},$$
$$= a^{\dagger} - a^{\dagger} P_0 + \sum_{n=1}^{d-1} q^n (a^{\dagger} P_{n-1} - a^{\dagger} P_n) + a^{d-1},$$
$$= \sum_{n=1}^{d-1} q^n a^{\dagger} (P_{n-1} - P_n) + a^{d-1},$$

where we have used (6.8). To obtain right hand side of the formula VU = qUV we calculate,

$$UV = (a^{\dagger} + a^{d-1}) \sum_{n=0}^{d-1} q^n (P_n - P_{n+1}),$$

$$= \sum_{n=0}^{d-1} q^n (a^{\dagger} (P_n - P_{n+1}) + a^{d-1} (P_n - P_{n+1})),$$

$$= \sum_{n=0}^{d-1} q^n a^{\dagger} (P_n - P_{n+1}) + \sum_{n=0}^{d-1} q^n a^{d-1} (P_n - P_{n+1}),$$

$$= \sum_{n=0}^{d-1} q^n a^{\dagger} (P_n - P_{n+1}) + \sum_{n=0}^{d-1} q^n a^{d-1} (1 - R_{d-n} - (1 - R_{d-n-1})),$$

$$= \sum_{n=0}^{d-1} q^n a^{\dagger} (P_n - P_{n+1}) + \sum_{n=0}^{d-1} q^n a^{d-1} (R_{d-n-1} - R_{d-n}).$$
(6.30)

Since $R_m = a^m a^{\dagger m}$, $a^{d-1}R_m = 0$ except m = 0. The element $R_{d-n-1} = R_0 = 1$ for n = d - 1 so we have,

$$UV = \sum_{n=0}^{d-1} q^n a^{\dagger} (P_n - P_{n+1}) + q^{d-1} a^{d-1}, \qquad (6.31)$$
$$qUV = \sum_{n=0}^{d-1} q^{n+1} a^{\dagger} (P_n - P_{n+1}) + q^d a^{d-1},$$
$$= \sum_{n=1}^{d} q^n a^{\dagger} (P_{n-1} - P_n) + a^{d-1},$$
$$= \sum_{n=1}^{d-1} q^n a^{\dagger} (P_{n-1} - P_n) + a^{d-1},$$

where we have used the facts that $q^d = 1$, $P_d = 0$ and $a^{\dagger}P_{d-1} = 0$. which is the same result given by equation (6.29).

6.3. Almost Unitary Operators in Terms of the Schwinger Algebra

It is also possible to write the almost unitary operators a and a^{\dagger} in terms of U and V

$$a^{\dagger} = U - \left(\frac{1 + V + V^2 + \dots + V^{d-1}}{d}\right)U.$$
 (6.32)

The expression in the parentheses is called \mathscr{P}_0 . It is shown that \mathscr{P}_0 is a projection operator (see Appendix E). We have found that more general projection operators \mathscr{P}_n are written as

$$\mathscr{P}_n = \frac{(1+q^nV+q^{2n}V^2+\dots+q^{(d-1)n}V^{(d-1)})}{d}.$$
(6.33)

From the definition, it is easily seen that

$$\mathscr{P}_{d-l} = \mathscr{P}_{-l}$$
 and $\mathscr{P}_{d+l} = \mathscr{P}_l$ and $\mathscr{P}_d = \mathscr{P}_0.$ (6.34)

The relationships between unitary operators and projection operators are found as

$$\mathcal{P}_{n}U^{m} = U^{m}\mathcal{P}_{n+m},$$

$$U^{\dagger m}\mathcal{P}_{n} = \mathcal{P}_{n+m}U^{\dagger m},$$
(6.35)

in the Appendix E. In addition, multiplication of \mathscr{P}_n and \mathscr{P}_m is found in the Appendix E as

$$\mathscr{P}_{n}\mathscr{P}_{m} = 0 \quad \text{for } m \neq n,$$

$$\mathscr{P}_{n}\mathscr{P}_{n} = \mathscr{P}_{n}.$$
(6.36)

We will show that

$$a^{\dagger}a^{\dagger l} = U^{l+1}(1 - \mathscr{P}_1 - \mathscr{P}_2 - \dots - \mathscr{P}_{l+1}),$$
 (6.37)

which implies the following relation

$$a^{\dagger d} = a^{\dagger} a^{\dagger d-1},$$

$$= U^{d} [1 - (\mathscr{P}_{1} + \mathscr{P}_{2} + \dots + \mathscr{P}_{d})],$$

$$= U^{d} [1 - (\mathscr{P}_{0} + \mathscr{P}_{1} + \dots + \mathscr{P}_{d-1})],$$

$$= \mathbb{1}(1-1),$$

$$= 0,$$
(6.38)

where we have used (6.34) and (E.23). The equation (6.38) is the second equation of the unitary algebra defined in (6.2). We will prove the equation (6.37) by the method

$$a^{\dagger}a^{\dagger} = (U - \mathscr{P}_{0}U)(U - \mathscr{P}_{0}U), \qquad (6.39)$$
$$= U(1 - \mathscr{P}_{1})U(1 - \mathscr{P}_{1}),$$
$$= U^{2}(1 - \mathscr{P}_{2})(1 - \mathscr{P}_{1}),$$
$$= U^{2}(1 - \mathscr{P}_{1} - \mathscr{P}_{2}),$$

where we have used (6.35) and (6.36). For n = l we assume that

$$a^{\dagger}a^{\dagger l} = U^{l+1}(1 - \mathscr{P}_1 - \mathscr{P}_2 - \dots - \mathscr{P}_{l+1}).$$
 (6.40)

Therefore for n = l + 1 we obtain

$$a^{\dagger}a^{\dagger l+1} = (a^{\dagger}a^{\dagger l})a^{\dagger}, \qquad (6.41)$$
$$= U^{l+1}(1 - \mathscr{P}_1 - \mathscr{P}_2 - \dots - \mathscr{P}_{l+1})U(1 - \mathscr{P}_1),$$
$$= U^{l+2}(1 - \mathscr{P}_2 - \mathscr{P}_3 - \dots - \mathscr{P}_{l+2})(1 - \mathscr{P}_1),$$
$$= U^{l+2}(1 - \mathscr{P}_1 - \mathscr{P}_2 + \mathscr{P}_2\mathscr{P}_1 - \dots - \mathscr{P}_{l+2} + \mathscr{P}_{l+2}\mathscr{P}_1),$$
$$= U^{l+2}(1 - \mathscr{P}_1 - \mathscr{P}_2 - \dots - \mathscr{P}_{l+2}).\Box$$

Now we will show that $aa^{\dagger} = 1 - a^{\dagger d - 1}a^{d - 1}$ in terms of U and V, at the left hand side we have,

$$aa^{\dagger} = U^{\dagger}(1 - \mathscr{P}_{0})(1 - \mathscr{P}_{0})U, \qquad (6.42)$$
$$= U^{\dagger}(1 - \mathscr{P}_{0} - \mathscr{P}_{0} + \mathscr{P}_{0}\mathscr{P}_{0})U,$$
$$= U^{\dagger}(1 - \mathscr{P}_{0})U,$$
$$= U^{\dagger}U(1 - \mathscr{P}_{1}),$$
$$= 1 - \mathscr{P}_{1},$$

where we have used (6.35) and (1). By using (6.37) we easily obtain

$$a^{\dagger d-1} = U^{d-1} (1 - \mathscr{P}_1 - \dots - \mathscr{P}_{d-1}),$$

$$= U^{d-1} (1 - (\mathscr{P}_1 + \mathscr{P}_2 + \dots + \mathscr{P}_{d-1})),$$

$$= U^{d-1} (1 - (1 - \mathscr{P}_0)),$$

$$= U^{d-1} \mathscr{P}_0,$$
(6.43)

where we have used (E.23). Taking Hermitian conjugate of this equation and using hermicity property of the projection operators which is shown by (E.21) we get

$$a^{d-1} = \mathscr{P}_0 U^{\dagger d-1}. \tag{6.44}$$

Then for the right hand side of $aa^{\dagger} = 1 - a^{\dagger d - 1}a^{d - 1}$ we have

$$1 - a^{\dagger d - 1} a^{d - 1} = 1 - U^{d - 1} \mathscr{P}_0 \mathscr{P}_0 U^{\dagger d - 1}, \qquad (6.45)$$
$$= 1 - U^{d - 1} \mathscr{P}_0 U^{\dagger d - 1}, \\= 1 - U^{d - 1} U^{\dagger d - 1} \mathscr{P}_1, \\= 1 - \mathscr{P}_1, \\= a a^{\dagger},$$

at the second line we used the relations given by (6.36) and at the last line we have used (6.42). This is the first equation defining the almost unitary algebra given by (6.2).

6.4. New Representations for Basis of $M_N(C)$

The e_{ij} satisfying (6.46) form the standard basis of $M_N(C)$

$$e_{ij}e_{kl} = \delta_{jk}e_{il} \qquad e_{ij}^{\dagger} = e_{ji}. \tag{6.46}$$

In this section, we give two new representations of the basis matrices. One of them is written in terms of the almost unitary algebra as

$$e_{mn} = a^{\dagger m} R_{d-1} a^n$$
 where $R_n = a^n a^{\dagger n}$ and $m, n = 0, 1, \cdots, d-1.$ (6.47)

The other one is written in terms of Schwinger U and V operators

$$e_{mn} = \begin{cases} U^{m-n} \mathscr{P}_{d-n} & \text{for} \quad m > n, \\ \mathscr{P}_{d-n} & \text{for} \quad m = n, \\ U^{\dagger n-m} \mathscr{P}_{d-n} & \text{for} \quad m < n, \end{cases}$$
(6.48)

with

$$\mathscr{P}_n = \frac{(1+q^nV+q^{2n}V^2+\dots+q^{(d-1)n}V^{(d-1)})}{d}$$

We prove these representations satisfy (6.46) in the last part of the Appendix E.

6.5. Multi-Dimensional Lattice in Terms of Lower Dimensional Lattices

Denoting a linear lattice with d elements by \mathscr{L}_d , we can show the Cartesian product $\mathscr{L}_{d_1} \times \mathscr{L}_{d_2}$ by the dots in the following figure. Corresponding to this Cartesian product of the lattices, we have the tensor product of the algebra $\mathscr{A}_{d_2} \otimes \mathscr{A}_{d_1}$. On the Cartesian product shown in the figure, the right translation operator corresponds to $a^{\dagger} \otimes 1$ and the up translation operator corresponds to $1 \otimes a^{\dagger}$. We denote the (right) translation operator on \mathscr{A}_{d_1} , \mathscr{A}_{d_2} , \mathscr{A}_{d} respectively by $a^{\dagger}_{d_1}$, $a^{\dagger}_{d_2}$, a^{\dagger}_{d} where $d = d_1 \times d_2$ and consider \mathscr{L}_{d_1} , \mathscr{L}_{d_2} as a one one-dimensional lattice as shown by the arrows in the figure. This satisfies an isomorphism

$$\mathscr{A}_d \xrightarrow{\Delta} \mathscr{A}_{d_2} \otimes \mathscr{A}_{d_1},$$
 (6.49)

and one can write

$$\triangle(a_d^{\dagger}) = \mathbb{1}_{d_2} \otimes a_{d_1}^{\dagger} + a_{d_2}^{\dagger} \otimes a_{d_1}^{d_1 - 1}.$$

$$(6.50)$$

One immediately can check that the action of a_d^{\dagger} is given by the arrows in the figure and satisfies the correct algebraic relations.



Figure 6.1. $d_1 \times d_2$ Lattice for $d_1 = 4$ and $d_2 = 3$

Similarly, we can express the translation operator for a $d_1 \times d_2$ dimensional periodic lattice by

$$\triangle(U_d) = \mathbb{1}_{d_2} \otimes a_{d_1}^{\dagger} + a_{d_2}^{\dagger} \otimes a_{d_1}^{d_1 - 1} + a_{d_2}^{d_2 - 1} \otimes a_{d_1}^{d_1 - 1}.$$
(6.51)

6.6. Discussion

We have shown that the Schwinger algebra can also be given by almost unitary operators which are physically related to the shift operators on a finite lattice. We have named these operators as almost unitary operators because relations

$$UU^{\dagger} = 1, \quad VV^{\dagger} = 1 \quad \text{and} \quad VU = qUV \quad \text{where} \quad q = e^{\frac{2\pi i}{d}}, \tag{6.52}$$

are replaced by

$$aa^{\dagger} = 1 - a^{\dagger d - 1} a^{d - 1}$$
 and $a^{\dagger} a = 1 - a^{d - 1} a^{\dagger d - 1}$, (6.53)

and the terms $a^{\dagger d-1}a^{d-1}$, $a^{d-1}a^{\dagger d-1}$ reflect violation of unitarity for a and a^{\dagger} . For U we have the relation $U^{\dagger} = U^{-1}$ due to the periodic nature of the lattice. Similarly a and a^{\dagger} can be considered as inverse of each other except at the end points. Note that a and a^{\dagger} play the role of U and U^{\dagger} where as V can be defined in terms of a and a^{\dagger} . It takes quite an effort to construct V which is given by (6.13) in terms of a and a^{\dagger}

In usual quantum mechanics where the position and the angular momentum operators have continuous eigenvalues the following equations are equivalent to each other

$$V(p_0)U(x_0) = e^{\frac{ix_0p_0}{\hbar}}U(x_0)V(p_0),$$
(6.54)

$$[X, P] = i\hbar, \tag{6.55}$$

$$[X, U(x_0)] = x_0 U(x_0), (6.56)$$

here
$$U(x_0) = exp(\frac{-ix_0P}{\hbar}), \quad V(p_0) = exp(\frac{ip_0X}{\hbar}).$$
 (6.57)

For the discrete periodic case X is defined only modulo $2\pi r$. However U and V are well defined. Thus we have only one corresponding equation

$$VU = qUV$$
 where $q = e^{\frac{2\pi i}{d}}$. (6.58)

On the other hand for the discrete non-periodic case, the position operator X, the right quasi-translation operator a^{\dagger} and the left quasi-translation operator a are well defined. Therefore we have only one equation corresponding to (6.56)

$$[X, a^{\dagger}] = \beta a^{\dagger}, \tag{6.59}$$

where β is the grid spacing.

We have shown how to construct almost unitary translation operators a, a^{\dagger} in terms of U, V and vice versa. In addition we have found the relation between basis matrices of $M_N(C)$ and the almost unitary operators and the relation between basis matrices of $M_N(C)$ and the Schwinger algebra. Furthermore we established an isomorphism between a multi-dimensional and periodic or non periodic linear lattices. One can also reach our calculations which are presented in this chapter in [182].

7. CONCLUSION

Studies in chapter 2 and chapter 3 differ from what has been done up to now by its mathematical techniques and their wide content. Field equations which govern the dynamics of the universe form a set of differential equations. In the original form they are nonlinear with independent variable time. We switch independent variable from time to scale factor "a". Then in the second chapter field equations of scalar field cosmology with minimal coupling can be reduced to linear first order differential equations. We have succeed to solve all of them simultaneously. We have constructed the most general solution in four different forms: solution for a given V(a), solution for a given $\phi(a)$, solution for a given H(a) and solution for a given $\rho(a)$. We would like to emphasize that our results are the most general. The scalar field has been taken as an effective field which averagely represents the underlying, more basic theories. Both single component universes and double component universes have been studied. We have explored mathematical structure for exotic matter which has time varying equation of state. Combination of this exotic matter and radiation in the early universe has a mathematical turn on and turn of structure for accelerated expansion. More importantly we have explained the late time accelerated expansion of the universe by cosmic domain walls. Results have been compared with observations of type Ia supernovae by taking care of the Hubble tension and the absolute magnitude tension. We have found $\Omega_{\omega} = 0.889$ and $\Omega_m = 0.111$ where ω denotes cosmic domain walls and m denotes matter.

On the other hand in the third chapter a change of independent variable turns the field equations of Brans-Dicke cosmology into a constraint equation and a Bernoulli type differential equation which can be linearized. We have constructed the most general solutions for supplied pairs of functions; ($\phi(a)$ and $\rho(a)$), ($\phi(a)$ and V(a)) and ($\phi(a)$ and H(a)). Early epoch of the universe and late-time era of the universe have been investigated. Both dark energy and domain wall dominated universe have been studied. It has been seen that the Hubble function shrinks to H(a) of the Einstein cosmology when $\omega \gg 1$. The most significant result of this part comes from the question: Does the cosmological constant need to correspond to a constant energy density or a constant potential? We have asked the question in another way Is it possible to have a constant term in the Hubble function when there is no constant energy density and no constant potential in the universe? We have found that it is possible to formulate the constant term in the Hubble function as a combination of initial values of the universe in Brans-Dicke cosmology which DOES NOT contain a constant energy density and a constant potential term. Furthermore we have found a(t) in an exponential form.

In the fourth chapter we have studied ϕ^4 potential in Brans-Dicke cosmology. We have taken time as an independent variable and we have proceeded by using the ansatz $\phi(t) = \frac{A}{a(t)}$. The exact solution results in radiation and stiff fluid mixed universe. Then we have studied early epoch of the universe by extrapolating our result in time. Two different exponential solutions have been found. We have also studied creation of matter in the late radiation dominated era. We have derived the today's value of the deceleration parameter by a new numerical method which has been introduced in Section 4.5.

In chapter 5 we have defined a deformed kinetic energy operator for a discrete position space with a finite number of points. Eigenvalues and eigenfunctions of the Hamiltonian for the periodic space and nonperiodic space have been calculated. As expected, in the continuum limit the solution for the nonperiodic case becomes the same as the solution of an infinite one dimensional square well and the periodic case solution becomes the same as the solution of a particle in a box with periodic boundary conditions.

In chapter 6 we have presented properties of the almost unitary Schwinger operators and properties of the unitary Schwinger operators. We have found relations between them. The famous Schwinger algebra has many applications in quantum optics, quantum communications, quantum probability and Galois quantum systems. We have not discussed these in detail, however we believe that our study will contribute to these branches of physics. Our results are interesting and can be useful as they allow potential transformations between different existing systems. We hope that our work will motivate scientists to apply the differences and the relations between these two sets of operators to their research fields.

REFERENCES

- 1. Nordström, G., Relativitätsprinzip und Gravitation, S. Hirzel, 1912.
- Nordström, G., "Zur theorie der Gravitation vom Standpunkt des Relativitätsprinzips", Annalen der Physik, Vol. 347, No. 13, pp. 533–554, 1913.
- Nordström, G., "Träge und Schwere Masse in der Relativitätsmechanik", Annalen der Physik, Vol. 345, No. 5, pp. 856–878, 1913.
- Nordström, G., "Uber die Moglichkeit, das Elektromagnetische Feld und das Gravitationsfeld zu Vereinigen", *Physikalische Zeitschrift*, Vol. 15, pp. 504–506, 1914.
- Willenbrock, S. S., "Cosmology of Nordström's First Theory of Gravitation", *American Journal of Physics*, Vol. 50, No. 3, pp. 229–231, 1982.
- Jordan, P., Schwerkraft und Weltall: Grundlagen der Theoretischen Kosmologie, Vol. 107, Friedr. Vieweg & Sohn, 1955.
- Dirac, P. A., "A New Basis for Cosmology", Proceedings of the Royal Society of London. Series A, Mathematical and Physical Sciences, pp. 199–208, 1938.
- Weyl, H., "Zur Gravitationstheorie", Annalen der Physik, Vol. 359, No. 18, pp. 117–145, 1917.
- Weyl, H., "Eine Neue Erweiterung der Relativitätstheorie", Annalen der Physik, Vol. 364, No. 10, pp. 101–133, 1919.
- Eddington, A., "Preliminary Note on the Masses of the Electron, the Proton, and the Universe", *Mathematical Proceedings of the Cambridge Philosophical Society*, Vol. 27, pp. 15–19, Cambridge University Press, 1931.

- Clifton, T., P. G. Ferreira, A. Padilla and C. Skordis, "Modified Gravity and Cosmology", *Physics reports*, Vol. 513, No. 1-3, pp. 1–189, 2012.
- Bergmann, P. G., "Comments on the Scalar-Tensor Theory", International Journal of Theoretical Physics, Vol. 1, No. 1, pp. 25–36, 1968.
- Nordtvedt Jr, K., "Post-Newtonian Metric for a General Class of Scalar-Tensor Gravitational Theories and Observational Consequences.", *The Astrophysical Journal*, Vol. 161, p. 1059, 1970.
- Wagoner, R. V., "Scalar-Tensor Theory and Gravitational Waves", *Physical Review D*, Vol. 1, No. 12, p. 3209, 1970.
- Guth, A. H., "Inflationary universe: A possible Solution to the Horizon and Flatness Problems", *Physical Review D*, Vol. 23, No. 2, p. 347, 1981.
- Linde, A. D., "A New Inflationary Universe Scenario: A Possible Solution of the Horizon, Fatness, Homogeneity, Isotropy and Primordial Monopole Problems", *Physics Letters B*, Vol. 108, No. 6, pp. 389–393, 1982.
- Riess, A. G., A. V. Filippenko, P. Challis, A. Clocchiatti, A. Diercks, P. M. Garnavich, R. L. Gilliland, C. J. Hogan, S. Jha, R. P. Kirshner, B. Leibundgut, M. M. Phillips, D. Reiss, B. P. Schmidt, R. A. Schommer, R. C. Smith, J. Spyromilio, C. Stubbs, N. B. Suntzeff and J. Tonry, "Observational Evidence from Supernovae for an Accelerating Universe and a Cosmological Constant", *The Astronomical Journal*, Vol. 116, No. 3, p. 1009, 1998.
- Perlmutter, S., G. Aldering, G. Goldhaber, R. Knop, P. Nugent, P. Castro, S. Deustua, S. Fabbro, A. Goobar, D. Groom, I. Hook, A. Kim, M. Kim, J. Lee, N. Nunes, R. Pain, C. Pennypacker, R. Quimby, C. Lidman, R. Ellis, M. Irwin, R. McMahon, P. Ruiz-Lapuente, N. Walton, B. Schaefer, B. Boyle, A. Filippenko, T. Matheson, A. Fruchter, Panagia, H. Newberg and W. Couch, "Measurements"

of Ω and Λ from 42 High-Redshift Supernovae", *The Astrophysical Journal*, Vol. 517, No. 2, p. 565, 1999.

- Tonry, J. L., B. P. Schmidt, B. Barris, P. Candia, P. Challis, A. Clocchiatti, A. L. Coil, A. V. Filippenko, P. Garnavich, C. Hogan, S. Holland, S. Jha, R. Kirshner, K. Krisciunas, B. Leibundgut, W. Li, T. Matheson, M. Phillips, A. Riess, R. Schommer, R. Smith, J. Sollerman, J. Spyromilio, C. Stubbs and N. Suntzeff, "Cosmological Results from High-z Supernovae", *The Astrophysical Journal*, Vol. 594, No. 1, p. 1, 2003.
- Weinberg, S., "The Cosmological Constant Problem", Reviews of modern physics, Vol. 61, No. 1, p. 1, 1989.
- Copeland, E. J., M. Sami and S. Tsujikawa, "Dynamics of Dark Energy", International Journal of Modern Physics D, Vol. 15, No. 11, pp. 1753–1935, 2006.
- Magana, J. and T. Matos, "A Brief Review of the Scalar Field Dark Matter Model", *Journal of Physics: Conference Series*, Vol. 378, p. 012012, IOP Publishing, 2012.
- Brans, C. and R. H. Dicke, "Mach's Principle and a Relativistic Theory of Gravitation", *Physical Review*, Vol. 124, No. 3, p. 925, 1961.
- Zee, A., "Broken-Symmetric Theory of Gravity", *Physical Review Letters*, Vol. 42, No. 7, p. 417, 1979.
- Smolin, L., "Towards a Theory of Spacetime Structure at Very Short Distances", *Nuclear Physics B*, Vol. 160, No. 2, pp. 253–268, 1979.
- Spokoiny, B., "Inflation and Generation of Perturbations in Broken-Symmetric Theory of Gravity", *Physics Letters B*, Vol. 147, No. 1-3, pp. 39–43, 1984.
- 27. Fakir, R. and W. G. Unruh, "Induced-Gravity Inflation", *Physical Review D*,

Vol. 41, No. 6, p. 1792, 1990.

- Accetta, F. S., D. J. Zoller and M. S. Turner, "Induced-Gravity Inflation", *Phys*ical Review D, Vol. 31, No. 12, p. 3046, 1985.
- Kaiser, D. I., "Constraints in the Context of Induced-Gravity Inflation", *Physical Review D*, Vol. 49, No. 12, p. 6347, 1994.
- Kaiser, D., "Induced-Gravity Inflation and the Density Perturbation Spectrum", arXiv preprint astro-ph/9405029, 1994.
- Cervantes-Cota, J. L. and H. Dehnen, "Induced Gravity Inflation in the Standard Model of Particle Physics", arXiv preprint astro-ph/9505069, 1995.
- La, D. and P. J. Steinhardt, "Extended Inflationary Cosmology", *Physical Review Letters*, Vol. 62, No. 4, p. 376, 1989.
- Steinhardt, P. J. and F. S. Accetta, "Hyperextended Inflation", *Physical Review Letters*, Vol. 64, No. 23, p. 2740, 1990.
- Holman, R., E. W. Kolb and Y. Wang, "Gravitational Couplings of the Inflaton in Extended Inflation", *Physical Review Letters*, Vol. 65, No. 1, p. 17, 1990.
- 35. Futamase, T. and K.-i. Maeda, "Chaotic Inflationary Scenario of the Universe with a Nonminimally Coupled "Inflaton" Field", *Physical Review D*, Vol. 39, No. 2, p. 399, 1989.
- Salopek, D., J. Bond and J. M. Bardeen, "Designing Density Fluctuation Spectra in Inflation", *Physical Review D*, Vol. 40, No. 6, p. 1753, 1989.
- Fakir, R., S. Habib and W. Unruh, "Cosmological Density Perturbations with Modified Gravity", *The Astrophysical Journal*, Vol. 394, pp. 396–400, 1992.

- Fakir, R. and W. G. Unruh, "Improvement on Cosmological Chaotic Inflation Through Nonminimal Coupling", *Physical Review D*, Vol. 41, No. 6, p. 1783, 1990.
- Makino, N. and M. Sasaki, "The Density Perturbation in the Chaotic Inflation with Non-Minimal Coupling", *Progress of Theoretical Physics*, Vol. 86, No. 1, pp. 103–118, 1991.
- Kaiser, D. I., "Primordial Spectral Indices from Generalized Einstein Theories", *Physical Review D*, Vol. 52, No. 8, p. 4295, 1995.
- Mukaigawa, S., T. Muta and S. D. Odintsov, "Finite Grand Unified Theories and Inflation", *International Journal of Modern Physics A*, Vol. 13, No. 16, pp. 2739–2745, 1998.
- 42. Komatsu, E. and T. Futamase, "Complete Constraints on A Nonminimally Coupled Chaotic Inflationary Scenario from the Cosmic Microwave Background", *Physical Review D*, Vol. 59, No. 6, p. 064029, 1999.
- 43. Linde, A., M. Noorbala and A. Westphal, "Observational Consequences of Chaotic Inflation with Nonminimal Coupling to Gravity", *Journal of Cosmology and Astroparticle Physics*, Vol. 2011, No. 03, p. 013, 2011.
- Vennin, V., K. Koyama and D. Wands, "Encyclopædia Curvatonis", Journal of Cosmology and Astroparticle Physics, Vol. 2015, No. 11, p. 008, 2015.
- 45. Kaiser, D. I. and A. T. Todhunter, "Primordial Perturbations from Multifield Inflation with Nonminimal Couplings", *Physical Review D*, Vol. 81, No. 12, p. 124037, 2010.
- 46. Kaiser, D. I., E. A. Mazenc and E. I. Sfakianakis, "Primordial Bispectrum from Multifield Inflation with Nonminimal Couplings", *Physical Review D*, Vol. 87, No. 6, p. 064004, 2013.

- Greenwood, R. N., D. I. Kaiser and E. I. Sfakianakis, "Multifield Dynamics of Higgs Inflation", *Physical Review D*, Vol. 87, No. 6, p. 064021, 2013.
- Kaiser, D. I. and E. I. Sfakianakis, "Multifield Inflation After Planck: The Case for Nonminimal Couplings", *Physical review letters*, Vol. 112, No. 1, p. 011302, 2014.
- Schutz, K., E. I. Sfakianakis and D. I. Kaiser, "Multifield Inflation After Planck: Isocurvature Modes from Nonminimal Couplings", *Physical Review D*, Vol. 89, No. 6, p. 064044, 2014.
- DeCross, M. P., D. I. Kaiser, A. Prabhu, C. Prescod-Weinstein and E. I. Sfakianakis, "Preheating After Multifield Inflation with Nonminimal Couplings. I. Covariant Formalism and Attractor Behavior", *Physical Review D*, Vol. 97, No. 2, p. 023526, 2018.
- 51. Aghanim, N., Y. Akrami, M. Ashdown, J. Aumont, C. Baccigalupi, M. Ballardini, A. J. Banday, R. B. Barreiro, N. Bartolo, S. Basak, R. Battye, K. Benabed, J.-P. Bernard, M. Bersanelli, P. Bielewicz, J. J. Bock, J. R. Bond, J. Borrill, F. R. Bouchet, F. Boulanger, M. Bucher, C. Burigana, R. C. Butler, E. Calabrese, J.-F. Cardoso, J. Carron, A. Challinor, H. C. Chiang, J. Chluba, L. P. L. Colombo, C. Combet, D. Contreras, B. P. Crill, F. Cuttaia, P. de Bernardis, G. de Zotti, J. Delabrouille, J.-M. Delouis, E. D. Valentino, J. M. Diego, O. Doré, M. Douspis, A. Ducout, X. Dupac, S. Dusini, G. Efstathiou, F. Elsner, T. A. Enßlin, H. K. Eriksen, Y. Fantaye, M. Farhang, J. Fergusson, R. Fernandez-Cobos, F. Finelli, F. Forastieri, M. Frailis, A. A. Fraisse, E. Franceschi, A. Frolov, S. Galeotta, S. Galli, K. Ganga, R. T. Génova-Santos, M. Gerbino, T. Ghosh, J. González-Nuevo, K. M. Górski, S. Gratton, A. Gruppuso, J. E. Gudmundsson, J. Hamann, W. Handley, F. K. Hansen, D. Herranz, S. R. Hildebrandt, E. Hivon, Z. Huang, A. H. Jaffe, W. C. Jones, A. Karakci, E. Keihänen, R. Keskitalo, K. Kiiveri, J. Kim, T. S. Kisner, L. Knox, N. Krachmalnicoff, M. Kunz, H. Kurki-Suonio, G. Lagache, J.-M. Lamarre, A. Lasenby, M. Lattanzi, C. R. Lawrence, M. L.

Jeune, P. Lemos, J. Lesgourgues, F. Levrier, A. Lewis, M. Liguori, P. B. Lilje, M. Lilley, V. Lindholm, M. López-Caniego, P. M. Lubin, Y.-Z. Ma, J. F. Macías-Pérez, G. Maggio, D. Maino, N. Mandolesi, A. Mangilli, A. Marcos-Caballero, M. Maris, P. G. Martin, M. Martinelli, E. Martínez-González, S. Matarrese, N. Mauri, J. D. McEwen, P. R. Meinhold, A. Melchiorri, A. Mennella, M. Migliaccio, M. Millea, S. Mitra, M.-A. Miville-Deschênes, D. Molinari, L. Montier, G. Morgante, A. Moss, P. Natoli, H. U. Nørgaard-Nielsen, L. Pagano, D. Paoletti, B. Partridge, G. Patanchon, H. V. Peiris, F. Perrotta, V. Pettorino, F. Piacentini, L. Polastri, G. Polenta, J.-L. Puget, J. P. Rachen, M. Reinecke, M. Remazeilles, A. Renzi, G. Rocha, C. Rosset, G. Roudier, J. A. Rubiño-Martín, B. Ruiz-Granados, L. Salvati, M. Sandri, M. Savelainen, D. Scott, E. P. S. Shellard, C. Sirignano, G. Sirri, L. D. Spencer, R. Sunyaev, A.-S. Suur-Uski, J. A. Tauber, D. Tavagnacco, M. Tenti, L. Toffolatti, M. Tomasi, T. Trombetti, L. Valenziano, J. Valiviita, B. V. Tent, L. Vibert, P. Vielva, F. Villa, N. Vittorio, B. D. Wandelt, I. K. Wehus, M. White, S. D. M. White, A. Zacchei and A. Zonca, "Planck 2018 results-VI", Astronomy & Astrophysics, Vol. 641, p. A6, 2020.

- 52. Linde, A., Particle Physics and Inflationary Cosmology, Vol. 5, CRC press, 1990.
- 53. Barbara, R., Introduction to Cosmology, Addison Wesley, 2003.
- 54. Weinberg, S., Cosmology, OUP Oxford, 2008.
- Lyth, D. H. and A. R. Liddle, *The Primordial Density Perturbation: Cosmology*, Inflation and the Origin of Structure, Cambridge University Press, 2009.
- Carroll, S. M., "An Introduction to General Relativity: Spacetime and Geometry", Addison Wesley, Vol. 101, p. 102, 2004.
- Garrett, K. and G. Duda, "Dark Matter: A primer", Advances in Astronomy, Vol. 2011, p. 968283, 2011.

- Milgrom, M., "A Modification of the Newtonian Dynamics as a Possible Alternative to the Hidden Mass Hypothesis", *The Astrophysical Journal*, Vol. 270, pp. 365–370, 1983.
- Skordis, C. and T. Złośnik, "New Relativistic Theory for Modified Newtonian Dynamics", *Physical review letters*, Vol. 127, No. 16, p. 161302, 2021.
- 60. Chae, K.-H., F. Lelli, H. Desmond, S. S. McGaugh, P. Li and J. M. Schombert, "Testing the Strong Equivalence Principle: Detection of the External Field Effect in Rotationally Supported Galaxies", *The Astrophysical Journal*, Vol. 904, No. 1, p. 51, 2020.
- Georgi, H. and S. L. Glashow, "Unity of All Elementary-Particle Forces", *Physical Review Letters*, Vol. 32, No. 8, p. 438, 1974.
- Kibble, T. W., "Topology of Cosmic Domains and Strings", Journal of Physics A: Mathematical and General, Vol. 9, No. 8, p. 1387, 1976.
- Kolb, E. W. and M. Turner, *The Early Universe (Frontiers in Physics)*, Westview Press Incorporated, 1994.
- Bunkov, Y. M. and H. Godfrin, Topological Defects and the Non-Equilibrium Dynamics of Symmetry Breaking Phase Transitions, Vol. 549, Springer Science & Business Media, 2000.
- Vachaspati, T., Kinks and Domain Walls: An Introduction to Classical and Quantum Solitons, Cambridge University Press, 2006.
- Battye, R. A., M. Bucher and D. Spergel, "Domain Wall Dominated Universes", arXiv preprint astro-ph/9908047, 1999.
- Conversi, L., A. Melchiorri, L. Mersini and J. Silk, "Are Domain Walls Ruled Out?", Astroparticle Physics, Vol. 21, No. 4, pp. 443–449, 2004.

- Friedland, A., H. Murayama and M. Perelstein, "Domain Walls as Dark Energy", *Physical Review D*, Vol. 67, No. 4, p. 043519, 2003.
- del Campo, S., R. Herrera and D. Pavón, "Late Universe Expansion Dominated by Domain Walls and Dissipative Dark Matter", *Physical Review D*, Vol. 70, No. 4, p. 043540, 2004.
- 70. Avelino, P. and L. Sousa, "Domain Wall Network Evolution in (N+ 1)-Dimensional FRW Universes", *Physical Review D*, Vol. 83, No. 4, p. 043530, 2011.
- Kirillov, A. A. and B. S. Murygin, "Domain Walls and Strings Formation in the Early Universe", arXiv preprint arXiv:2011.07041, 2020.
- Bassett, B. A., S. Tsujikawa and D. Wands, "Inflation Dynamics and Reheating", *Reviews of Modern Physics*, Vol. 78, No. 2, p. 537, 2006.
- Baumann, D., "TASI Lectures on Inflation", arXiv preprint arXiv:0907.5424, 2009.
- Martin, J., C. Ringeval and V. Vennin, "Encyclopædia Inflationaris", *Physics of the Dark Universe*, Vol. 5, pp. 75–235, 2014.
- Lyth, D. H. and A. Riotto, "Particle Physics Models of Inflation and the Cosmological Density Perturbation", *Physics Reports*, Vol. 314, No. 1-2, pp. 1–146, 1999.
- 76. Linder, C. C., Particle Physics and Inflationary Cosmology, CRC press, 1990.
- Fujii, Y. and K.-i. Maeda, *The Scalar-Tensor Theory of Gravitation*, Cambridge University Press, 2003.
- 78. Faraoni, V., Cosmology in Scalar Tensor Gravity, Springer Science & Business

Media, 2004.

- Asselmeyer-Maluga, T., At the Frontier of Spacetime: Scalar-Tensor Theory, Bells Inequality, Machs Principle, Exotic Smoothness, Vol. 183, Springer, 2016.
- Bailin, D. and A. Love, Cosmology in Gauge Field Theory and String Theory, Taylor & Francis, 2017.
- Wainwright, J. and G. F. R. Ellis, *Dynamical Systems in Cosmology*, Cambridge University Press, 1997.
- Coley, A. A., Dynamical Systems and Cosmology, Vol. 291, Springer Science & Business Media, 2003.
- Böhmer, C. G. and N. Chan, "Dynamical Systems in Cosmology", Dynamical and Complex Systems, pp. 121–156, World Scientific, 2017.
- Bahamonde, S., C. G. Böhmer, S. Carloni, E. J. Copeland, W. Fang and N. Tamanini, "Dynamical Systems Applied to Cosmology: Dark Energy and Modified Gravity", *Physics Reports*, Vol. 775, pp. 1–122, 2018.
- Steinhardt, P. J. and M. S. Turner, "Prescription for Successful New Inflation", *Physical Review D*, Vol. 29, No. 10, p. 2162, 1984.
- Liddle, A. R. and D. H. Lyth, "COBE, Gravitational Waves, Inflation and Extended Inflation", *Physics Letters B*, Vol. 291, No. 4, pp. 391–398, 1992.
- Kruger, A. T. and J. W. Norbury, "Another Exact Inflationary Solution", *Physical Review D*, Vol. 61, No. 8, p. 087303, 2000.
- Fomin, I. and S. Chervon, "Exact and Approximate Solutions in the Friedmann Cosmology", *Russian Physics Journal*, Vol. 60, No. 3, pp. 427–440, 2017.

- Beesham, A., S. Chervon, S. Maharaj and A. Kubasov, "Exact Inflationary Solutions Inspired by the Emergent Universe Scenario", *International Journal of Theoretical Physics*, Vol. 54, No. 3, pp. 884–895, 2015.
- Makarenko, A. N. and V. V. Obukhov, "Exact Solutions in Modified Gravity Models", *Entropy*, Vol. 14, No. 7, pp. 1140–1153, 2012.
- Pintus, N. and S. Mignemi, "Mathematical Aspects of an Exactly Solvable Inflationary Model", *Journal of Physics: Conference Series*, Vol. 956, p. 012022, IOP Publishing, 2018.
- Rasouli, S. and P. V. Moniz, "Exact Cosmological Solutions in Modified Brans– Dicke Theory", *Classical and Quantum Gravity*, Vol. 33, No. 3, p. 035006, 2016.
- 93. Belinchon, J., T. Harko and M. Mak, "Exact Scalar-Tensor Cosmological Solutions via Noether Symmetry", Astrophysics and Space Science, Vol. 361, No. 2, p. 52, 2016.
- 94. Massaeli, E., M. Motaharfar and H. R. Sepangi, "General Scalar–Tensor Cosmology: Analytical Solutions via Noether Symmetry", *The European Physical Journal C*, Vol. 77, No. 2, p. 124, 2017.
- Ellis, G. and M. Madsen, "Exact Scalar Field Cosmologies", *Classical and Quantum Gravity*, Vol. 8, No. 4, p. 667, 1991.
- Morganstern, R., "Exact Solutions to Radiation-Filled Brans-Dicke Cosmologies", *Physical Review D*, Vol. 4, No. 2, p. 282, 1971.
- 97. Morganstern, R., "Exact Solutions to Brans-Dicke Cosmologies in Flat Friedmann Universes", *Physical Review D*, Vol. 4, No. 4, p. 946, 1971.
- Chauvet, P. and E. Guzmán, "Exact Solutions in Jordan-Brans-Dicke Homogeneous Universes I", Astrophysics and space science, Vol. 126, No. 1, pp. 133–141,

1986.

- Capozziello, S., R. d. Ritis, C. Rubano and P. Scudellaro, "Exact Solutions in Brans-Dicke Matter Cosmologies", *International Journal of Modern Physics D*, Vol. 5, No. 01, pp. 85–98, 1996.
- 100. Paliathanasis, A., "Conservation Laws and Exact Solutions in Brans–Dicke Cosmology with a Scalar Field", *General Relativity and Gravitation*, Vol. 51, No. 8, pp. 1–20, 2019.
- 101. Mukherjee, P. and S. Chakrabarti, "Exact Solutions and Accelerating Universe in Modified Brans–Dicke Theories", *The European Physical Journal C*, Vol. 79, No. 8, pp. 1–14, 2019.
- Neumann, J. V., Mathematical Foundations of Quantum Mechanics, Princeton university press, 1955.
- 103. Vourdas, A., "Quantum Systems with Finite Hilbert Space", Reports on Progress in Physics, Vol. 67, No. 3, p. 267, 2004.
- 104. Schwinger, J., "Unitary Operator Bases", Proceedings of the national academy of sciences of the United States Of America, Vol. 46, No. 4, p. 570, 1960.
- 105. Durt, T., "About Mutually Unbiased Bases in Even and Odd Prime Power Dimensions", Journal of Physics A: Mathematical and General, Vol. 38, No. 23, p. 5267, 2005.
- 106. Spengler, C., M. Huber, S. Brierley, T. Adaktylos and B. C. Hiesmayr, "Entanglement Detection via Mutually Unbiased Bases", *Physical Review A*, Vol. 86, No. 2, p. 022311, 2012.
- 107. Durt, T., B.-G. Englert, I. Bengtsson and K. Życzkowski, "On Mutually Unbiased Bases", International journal of quantum information, Vol. 8, No. 04, pp. 535–

- 108. Wootters, W. K. and B. D. Fields, "Optimal State-Determination by Mutually Unbiased Measurements", Annals of Physics, Vol. 191, No. 2, pp. 363–381, 1989.
- 109. Miquel, C., J. P. Paz and M. Saraceno, "Quantum Computers in Phase Space", *Physical Review A*, Vol. 65, No. 6, p. 062309, 2002.
- 110. Wootters, W. K., "Picturing Qubits in Phase Space", IBM Journal of Research and Development, Vol. 48, No. 1, pp. 99–110, 2004.
- 111. Paz, J. P., "Discrete Wigner Functions and the Phase-Space Representation of Quantum Teleportation", *Physical Review A*, Vol. 65, No. 6, p. 062311, 2002.
- 112. Davies, E. B. and J. T. Lewis, "An Operational Approach to Quantum Probability", *Communications in Mathematical Physics*, Vol. 17, No. 3, pp. 239–260, 1970.
- 113. Vourdas, A., "Galois Quantum Systems", Journal of Physics A: Mathematical and General, Vol. 38, No. 39, p. 8453, 2005.
- 114. Vourdas, A., "Quantum Systems with Finite Hilbert Space: Galois Fields in Quantum Mechanics", Journal of Physics A: Mathematical and Theoretical, Vol. 40, No. 33, p. R285, 2007.
- 115. Scully, M. O., B.-G. Englert and H. Walther, "Quantum Optical Tests of Complementarity", *Nature*, Vol. 351, No. 6322, p. 111, 1991.
- 116. Vourdas, A., "Quantum Systems with Finite Hilbert Space", Reports on Progress in Physics, Vol. 67, No. 3, p. 267, 2004.
- Rainville, E. D., P. E. Bedient and R. Bedient, *Elementary Differential Equations*, 7th, Maxwell Macmillan international Editions, Singapore, 1989.

- 118. Chimento, L. P. and A. S. Jakubi, "Scalar Field Cosmologies with Perfect Fluid in Robertson-Walker Metric", *International Journal of Modern Physics D*, Vol. 5, No. 01, pp. 71–84, 1996.
- 119. Korotkov, N. E. and A. N. Korotkov, Integrals Related to the Error Function, Chapman and Hall/CRC, 2020.
- 120. Matos, T., F. S. Guzmán and L. A. Urena-López, "Scalar Field as Dark Matter in the Universe", *Classical and Quantum Gravity*, Vol. 17, No. 7, p. 1707, 2000.
- 121. Scolnic, D. M., D. O. Jones, A. Rest, Y. C. Pan, R. Chornock, R. J. Foley, M. E. Huber, R. Kessler, G. Narayan, A. G. Riess, S. Rodney, E. Berger, D. J. Brout, P. J. Challis, M. Drout, D. Finkbeiner, R. Lunnan, R. P. Kirshner, N. E. Sanders, E. Schlafly, S. Smartt, C. W. Stubbs, J. Tonry, W. M. Wood-Vasey, M. Foley, J. Hand, E. Johnson, W. S. Burgett, K. C. Chambers, P. W. Draper, K. W. Ho-dapp, N. Kaiser, R. P. Kudritzki, E. A. Magnier, N. Metcalfe, F. Bresolin, E. Gall, R. Kotak, M. McCrum and K. W. Smith, "The Complete Light-curve Sample of Spectroscopically Confirmed SNe Ia from Pan-STARRS1 and Cosmological Constraints from the Combined Pantheon Sample", *The Astrophysical Journal*, Vol. 859, No. 2, p. 101, 2018.
- 122. Hobson, M. P., G. P. Efstathiou and A. N. Lasenby, General Relativity: An Introduction for Physicists, Cambridge University Press, 2006.
- 123. Riess, A. G., S. Casertano, W. Yuan, L. M. Macri and D. Scolnic, "Large Magellanic Cloud Cepheid Standards Provide a 1% Foundation for the Determination of the Hubble Constant and Stronger Evidence for Physics Beyond ΛCDM", *The Astrophysical Journal*, Vol. 876, No. 1, p. 85, 2019.
- 124. Efstathiou, G., "To H0 or Not to H0?" arXiv e-prints", arXiv preprint arXiv:2103.08723, Vol. 2103, 2021.

- 125. Camarena, D. and V. Marra, "On the Use of the Local Prior on the Absolute Magnitude of Type Ia Supernovae in Cosmological Inference", *Monthly Notices* of the Royal Astronomical Society, Vol. 504, No. 4, pp. 5164–5171, 2021.
- 126. Nunes, R. C. and E. Di Valentino, "Dark Sector Interaction and the Supernova Absolute Magnitude Tension", *Physical Review D*, Vol. 104, No. 6, p. 063529, 2021.
- 127. Colaço, L., R. Holanda and R. C. Nunes, "Varying-α in Scalar-Tensor Theory: Implications in Light of the Supernova Absolute Magnitude Tension and Forecast from GW Standard Sirens", arXiv preprint arXiv:2201.04073, 2022.
- 128. Camarena, D. and V. Marra, "A New Method to Build the (Inverse) Distance Ladder", Monthly Notices of the Royal Astronomical Society, Vol. 495, No. 3, pp. 2630–2644, 2020.
- 129. Ildes, M. and M. Arik, "Analytic Solutions of Scalar Field Cosmology, Mathematical Structures for Early Inflation and Late Time Accelerated Expansion", arXiv preprint arXiv:2203.16449, 2022.
- 130. Zhang, X., J. Yu, T. Liu, W. Zhao and A. Wang, "Testing Brans-Dicke Gravity Using the Einstein Telescope", *Physical Review D*, Vol. 95, No. 12, p. 124008, 2017.
- 131. Alsing, J., E. Berti, C. M. Will and H. Zaglauer, "Gravitational Radiation From Compact Binary Systems in the Massive Brans-Dicke Theory of Gravity", *Physical Review D*, Vol. 85, No. 6, p. 064041, 2012.
- Jordan, P., "Zur Empirischen Kosmologie", Naturwissenschaften, Vol. 26, No. 26, pp. 417–421, 1938.
- 133. Jordan, P., "Erweiterung der Projektiven Relativitätstheorie", Annalen der Physik, Vol. 436, No. 4-5, pp. 219–228, 1947.

- 134. Jordan, P., "Zum Gegenwärtigen Stand der Diracschen Kosmologischen Hypothesen", Zeitschrift für Physik, Vol. 157, No. 1, pp. 112–121, 1959.
- 135. Thiry, Y., "* Geometrie-les Equations de la Theorie Unitaire de KALUZA", Comptes Rendus Hebdomadaires des Seances de l Academie des Sciences, Vol. 226, No. 3, pp. 216–218, 1948.
- 136. Arik, M. and M. Calik, "Primordial and Late-Time Inflation in Brans-Dicke Cosmology", Journal of Cosmology and Astroparticle Physics, Vol. 2005, No. 01, p. 013, 2005.
- 137. Arik, M. and M. Calik, "Can Brans–Dicke Scalar Field Account for Dark Energy and Dark Matter?", *Modern Physics Letters A*, Vol. 21, No. 15, pp. 1241–1248, 2006.
- Arik, M., M. Calik and M. Sheftel, "Friedmann Equation for Brans–Dicke Cosmology", International Journal of Modern Physics D, Vol. 17, No. 02, pp. 225–235, 2008.
- Milgrom, M., "A Modification of the Newtonian Dynamics-Implications for Galaxies", The Astrophysical Journal, Vol. 270, pp. 371–383, 1983.
- 140. Ildes, M. and M. Arik, "Analytic Solutions of Brans-Dicke Cosmology: Early Inflation and Late Time Accelerated Expansion", arXiv preprint arXiv:2204.10396, 2022.
- 141. Sen, S. and T. Seshadri, "Self Interacting Brans–Dicke Cosmology and Quintessence", International Journal of Modern Physics D, Vol. 12, No. 03, pp. 445–460, 2003.
- 142. Arık, M. and M. B. Sheftel, "Symmetry Analysis and Exact Solutions of Modified Brans-Dicke Cosmological Equations", *Physical Review D*, Vol. 78, No. 6, p. 064067, 2008.
- 143. Mak, M. and T. Harko, "Brans-Dicke Cosmology with a Scalar Field Potential", *Europhysics Letters*, Vol. 60, No. 1, p. 155, 2002.
- 144. Chubaryan, E., R. Avakyan, G. Harutyunyan and A. Piloyan, "Role of Scalar Fields in Cosmological Models", *Advances in Space Research*, Vol. 44, No. 11, pp. 1359–1365, 2009.
- 145. Tahmasebzadeh, B. and K. Karami, "Generalized Brans–Dicke Inflation with a Quartic Potential", Nuclear Physics B, Vol. 918, pp. 1–10, 2017.
- 146. Santos, C. and R. Gregory, "Cosmology in Brans–Dicke Theory with a Scalar Potential", Annals of Physics, Vol. 258, No. 1, pp. 111–134, 1997.
- 147. L'DOVICH, Y. B. Z., "The Equation of State at Ultrahigh Densities and Its Relativistic Limitations", Journal of Experimental and Theoretical Physics, Vol. 14, No. 5, 1962.
- 148. Oliveira-Neto, G., G. Monerat, E. Corrêa Silva, C. Neves and L. Ferreira Filho, "An Early Universe Model with Stiff Matter and a Cosmological Constant", *International Journal of Modern Physics: Conference Series*, Vol. 3, pp. 254–265, World Scientific, 2011.
- 149. Dutta, S. and R. J. Scherrer, "Big Bang Nucleosynthesis with a Stiff Fluid", *Physical Review D*, Vol. 82, No. 8, p. 083501, 2010.
- 150. Nair, K. R. and T. K. Mathew, "Bulk Viscous Zel'dovich Fluid Model and Its Asymptotic Behavior", *The European Physical Journal C*, Vol. 76, No. 10, p. 519, 2016.
- Chavanis, P.-H., "Cosmology with a Stiff Matter Era", *Physical Review D*, Vol. 92, No. 10, p. 103004, 2015.
- 152. Barrow, J. D., "Quiescent Cosmology", Nature, Vol. 272, No. 5650, p. 211, 1978.

- 153. Joyce, M., "Electroweak Baryogenesis and the Expansion Rate of the Universe", *Physical Review D*, Vol. 55, No. 4, p. 1875, 1997.
- 154. Joyce, M. and T. Prokopec, "Turning Around the Sphaleron Bound: Electroweak Baryogenesis in an Alternative Post-Inflationary Cosmology", *Physical Review D*, Vol. 57, No. 10, p. 6022, 1998.
- 155. Colistete Jr, R., J. Fabris and N. Pinto-Neto, "Gaussian Superpositions in Scalar-Tensor Quantum Cosmological Models", *Physical Review D*, Vol. 62, No. 8, p. 083507, 2000.
- 156. Mathew, T. K., M. Aswathy and M. Manoj, "Cosmology and Thermodynamics of FLRW Universe with Bulk Viscous Stiff Fluid", *The European Physical Journal* C, Vol. 74, No. 12, p. 3188, 2014.
- 157. Barrow, J. D., A. Burd and D. Lancaster, "Three-Dimensional Classical Spacetimes", *Classical and Quantum Gravity*, Vol. 3, No. 4, p. 551, 1986.
- 158. Kamionkowski, M. and M. S. Turner, "Thermal Relics: Do We Know Their Abundances?", *Physical Review D*, Vol. 42, No. 10, p. 3310, 1990.
- Lorenz-Petzold, D., "Exact Perfect Fluid Solutions in the Brans-Dicke-Theory", Astrophysics and space science, Vol. 98, No. 2, pp. 249–254, 1984.
- 160. Li, B., T. Rindler-Daller and P. R. Shapiro, "Cosmological Constraints on Bose-Einstein-Condensed Scalar Field Dark Matter", *Physical Review D*, Vol. 89, No. 8, p. 083536, 2014.
- Casimir, H., Proceedings of the Koninklijke Nederlandse Akademie van Wetenschappen, Vol. 51, p. 793, 1948.
- 162. Hawking, S. W., "Particle Creation by Black Holes", Communications in mathematical physics, Vol. 43, No. 3, pp. 199–220, 1975.

- 163. Fulling, S. A. and P. C. Davies, "Radiation from a Moving Mirror in Two Dimensional Space-Time: Conformal Anomaly", *Proceedings of the Royal Society of London*, Vol. 348, No. 1654, pp. 393–414, 1976.
- 164. Davies, P. C. and S. A. Fulling, "Radiation From Moving Mirrors and from Black Holes", *Proceedings of the Royal Society of London*, Vol. 356, No. 1685, pp. 237– 257, 1977.
- 165. Elizalde, E., J. E. Lidsey, S. Nojiri and S. D. Odintsov, "Born–Infeld Quantum Condensate as Dark Energy in the Universe", *Physics Letters B*, Vol. 574, No. 1-2, pp. 1–7, 2003.
- 166. Brandenberger, R., "Matter Bounce in Horava-Lifshitz Cosmology", Physical Review D, Vol. 80, No. 4, p. 043516, 2009.
- Gibbons, G., "Phantom Matter and the Cosmological Constant", arXiv preprint hep-th/0302199, 2003.
- 168. Nojiri, S. and S. D. Odintsov, "Quantum de Sitter Cosmology and Phantom Matter", *Physics Letters B*, Vol. 562, No. 3-4, pp. 147–152, 2003.
- 169. Hertog, T., G. T. Horowitz and K. Maeda, "Negative Energy in String Theory and Cosmic Censorship Violation", *Physical Review D*, Vol. 69, No. 10, p. 105001, 2004.
- 170. Ellis, J. R., N. E. Mavromatos, V. A. Mitsou and D. V. Nanopoulos, "Confronting Dark Energy Models with Astrophysical Data: Non-Equilibrium vs. Conventional Cosmologies", Astroparticle Physics, Vol. 27, No. 2-3, pp. 185–198, 2007.
- 171. Ildes, M., M. Arik and M. B. Sheftel, "Inflation and Linear Expansion in the Radiation Dominated Era in Jordan–Brans–Dicke Cosmology", *International Journal* of Modern Physics D, Vol. 28, No. 04, p. 1950066, 2019.

- 172. Arik, M., M. Ildes and M. B. Sheftel, "Inflation and Linear Expansion in the Radiation Dominated Era in Jordan-Brans-Dicke Cosmology", arXiv preprint arXiv:1801.03031, 2018.
- 173. Santhanam, T. and A. Tekumalla, "Quantum Mechanics in Finite Dimensions", *Foundations of Physics*, Vol. 6, No. 5, pp. 583–587, 1976.
- 174. Bonatsos, D., C. Daskaloyannis, D. Ellinas and A. Faessler, "Discretization of the Phase Space for a q-Deformed Harmonic Oscillator with q a Root of Unity", *Physics Letters B*, Vol. 331, No. 1, pp. 150–156, 1994.
- 175. Biedenharn, L., "The Quantum Group SUq (2) and a q-Analogue of the Boson Operators", Journal of Physics A: Mathematical and General, Vol. 22, No. 18, p. L873, 1989.
- 176. Macfarlane, A., "On q-Analogues of the Quantum Harmonic Oscillator and the Quantum Group SU (2) q", Journal of Physics A: Mathematical and general, Vol. 22, No. 21, p. 4581, 1989.
- 177. Arik, M. and D. Coon, "Hilbert Spaces of Analytic Functions and Generalized Coherent States", *Journal of Mathematical Physics*, Vol. 17, No. 4, pp. 524–527, 1976.
- 178. Chung, W., "q-Oscillators with q a Root of Unity", *Helvetica Physica Acta*, Vol. 70, pp. 367–371, 1997.
- 179. Schwinger, J. and B.-G. Englert, Quantum Mechanics: Symbolism of Atomic Measurements, Springer, 2001.
- 180. Viet, N. A. and K. C. Wali, "Noncommutative Geometry and a Discretized Version of Kaluza-Klein Theory with a Finite Field Content", *International Journal of Modern Physics A*, Vol. 11, No. 03, pp. 533–551, 1996.

- 181. Arik, M. and M. Ildes, "Quantum Mechanics in a Space with a Finite Number of Points", Progress of Theoretical and Experimental Physics, Vol. 2016, No. 4, 2016.
- 182. Arik, M. and M. Ildes, "Quantum Mechanics on Periodic and Non-Periodic Lattices and Almost Unitary Schwinger Operators", *Journal of Mathematical Physics*, Vol. 59, No. 5, p. 053506, 2018.
- 183. Vecchiato, A., "Variational Approach to Gravity Field Theories", From Newton to Einstein and Beyond, Undergraduate Lecture Notes in Physics, Vol. 9, 2017.
- 184. Plebanski, J. and A. Krasinski, An Introduction to General Relativity and Cosmology, Cambridge University Press, 2006.
- 185. Edwards, C. H. and D. E. Penney, Differential Equations and Boundary Value Problems: Computing and Modeling, Pearson Educación, 2000.
- Seaborn, J. B., Hypergeometric Functions and Their Applications, Vol. 8, Springer Science & Business Media, 2013.
- 187. Faraoni, V. and S. Capozziello, Beyond Einstein Gravity: A Survey of Gravitational Theories for Cosmology and Astrophysics, Springer, 2011.
- 188. Olmo, G. J., "Palatini Approach to Modified Gravity: f (R) Theories and Beyond", International Journal of Modern Physics D, Vol. 20, No. 04, pp. 413–462, 2011.
- 189. Capozziello, S., F. Darabi and D. Vernieri, "Equivalence Between Palatini and Metric Formalisms of f (R)-Gravity by Divergence-Free Current", *Modern Physics Letters A*, Vol. 26, No. 01, pp. 65–72, 2011.

APPENDIX A: APPENDIX FOR CHAPTER 2

A.1. Einstein Tensor

Since derivation of the field equations is independent of the sign of the metric we continue with the metric sign $\{+, -, -, -\}$. Thus

$$ds^{2} = dt^{2} - a^{2}(t)\left[\frac{dr^{2}}{1 - k\frac{r^{2}}{L^{2}}} + r^{2}(d\theta^{2} + \sin^{2}\theta d\varphi^{2})\right].$$
 (A.1)

First, we would like remind the relation between coordinate basis vectors $\frac{\partial}{\partial x^{\alpha}}$ and coordinate basis one-forms dx^{α}

$$<\frac{\partial}{\partial x^{\alpha}}, dx^{\beta}> = \delta^{\beta}_{\alpha}.$$
 (A.2)

On the other hand orthonormal basis 1-forms are defined as

$$e^a = e^a_{\ \mu} dx^\mu, \tag{A.3}$$

obey the similar rule

$$\langle \vec{e_a}, e^b \rangle = \delta^b_a,$$
 (A.4)

where $\vec{e}_{\alpha} = E^{\nu}_{\alpha} \frac{\partial}{\partial x^{\nu}}$. Thus we obtain

$$e^{0} = dt, \qquad e^{1} = \frac{iadr}{\sqrt{1 - \frac{kr^{2}}{L^{2}}}},$$
 (A.5)

$$e^2 = iard\theta, \qquad e^3 = iarSin\theta d\varphi.$$
 (A.6)

Then we apply Cartan's first structure equation with no torsion;

$$de^a + \omega^a_b \wedge e^b = 0. \tag{A.7}$$

The connection 1-form has following properties

$$\omega_j^i = \omega_{ij} = \omega^{ij}, \qquad \omega_{ij} = -\omega_{ji}. \tag{A.8}$$

By using these features non-zero connection 1-forms are found as

$$\omega_{01} = -\frac{\dot{a}}{a}e^{1}, \qquad \omega_{12} = \frac{i\sqrt{1 - kr^{2}/L^{2}}}{ar}e^{2}, \qquad (A.9)$$

$$\omega_{02} = -\frac{\dot{a}}{a}e^{2}, \qquad \omega_{13} = \frac{i\sqrt{1 - kr^{2}/L^{2}}}{ar}e^{3}, \qquad \omega_{03} = -\frac{\dot{a}}{a}e^{3}, \qquad \omega_{23} = \frac{i\cot\theta}{ar}e^{3}.$$

Now we can calculate curvature two-forms by applying Cartan's second structure equation;

$$\Omega^a_b = d\omega^a_b + \omega^a_c \wedge \omega^c_b. \tag{A.10}$$

Hence we obtain

$$\Omega_1^0 = -\frac{\ddot{a}}{a}e^0 \wedge e^1, \qquad \Omega_2^1 = -(\frac{k}{L^2a^2} + \frac{\dot{a}^2}{a^2})e^1 \wedge e^2, \tag{A.11}$$

$$\Omega_{2}^{0} = -\frac{a}{a}e^{0} \wedge e^{2}, \qquad \Omega_{3}^{1} = -(\frac{k}{L^{2}a^{2}} + \frac{a^{2}}{a^{2}})e^{1} \wedge e^{3},$$

$$\Omega_{3}^{0} = -\frac{\ddot{a}}{a}e^{0} \wedge e^{3}, \qquad \Omega_{3}^{2} = -(\frac{k}{L^{2}a^{2}} + \frac{\dot{a}^{2}}{a^{2}})e^{2} \wedge e^{3}.$$
(A.12)

The relation between the Riemann curvature tensor and curvature two-forms is given by

$$\Omega^a_b = \frac{1}{2} R^a_{bcd} e^c \wedge e^d = \sum_{c < d} R^a_{bcd} e^c \wedge e^d.$$
(A.13)

Hence related terms are

$$R_{101}^{0} = R_{202}^{0} = R_{303}^{0} = -\frac{\ddot{a}}{a},$$

$$R_{212}^{1} = R_{313}^{1} = R_{323}^{2} = -(\frac{k}{L^{2}a^{2}} + \frac{\dot{a}^{2}}{a^{2}}).$$
(A.14)

Symmetry properties of the Riemann tensor according to it's indices are

$$R^{a}_{bcd} = -R^{a}_{bdc} = -R^{b}_{acd}.$$
 (A.15)

Thus

$$R_{010}^{1} = R_{020}^{2} = R_{030}^{3} = -\frac{\ddot{a}}{a},$$

$$R_{121}^{2} = R_{131}^{3} = R_{232}^{3} = -(\frac{k}{L^{2}a^{2}} + \frac{\dot{a}^{2}}{a^{2}}).$$
(A.16)

The Ricci tensor is defined as

$$R_{ab} = R_{acb}^c. \tag{A.17}$$

Then for orthonormal basis we find

$$R_{00} = R_{000}^{0} + R_{010}^{1} + R_{020}^{2} + R_{030}^{3} = -3\frac{\ddot{a}}{a}, \qquad (A.18)$$

$$R_{11} = R_{101}^{0} + R_{111}^{1} + R_{121}^{2} + R_{131}^{3} = -\frac{\ddot{a}}{a} - 2(\frac{k}{L^{2}a^{2}} + \frac{\dot{a}^{2}}{a^{2}}),$$

$$R_{22} = R_{202}^{0} + R_{212}^{1} + R_{222}^{2} + R_{232}^{3} = -\frac{\ddot{a}}{a} - 2(\frac{k}{L^{2}a^{2}} + \frac{\dot{a}^{2}}{a^{2}}),$$

$$R_{33} = R_{303}^{0} + R_{313}^{1} + R_{323}^{2} + R_{333}^{3} = -\frac{\ddot{a}}{a} - 2(\frac{k}{L^{2}a^{2}} + \frac{\dot{a}^{2}}{a^{2}}), \qquad (A.19)$$

where $R_{000}^0 = R_{111}^1 = R_{222}^2 = R_{333}^3 = 0.$

To find the Ricci tensor in coordinate basis we use the following relation

$$R_{ab}e^a \otimes e^b = R_{\mu\nu}e^\mu \otimes e^\nu, \qquad (A.20)$$

where e^a are presented in (A.5-A.6). Therefore Ricci tensor in coordinate basis are

$$R_{00} = -3\frac{\ddot{a}}{a}, \qquad R_{ii} = -\left[\frac{\ddot{a}}{a} + 2\left(\frac{k}{L^2 a^2} + \frac{\dot{a}^2}{a^2}\right)\right]g_{ii}.$$
 (A.21)

Then the Ricci scalar is found as

$$R = R^{\mu}_{\mu} = g^{\mu\nu}R_{\mu\nu} = -6(\frac{k}{L^2a^2} + \frac{\dot{a}^2}{a^2} + \frac{\ddot{a}}{a}).$$
(A.22)

Einstein tensor is defined as

$$G_{\mu\nu} = R_{\mu\nu} - \frac{1}{2}g_{\mu\nu}R.$$
 (A.23)

Thus components of the Einstein tensor is found as

$$G_{00} = 3\left(\frac{\dot{a}^2}{a^2} + \frac{k}{L^2 a^2}\right),\tag{A.24}$$

$$G_{ii} = \left(\frac{k}{L^2 a^2} + \frac{\dot{a}^2}{a^2} + 2\frac{\ddot{a}}{a}\right)g_{ii}.$$
 (A.25)

A.2. Action Variation for Non-minimal Coupling

A.2.1. Action Variation with Respect to Metric

Since we flip the sign of the metric to $\{+, -, -, -\}$ we flip the sign in front of the related terms in the action. According to (A.34) and (A.35) the Riemann tensor does not change sign when we change sign of the metric. Thus the Ricci tensor $R_{\mu\nu} = R^{\lambda} \mu \lambda \nu$ remains the same. However the Ricci scalar $R = g^{\mu\nu}R_{\mu\nu}$ changes sign. Thus for the

case of minimal coupling action is given by

$$S = \int d^4x \sqrt{-g} \left[-\frac{1}{2\kappa} (R + 2\Lambda) + \frac{1}{2} g^{\mu\nu} \partial_\mu \phi \partial_\nu \phi - V(\phi) + \mathcal{L}_M \right],$$
(A.26)

where R is the Ricci scalar and $\kappa = 8\pi G$.

We would like to present some useful formulas before going further. If one starts from

$$g^{\mu\lambda}g_{\lambda\nu} = \delta^{\mu}_{\nu}, \tag{A.27}$$

one can obtain

$$\delta g_{\mu\nu} = -g_{\mu\alpha}g_{\lambda\nu}(\delta g^{\alpha\lambda}). \tag{A.28}$$

By using

$$ln(detM) = Tr(lnM), \tag{A.29}$$

$$\frac{1}{detM}\delta(detM) = Tr(M^{-1}\delta M), \tag{A.30}$$

with $M = g_{\mu\nu}$ one can show that

$$\delta g = -gg_{\mu\nu}\delta g^{\mu\nu}.\tag{A.31}$$

Here $g = det g_{\mu\nu} < 0$ so $\sqrt{-g} = \sqrt{|g|}$. Thus

$$\delta\sqrt{-g} = -\frac{1}{2}\sqrt{-g}g_{\mu\nu}\delta g^{\mu\nu}.$$
(A.32)

On the other hand we also have

$$\delta R = \delta(g^{\mu\nu}R_{\mu\nu}) = \delta(g^{\mu\nu})R_{\mu\nu} + g^{\mu\nu}\delta(R_{\mu\nu}). \tag{A.33}$$

To calculate $\delta R_{\mu\nu}$ we follow the Palatini approach. The Riemann tensor has been defined as

$$R^{\mu}_{\nu\sigma\rho} = \partial_{\sigma}\Gamma^{\mu}_{\nu\rho} - \partial_{\rho}\Gamma^{\mu}_{\sigma\nu} + \Gamma^{\mu}_{\sigma\lambda}\Gamma^{\lambda}_{\rho\nu} - \Gamma^{\mu}_{\lambda\rho}\Gamma^{\lambda}_{\sigma\nu}, \qquad (A.34)$$

where connection coefficients are given as

$$\Gamma^{\mu}_{\nu\sigma} = \frac{1}{2} g^{\mu\lambda} (\partial_{\nu} g_{\lambda\sigma} + \partial_{\sigma} g_{\nu\lambda} - \partial_{\lambda} g_{\sigma\nu}).$$
(A.35)

Thus we have

$$\delta R_{\mu\nu} = \delta R^{\lambda}_{\mu\lambda\nu} = \nabla_{\lambda} \delta \Gamma^{\lambda}_{\mu\nu} - \nabla_{\nu} \delta \Gamma^{\lambda}_{\mu\lambda}. \tag{A.36}$$

In this formalism connection coefficients and metric are accepted as two independent fields [183, 184]. Therefore we should vary the action with respect to inverse of the metric and connection coefficients. It is obvious that variation of $R_{\mu\nu}$ with respect to $g^{\mu\nu}$ is zero while variation of $R_{\mu\nu}$ with respect to connection coefficients imply (A.35).

One can write the action sum of the following parts

$$S_{EH\Lambda} = \int d^4x \sqrt{-g} \left[-\frac{1}{2\kappa} (R + 2\Lambda) \right], \tag{A.37}$$

$$S_{\phi} = \int d^4x \sqrt{-g} \left[\frac{1}{2}g^{\mu\nu}\partial_{\mu}\phi\partial_{\nu}\phi - V(\phi)\right], \qquad (A.38)$$

$$S_M = \int d^4x \sqrt{-g} \mathcal{L}_M. \tag{A.39}$$

Then

$$\delta S_{EH\Lambda} = \int d^4x \Big\{ \delta(\sqrt{-g}) \left[-\frac{1}{2\kappa} (R+2\Lambda) \right] - \sqrt{-g} \left(\frac{1}{\kappa} \delta R \right) \Big\}.$$
(A.40)

By using (A.32-A.36) we obtain

$$\delta S_{EH\Lambda} = \int d^4x \Big\{ \frac{1}{2} (\sqrt{-g}) g_{\mu\nu} \delta g^{\mu\nu} [\frac{1}{2\kappa} (R+2\Lambda)] - \sqrt{-g} [\frac{1}{\kappa} \delta(g^{\mu\nu}) R_{\mu\nu}] \Big\}.$$
(A.41)

Then variation with respect to $g^{\mu\nu}$ leads to

$$\frac{\delta S_{EH\Lambda}}{\sqrt{-g}\delta g^{\mu\nu}} = \frac{1}{2\kappa} \Big[-R_{\mu\nu} + \frac{1}{2}g_{\mu\nu}(R+2\Lambda) \Big] + \frac{1}{\sqrt{-g}}\frac{\delta S_M}{\delta g^{\mu\nu}} = 0.$$
(A.42)

The last term is zero and this will be shown later. We use the energy momentum tensor which is defined as

$$T_{\mu\nu} = \frac{2}{\sqrt{-g}} \frac{\delta S_M}{\delta g^{\mu\nu}},\tag{A.43}$$

where

$$T_{\mu\nu} = g_{\mu\lambda} T^{\lambda} \nu, \qquad T^{\mu}_{\nu} = diag\{\rho, -p, -p - p\}.$$
 (A.44)

Hence we obtain

$$R_{\mu\nu} - \frac{1}{2}g_{\mu\nu}R - \Lambda g_{\mu\nu} = 8\pi G_N T_{\mu\nu}, \qquad (A.45)$$

where we have used $\kappa = 8\pi G_N$. By using definition of the Einstein tensor which is given by (A.23) the last equation becomes

$$G_{\mu\nu} - \Lambda g_{\mu\nu} = 8\pi G_N T_{\mu\nu}.\tag{A.46}$$

$$\frac{\dot{a}^2}{a^2} = \frac{8\pi G}{3}\rho + \frac{\Lambda}{3} - \frac{k}{L^2 a^2},\tag{A.47}$$

$$\frac{\ddot{a}}{a} = -\frac{4\pi G}{3}(\rho + 3p) + \frac{\Lambda}{3}.$$
 (A.48)

Now we take care of the remaining term according to the action principle

$$\delta S_{\phi} = \int d^4x \Big\{ \delta(\sqrt{-g}) [\frac{1}{2} g^{\mu\nu} \partial_{\mu} \phi \partial_{\nu} \phi - V(\phi)] + \sqrt{-g} [\frac{1}{2} (\delta g^{\mu\nu}) \partial_{\mu} \phi \partial_{\nu} \phi] \Big\},$$
(A.49)
$$\delta S_{\phi} = \int d^4x \Big\{ -\frac{1}{2} \sqrt{-g} g_{\mu\nu} (\delta g^{\mu\nu}) [\frac{1}{2} g^{\alpha\beta} \partial_{\alpha} \phi \partial_{\beta} \phi - V(\phi)] + \sqrt{-g} [\frac{1}{2} (\delta g^{\mu\nu}) \partial_{\mu} \phi \partial_{\nu} \phi] \Big\},$$
(A.50)

By comparing the last equation with (A.57) we have

$$T^{\phi}_{\mu\nu} = 2 \frac{\delta S_{\phi}}{\sqrt{-g} \delta g^{\mu\nu}}, \qquad (A.51)$$
$$= \partial_{\mu} \phi \partial_{\nu} \phi - g_{\mu\nu} [\frac{1}{2} \partial_{\alpha} \phi \partial^{\alpha} \phi - V(\phi)].$$

By using perfect fluid energy momentum tensor for a scalar field

$$T_{\mu\nu} = (\rho + p)u_{\mu}u_{\nu} - pg_{\mu\nu}, \qquad (A.52)$$

we obtain

$$\rho_{\phi} = \frac{1}{2}\dot{\phi}^2 + V(\phi) + \frac{1}{2}(\nabla\phi)^2, \qquad (A.53)$$

$$p_{\phi} = \frac{1}{2}\dot{\phi}^2 - V(\phi) + \frac{1}{2}(\nabla\phi)^2.$$
 (A.54)

<u>A.2.1.1.</u> Action Principle. According to action principle S_M is invariant under coordinate transformations [54]. By adding infinitesimal vector field $\varepsilon^{\mu}(x)$ to our coordinates

we have

$$x^{\prime \mu} = x^{\mu} + \varepsilon^{\mu}(x). \tag{A.55}$$

One can show that [122]

$$\delta g_{\mu\nu} = -(\nabla_{\mu}\varepsilon_{\nu} + \nabla_{\nu}\varepsilon_{\mu}). \tag{A.56}$$

On the other hand by using definition given (A.43) one can write

$$\delta S_M \equiv \int_{\mathcal{R}} \frac{\delta \mathcal{L}_M}{\delta g^{\mu\nu}} \delta g^{\mu\nu} d^4 x = \frac{1}{2} \int_{\mathcal{R}} T_{\mu\nu} \delta g^{\mu\nu} \sqrt{-g} d^4 x \quad = -\frac{1}{2} \int_{\mathcal{R}} T^{\mu\nu} \delta g_{\mu\nu} \sqrt{-g} d^4 x, \quad (A.57)$$

where we have used $\delta g^{\mu\lambda}g_{\lambda\nu} = -g^{\mu\lambda}\delta g_{\lambda\nu}$. By using (A.56) and remembering $T^{\mu\nu}$ is symmetric we obtain

$$\delta S_M = \int_{\mathcal{R}} T^{\mu\nu} (\nabla_\mu \varepsilon_\nu) \sqrt{-g} d^4 x = 0.$$
 (A.58)

By applying integration by parts, one can write

$$\delta S_M = \int_{\mathcal{R}} \nabla_\mu (T^{\mu\nu} \varepsilon_\nu) \sqrt{-g} d^4 x - \int_{\mathcal{R}} \nabla_\mu (T^{\mu\nu}) \varepsilon_\nu \sqrt{-g} d^4 x = 0.$$
 (A.59)

By using divergence theorem

$$\delta S_M = \int_{\partial \mathcal{R}} n_\mu T^{\mu\nu} \varepsilon_\nu \sqrt{|\gamma|} d^3 x - \int_{\mathcal{R}} \nabla_\mu (T^{\mu\nu}) \varepsilon_\nu \sqrt{-g} d^4 x = 0, \qquad (A.60)$$

where γ is determinant of the induced metric and n_{μ} is the unit normal to the boundary. Since surface integral vanishes we end up with

$$\nabla_{\mu}(T^{\mu\nu}) = 0. \tag{A.61}$$

A.2.2. Action Variation with Respect to the Scalar Field

In general the action is written as

$$S = \int \mathcal{L}(\phi^a, \nabla_\mu \phi^a) \sqrt{-g} d^4 x.$$
 (A.62)

Under variation of the scalar field

$$\nabla_{\mu}\phi^{a} \to \nabla_{\mu}\phi^{a} + \nabla_{\mu}(\delta\phi^{a}), \qquad (A.63)$$

$$\nabla_{\mu}(\delta\phi^a) = \delta(\nabla_{\mu}\phi^a). \tag{A.64}$$

Then

$$\delta S = \int \delta \mathcal{L} \sqrt{-g} d^4 x, \qquad (A.65)$$

$$= \int \left[\frac{\partial \mathcal{L}}{\partial \phi^a} \delta \phi^a + \frac{\partial \mathcal{L}}{\partial (\nabla_\mu \phi^a)} \delta (\nabla_\mu \phi^a) \right] \sqrt{-g} d^4 x.$$
 (A.66)

By applying (A.64) and by applying integration by parts respectively, the second term on the right side becomes

$$\int \frac{\partial \mathcal{L}}{\partial (\nabla_{\mu} \phi^{a})} \nabla_{\mu} (\delta \phi^{a}) \sqrt{-g} d^{4}x = \int \nabla_{\mu} [\frac{\partial \mathcal{L}}{\partial (\nabla_{\mu} \phi^{a})} \delta \phi^{a}] \sqrt{-g} d^{4}x \qquad (A.67)$$
$$- \int \nabla_{\mu} [\frac{\partial \mathcal{L}}{\partial (\nabla_{\mu} \phi^{a})}] \delta \phi^{a} \sqrt{-g} d^{4}x,$$

where the first term vanishes at the boundaries. Thus we are left with

$$\delta S = \int \left\{ \frac{\partial \mathcal{L}}{\partial \phi^a} - \nabla_\mu [\frac{\partial \mathcal{L}}{\partial (\nabla_\mu \phi^a)}] \right\} \delta \phi^a \sqrt{-g} d^4 x, \tag{A.68}$$

$$= \int \frac{\delta \mathcal{L}}{\delta \phi^a} \delta \phi^a \sqrt{-g} d^4 x. \tag{A.69}$$

where

$$\frac{\delta \mathcal{L}}{\delta \phi^a} = \frac{\partial \mathcal{L}}{\partial \phi^a} - \nabla_{\mu} \left[\frac{\partial \mathcal{L}}{\partial (\nabla_{\mu} \phi^a)} \right] = 0.$$
(A.70)

Thus we have shown that variation of the action with respect to the scalar field results in Euler-Lagrange equations.

Now we have

$$S_{\phi} = \int d^4x \sqrt{-g} \left[\frac{1}{2}g^{\mu\nu}\partial_{\mu}\phi\partial_{\nu}\phi - V(\phi)\right].$$
(A.71)

Since for scalars, the covariant derivative equals to a partial derivative we can write

$$S_{\phi} = \int d^4x \sqrt{-g} \left[\frac{1}{2}g^{\mu\nu} \nabla_{\mu} \phi \nabla_{\nu} \phi - V(\phi)\right].$$
(A.72)

Then

$$\frac{\partial \mathcal{L}}{\partial \phi} = -\frac{dV(\phi)}{d\phi},\tag{A.73}$$

$$\frac{\partial \mathcal{L}}{\partial (\nabla_{\mu} \phi)} = \frac{\partial}{\partial (\nabla_{\mu} \phi)} [\frac{1}{2} g^{\rho \sigma} \nabla_{\rho} \phi \nabla_{\sigma} \phi] = g^{\mu \nu} \nabla_{\nu} \phi.$$
(A.74)

Hence the Euler-Lagrange equation becomes

$$\frac{dV(\phi)}{d\phi} + g^{\mu\nu}\nabla_{\nu}\phi = 0.$$
(A.75)

Since $\nabla_u g^{\mu\nu} = 0$, we can write

$$\Box \phi + \frac{dV(\phi)}{d\phi} = 0, \tag{A.76}$$

where $\Box \equiv g^{\mu\nu} \nabla_{\mu} \nabla_{\nu}$. Then by using the following property

$$\nabla_i \nabla_j \phi = \nabla_i \partial_j \phi, \tag{A.77}$$

$$= \partial_i \partial_j \phi - \partial_k \phi \Gamma^k_{ij},$$

one can show that

$$\Box \phi = \ddot{\phi} + 3\frac{\dot{a}}{a}\dot{\phi}.$$
 (A.78)

Finally we obtain our last equation as

$$\ddot{\phi} + 3\frac{\dot{a}}{a}\dot{\phi} + \frac{dV(\phi)}{d\phi} = 0.$$
(A.79)

A.3. Change of Independent Variable

$$H(a) = \frac{\dot{a}}{a},\tag{A.80}$$

therefore our new independent variable becomes a scale factor 'a'. For this reason we write all other variables in terms of the new variable as

$$\phi = \phi(a), \quad V(\phi(a)) = V(a), \quad \dot{a} = aH(a).$$
 (A.81)

As a result we obtain by change of variable

$$\frac{d\phi}{dt} = \frac{d\phi(a)}{dt}\frac{da}{dt} = \phi'\dot{a} = \phi'aH, \tag{A.82}$$

$$\dot{\phi} = \phi' a H, \tag{A.83}$$

$$\frac{d^2\phi}{dt^2} = \frac{d}{dt}(\phi' a H) = \frac{d}{da}(\phi' a H)\frac{da}{dt},$$
(A.84)

$$\ddot{\phi} = \phi'' a^2 H^2 + \phi' a H^2 + \phi' a^2 H H'$$
(A.85)

By the help of the chain rule

$$\frac{dV}{d\phi} = \frac{dV}{da}\frac{da}{d\phi} = \frac{dV}{da}\frac{1}{\frac{d\phi}{da}},\tag{A.86}$$

$$\frac{dV}{d\phi} = V' \frac{1}{\phi'}.$$
(A.87)

On the other hand by starting from our definition we get followings

$$\frac{\dot{a}}{a} = H(a),\tag{A.88}$$

$$\frac{\ddot{a}}{a} - \frac{\dot{a}^2}{a^2} = \frac{dH}{da}\frac{da}{dt},\tag{A.89}$$

$$\frac{\ddot{a}}{a} = H'aH + H^2. \tag{A.90}$$

A.4. Integration by Parts

By using functions u and v a theorem integration by parts is written as

$$\int_{x_0}^x u(x)dv(x) = u(x)v(x) \bigg|_{x_0}^x - \int_{x_0}^x v(x)du(x.)$$
(A.91)

When calculating the function $\gamma(a)$ we choose

$$u = -a^{'6}, \qquad dv = V^{'}da^{'},$$
 (A.92)

thus

$$du = -6a^{'5}da^{'}, \quad v = V(a^{'}),$$
 (A.93)

results in

$$\int_{a_{in}}^{a} (-a'^{6}V')da' = -a'^{6}V(a') \bigg|_{a_{in}}^{a} + 6 \int_{a_{in}}^{a} V(a')a'^{5}da',$$

$$\int_{a_{in}}^{a} (-a'^{6}V')da' = -a^{6}V(a) + 6 \int_{a_{in}}^{a} V(a')a'^{5}da' + a_{in}^{6}V(a_{in}).$$
(A.94)

A.5. The Linear First Order Differential Equation

The initial value problem

$$\frac{dy}{dx} + P(x)y = Q(x), \qquad y(x_0) = y_0,$$
 (A.95)

has the following solution [185]

$$y(x) = \frac{1}{\rho(x)} \Big[\int_{x_0}^x \rho(x') Q(x') dx' + y_0 \Big],$$
(A.96)
$$\rho(x) = exp \Big(\int_{x_0}^x P(x') dx' \Big).$$

A.6. Hypergeometric Functions

The function $(1-z)^s$ can be represented by Maclaurin series as

$$(1-z)^{s} = \sum_{k=0}^{\infty} \frac{[f^{(k)(z)}]_{z=0}}{k!} z^{k},$$

$$= \sum_{k=0}^{\infty} \frac{(-1)^{k} s(s-1) \dots (s-k+1)}{k!} z^{k}.$$
(A.97)

By using Pochhammer symbol $(b)_n$ which is defined as

$$(b)_n = \begin{cases} 1 & \text{if } n = 0, \\ b(b+1)(b+2)\dots(b+n-1), & \text{if } n = 1, 2, \dots \end{cases}$$
(A.98)

one can write

$$(1-z)^s = \sum_{k=0}^{\infty} \frac{(-s)_k}{k!} z^k, \quad (-s)_k = (-1)^k (s-k+1)_k.$$
 (A.99)

The hypergeometric function F(a, b; c; z) is defined as [186]

$$F(a,b;c;z) = \sum_{n=0}^{\infty} \frac{(a)_n (b)_n}{n! (c)_n} z^n.$$
 (A.100)

On the other hand the generalized Hypergeometric Function is defined as

$${}_{p}F_{q}(a_{1}, a_{2}, \dots, a_{p}; b_{1}, b_{2}, \dots, b_{q}; z) = \sum_{n=0}^{\infty} \frac{(a_{1})_{n}(a_{2})_{n} \dots (a_{p})_{n}}{n!(b_{1})_{n}(b_{2})_{n} \dots (b_{q})_{n}} z^{n}.$$
 (A.101)

This function is also denoted as ${}_{p}F_{q}(a;b;z)$. By comparing (A.100) and (A.101) one can write

$$_{2}F_{1}(a,b;c;z) = F(a,b;c;z).$$
 (A.102)

Thus we can express the function $(1-z)^s$ as

$$(1-z)^s = F(-s,b;b;z) = {}_1F_0(-s;-;z) = \sum_{k=0}^{\infty} \frac{(-s)_k}{k!} z^k.$$
 (A.103)

Furthermore, by using definition (A.101) one can show that

$$\frac{1}{a}(z\frac{d}{dz}+a)_2F_1(a,b;c;z) = \frac{1}{a}\sum_{n=0}^{\infty}\frac{(a+n)(a)_n(b)_n}{(c)_n n!}z^n,$$

$$= {}_2F_1(a+1,b;c;z).$$
(A.104)

where we have used $(a + n) = a(a + 1)_n$. When we choose c = a + 1 the last equation reduce to

$$\frac{1}{a}(z\frac{d}{dz}+a)_2F_1(a,b;c;z) = {}_1F_0(b;-;z), \qquad (A.105)$$
$$= (1-z)^{-b}.$$

We use the following trick

$$\frac{1}{a}(z\frac{d}{dz}+a)f(z) = \frac{1}{a}z^{1-a}\frac{d}{dz}[z^a f(z)],$$
(A.106)

which is valid for all functions. Thus

$$\frac{1}{a}z^{1-a}\frac{d}{dz}[z^a{}_2F_1(a,b;a+1;z)] = {}_1F_0(b;-;z),$$
(A.107)

where we have used (A.105). Then we obtain

$$\frac{d}{dz}[z^{a}{}_{2}F_{1}(a,b;a+1;z)] = az^{a-1}{}_{1}F_{0}(b;-;z), \qquad (A.108)$$
$$= az^{a-1}(1-z)^{-b}.$$

where we have used second line of (A.105).

Second trick is to replace a = k + 1 and b = -m. According to (A.100) we can swap the places of a and b, thus we obtain

$$\frac{d}{du}[u^{k+1}{}_2F_1(-m,k+1;k+2;u)] = (k+1)u^k(1-u)^m,$$
(A.109)

where we have changed our argument of the function form z to u since it will be useful later. Then by taking integral of both sides we find

$$\int (1-u)^m u^k du = \frac{u^{k+1}}{k+1} {}_2F_1(-m,k+1;k+2;u).$$
(A.110)

A.6.1. Some Integrals Expressed as Hypergeometric Function

We have used following technique to find relation between time and the scale factor at the integral given in Section 2.3.2.2 by the (2.114)

$$t = \int_{a_{in}}^{a} \frac{da'}{a'H(a')},$$

$$t = \int_{a_{in}}^{a} \frac{da'}{a'\sqrt{\frac{8\pi G}{3}(\frac{f}{a'^{6}} + \frac{6v_{n}}{(6-n)a'^{n}})}},$$
(A.111)

where $f = a_{in}^6 \tilde{\gamma}(a_{in})$. We continue our calculations as

$$t = \sqrt{\frac{3}{8\pi G f}} \int_{a_{in}}^{a} \frac{a^{'2}}{\sqrt{1 + \frac{6v_n a^{'(6-n)}}{(6-n)f}}} da^{'}, \tag{A.112}$$

$$t = \sqrt{\frac{3}{8\pi G f}} \int_{a_{in}}^{a} a^{\prime 2} [1 + \frac{6v_n a^{\prime (6-n)}}{(6-n)f}]^{-1/2} da^{\prime}.$$
(A.113)

Now we apply the following change of variable

$$u = -\frac{6v_n a^{\prime(6-n)}}{(6-n)f}.$$
(A.114)

Thus

$$a = \left[-\frac{(6-n)f}{6v_n}u\right]^{\frac{1}{6-n}}, \qquad da = \frac{1}{6-n}\left[-\frac{(6-n)f}{6v_n}\right]^{\frac{1}{6-n}}u^{\frac{1}{6-n}-1}.$$
 (A.115)

Then we have

$$t = \sqrt{\frac{3}{8\pi G f}} \frac{1}{(6-n)} \left[-\frac{(6-n)f}{6v_n} \right]^{\frac{3}{6-n}} \int_{a_{in}}^a u^{\frac{3}{6-n}-1} (1-u)^{-1/2} du.$$
(A.116)

By comparing the integral at the right side with the integral given in (A.110) we obtain

$$m = -\frac{1}{2}, \qquad k = \frac{3}{6-n} - 1.$$
 (A.117)

Hence

$$t = \sqrt{\frac{3}{8\pi G f}} \frac{1}{(6-n)} \left[-\frac{(6-n)f}{6v_n} \right]^{\frac{3}{6-n}} \left[\frac{u^{3/(6-n)}}{\frac{3}{6-n}} {}_2F_1(\frac{1}{2}, \frac{3}{6-n}; \frac{3}{6-n} + 1; u) \right]_{u(a_{in})}^{u(a)}.$$
(A.118)

By back substitution $u = -\frac{6v_n a^{'(6-n)}}{(6-n)f}$ we obtain

$$t = \sqrt{\frac{1}{24\pi G a_{in}^6 \tilde{\gamma}(a_{in})}} \left[a^{'3}{}_2F_1(\frac{1}{2}, \frac{3}{6-n}; \frac{3}{6-n} + 1; -\frac{6v_n}{(6-n)a_{in}^6 \tilde{\gamma}(a_{in})} a^{'(6-n)}) \right]_{a_{in}}^a.$$
(A.119)

We have used the same technique to find relation between time and the scale factor at the integral given in Section 2.4.2.2 by the (2.163)

$$t = \int_{a_{in}}^{a} \frac{da'}{a'H(a')},$$

$$t = \int_{a_{in}}^{a} \frac{da'}{a'\sqrt{\frac{8\pi G}{3}(\frac{\rho_w}{a'} + \frac{\rho_n}{a'n})}},$$

$$t = \sqrt{\frac{3}{8\pi G\rho_w}} \int_{a_{in}}^{a} a'^{-1/2}(1 + \frac{\rho_n a'^{(1-n)}}{\rho_w})^{-1/2} da'.$$
(A.120)

We choose the following change of variable

$$u = -\frac{\rho_n}{\rho_w} a^{\prime(1-n)}.$$
 (A.121)

Thus

$$a = \left(-\frac{\rho_w}{\rho_n}u\right)^{\frac{1}{1-n}}, \qquad da = \frac{1}{1-n}\left(-\frac{\rho_w}{\rho_n}\right)^{\frac{1}{1-n}}u^{\frac{1}{1-n}-1}du.$$
(A.122)

Then we have

$$t = \left(\frac{1}{1-n}\right)\sqrt{\frac{3}{8\pi G\rho_w}} \left(-\frac{\rho_w}{\rho_n}\right)^{\frac{1/2}{1-n}} \int_{a_{in}}^a u^{\frac{1}{2-2n}-1} (1-u)^{-1/2} du.$$
(A.123)

By comparing the last integral with the integral given in (A.110) yields

$$m = -\frac{1}{2}, \qquad k = \frac{1}{2-2n} - 1.$$
 (A.124)

Thus

$$t = \left(\frac{1}{1-n}\right)\sqrt{\frac{3}{8\pi G\rho_w}} \left(-\frac{\rho_w}{\rho_n}\right)^{\frac{1/2}{1-n}} \left[\frac{u^{\frac{1}{2-2n}}}{\frac{1}{2-2n}} {}_2F_1\left(\frac{1}{2}, \frac{1}{2-2n}; \frac{1}{2-2n}+1; u\right)\right]_{u(a_{in})}^{u(a)}.$$
 (A.125)

By back substitution $u = -\frac{\rho_n}{\rho_w} a^{\prime(1-n)}$ we obtain

$$t = \sqrt{\frac{3}{2\pi G\rho_w}} \left[\sqrt{a'_2} F_1(\frac{1}{2}, \frac{1}{2-2n}; \frac{1}{2-2n} + 1; -\frac{\rho_n}{\rho_w} a'^{(1-n)}) \right]_{a_{in}}^a.$$
 (A.126)

A.7. Integral Expressed as Erfi Function

We calculate the scale factor as a function of time in Section 2.3.2.2 for case n = 6by following steps. First, we use the H(a) given in (2.99) with k=0, so we have

$$t = \int_{a_{in}}^{a} \frac{da'}{a'H(a')},$$
 (A.127)

$$t = \int_{a_{in}}^{a} \frac{da'}{a'\sqrt{\frac{8\pi G}{3}(\frac{f}{a'^{6}} + \frac{gln(\frac{a'}{c})}{a'^{6}})}},$$
(A.128)

where $f = V_6 + a_{in}^6 \gamma(a_{in})$, $g = 6V_6$ and $c = a_{in}$. Then we have

$$t = \sqrt{\frac{3}{8\pi G}} \int_{a_{in}}^{a} \frac{a^{'2} da^{'}}{\sqrt{f + gln(\frac{a^{'}}{c})}}.$$
 (A.129)

We apply the change of variable as $u = \frac{a'}{c}$ and $du = \frac{da'}{c}$ so

$$I = \int \frac{a^{'2} da'}{\sqrt{f + gln(\frac{a'}{c})}},$$

$$= c^3 \int \frac{u^2}{\sqrt{f + gln(u)}} du.$$
(A.130)

We apply the second change of variable as v = f + g lnu and $dv = \frac{g}{u} du$ so

$$I = \frac{c^3 e^{-3f/g}}{g} \int \frac{e^{3v/g}}{\sqrt{v}} dv.$$
(A.131)

Now we choose $w = \sqrt{\frac{3v}{g}}$ so $dw = \frac{1}{2}\sqrt{\frac{3}{g}}\frac{1}{\sqrt{v}}dv$. Thus the integral I becomes

$$I = \sqrt{\frac{g}{3}} \frac{c^3 e^{-3f/g}}{g} \sqrt{\pi} \int \frac{2e^{w^2}}{\sqrt{\pi}} dw,$$
 (A.132)

$$=\frac{c^3 e^{-3f/g}}{\sqrt{3g}}\sqrt{\pi} erfi(w), \qquad (A.133)$$

where we have used the following definition

$$\int exp[(\alpha z + \beta)^2]dz = \frac{\sqrt{\pi}}{2\alpha} erfi(\alpha z + \beta), \qquad (A.134)$$

with $\alpha = 1$ and $\beta = 0$ [119]. Thus when one has a definite integral this definition becomes

$$erfi(\theta) = \frac{2}{\sqrt{\pi}} \int_0^{\theta} exp(z^2) dz.$$
 (A.135)

According to this definition erfi(0) = 0. Then by applying back substitutions $w = \sqrt{\frac{3v}{g}}$, v = f + glnu and $u = \frac{a'}{c}$ respectively we obtain

$$I = \sqrt{\frac{\pi}{3g}} c^3 e^{-3f/g} erfi[\sqrt{\frac{3(f + gln(\frac{a'}{c}))}{g}}].$$
 (A.136)

By writing our abbreviations in their original form $f = V_6 + a_{in}^6 \gamma(a_{in})$, $g = 6V_6$ and $c = a_{in}$ we have

$$t = \sqrt{\frac{3}{8\pi G}} a_{in}^3 exp[-\frac{1}{2} - \frac{a_{in}^6 \gamma(a_{in})}{2V_6}] \Big\{ erfi[\sqrt{\frac{1}{2}(1 + 6ln(\frac{a}{a_{in}}) + \frac{a_{in}^6 \gamma(a_{in})}{2V_6})}] - erfi[\sqrt{\frac{1}{2}(1 + \frac{a_{in}^6 \gamma(a_{in})}{2V_6})}] \Big\}.$$
(A.137)

As a result we have

$$a(t) = a_{in} exp \left\{ \frac{1}{6} \left[2 \left(erfi^{-1}(\mu t + b) \right)^2 - 1 - \frac{a_{in}^6 \gamma(a_{in})}{V_6} \right] \right\},\tag{A.138}$$

$$\mu = \sqrt{\frac{8\pi G}{3}} \frac{1}{a_{in}^3} exp[\frac{1}{2} + \frac{a_{in}^6 \gamma(a_{in})}{2V_6}], \qquad b = erfi[\frac{1}{2}\left(1 + \frac{a_{in}^6 \gamma(a_{in})}{V_6}\right)].$$
(A.139)

A.8. Cosmological Fluid

The energy-momentum conservation states that

$$\nabla_{\mu}T^{\mu\nu} = 0, \qquad (A.140)$$

where

$$T_{\mu\nu} = (\rho + p)u_{\mu}u_{\nu} - pg_{\mu\nu}.$$
 (A.141)

Then (A.140) becomes [122]

$$\dot{\rho} + 3(\frac{\dot{a}}{a})(\rho + p) = 0.$$
 (A.142)

The last equation is also known as the continuity equation in cosmology. One can write it in the following form

$$\frac{d(\rho a^3)}{da} = -3pa^2. \tag{A.143}$$

In cosmology it is usually assumed that each component of the fluid obeys the equation of state which is given by

$$p = \nu \rho. \tag{A.144}$$

Then (A.143) turns into the following form

$$\frac{d(\rho a^3)}{da} = -3\nu\rho a^2,\tag{A.145}$$

which has a solution

$$\rho \sim a^{-3(1+\nu)}.$$
(A.146)

It is known that $\nu = -1$ corresponds to cosmological constant while $\nu = -2/3$ stands for cosmic domain walls. Pressureless dust which is nonrelativistic matter has $\nu = 0$ and radiation which corresponds to relativistic particles and photons has $\nu = 1/3$. Stiff fluid can be defined with $\nu = 1$.

A.9. Redshift in Cosmology

Suppose we measure a wavelength of a observed distant galaxy as λ_{ob} where it emits the wavelength λ_{em} . Then redshift of this galaxy is calculated according to the following formula

$$z = \frac{\lambda_{ob} - \lambda_{em}}{\lambda_{em}}.$$
(A.147)

In [53] it is shown that the relation between the redshift and the scale factor of the universe is

$$1 + z = \frac{a(t_0)}{a(t_{em})} = \frac{1}{a(t_{em})},$$
(A.148)

where we have used the usual agreement such that $a(t_0) = 1$.

APPENDIX B: APPENDIX FOR CHAPTER 3

B.1. Action Variation for Brans-Dicke Cosmology

We use the Brans-Dicke action in the following form

$$S = \int d^4x \sqrt{-g} \left[-\frac{1}{8\omega} \phi^2 R + \frac{1}{2} g^{\mu\nu} \partial_\mu \phi \partial_\nu \phi - V(\phi) + L_M \right].$$
(B.1)

B.1.1. Action Variation with Respect to Metric

Let us start variation of the action with respect to metric by using $R = R_{\mu\nu}g^{\mu\nu}$

$$\delta S = \int d^4x \left\{ \left(\delta \sqrt{-g} \right) \left[-\frac{1}{8\omega} \phi^2 R + \frac{1}{2} g^{\mu\nu} \partial_\mu \phi \partial_\nu \phi - V(\phi) + L_M \right]$$

$$+ \left[-\frac{1}{8\omega} \phi^2 (\delta R_{\mu\nu}) g^{\mu\nu} - \frac{1}{8\omega} \phi^2 R_{\mu\nu} (\delta g^{\mu\nu}) + \frac{1}{2} (\delta g^{\mu\nu}) \partial_\mu \phi \partial_\nu \phi \right] \right\}$$

$$+ \int d^4x \sqrt{-g} \frac{\delta g^{\mu\nu} T_{\mu\nu}}{2},$$
(B.2)

where we have used

$$T_{\mu\nu} = \frac{2}{\sqrt{-g}} \frac{\delta S_M}{\delta g^{\mu\nu}}, \qquad S_M = \int d^4x \sqrt{-g} L_M. \tag{B.3}$$

Then we use (A.28) and (A.23) we obtain

$$\delta S = \int d^4x \sqrt{-g} \delta(g^{\mu\nu}) \Big\{ -\frac{1}{8\omega} \phi^2 G_{\mu\nu} - \frac{1}{4} g_{\mu\nu} \partial_\alpha \phi \partial^\alpha \phi + g_{\mu\nu} \frac{V(\phi)}{2} \tag{B.4}$$

$$+\frac{1}{2}\partial_{\mu}\phi\partial_{\nu}\phi + \frac{T_{\mu\nu}}{2}\Big\} - \frac{1}{8\omega}\int d^{4}x\sqrt{-g}\phi^{2}(\delta R_{\mu\nu})g^{\mu\nu}.$$
 (B.5)

In the original Brans-Dicke and Jordan variations, S was taken as a functional of $g^{\mu\nu}$ and ϕ . We would like to notice that the palatini approach gives different field

equations [187-189]. Thus one has [56]

$$\int d^4x \sqrt{-g} \phi^2(\delta R_{\mu\nu}) g^{\mu\nu} = \int d^4x \sqrt{-g} \delta g^{\mu\nu} (g_{\mu\nu} \Box \phi^2 - \nabla_\mu \nabla_\nu \phi^2), \qquad (B.6)$$

where one uses (A.36), $\nabla_{\lambda}g^{\mu\nu} = 0$ and integration by parts. Thus we obtain

$$\frac{\delta S}{\delta g^{\mu\nu}} = -\frac{1}{8\omega} \Big[G_{\mu\nu} \phi^2 + g_{\mu\nu} \Box \phi^2 - \nabla_\mu \nabla_\nu \phi^2 \Big] - \frac{1}{4} g_{\mu\nu} g^{\alpha\beta} \partial_\alpha \phi \partial_\beta \phi \tag{B.7}$$

$$+ g_{\mu\nu} \frac{V(\phi)}{2} + \frac{1}{2} \partial_{\mu} \phi \partial_{\nu} \phi + \frac{T_{\mu\nu}}{2} = 0.$$
 (B.8)

We choose $\mu = \nu = 0$ and by using (A.24), (A.44) and (B.17) we obtain

$$\frac{3}{4\omega}(\frac{\dot{a}^2}{a^2} + \frac{k}{a^2})\phi^2 + \frac{3}{2\omega}\frac{\dot{a}}{a}\phi\dot{\phi} - \frac{\dot{\phi}^2}{2} - V(\phi) = \rho_m.$$
 (B.9)

Then we choose $\mu = \nu = i$ and by using (A.25), (A.44) and (B.18) we obtain

$$-\frac{1}{4\omega}\phi^2\left(2\frac{\ddot{a}}{a} + \frac{\dot{a}^2}{a^2} + \frac{k}{a^2}\right) - \frac{1}{\omega}\frac{\dot{a}}{a}\dot{\phi}\phi - \frac{1}{2\omega}\ddot{\phi}\phi - \left(\frac{1}{2} + \frac{1}{2\omega}\right)\dot{\phi}^2 + V(\phi) = p_m.$$
 (B.10)

<u>B.1.1.1.</u> Some Necessary Calculations. To find $(g_{\mu\nu}\Box\phi^2 - \nabla_{\mu}\nabla_{\nu}\phi^2)$ we start with

$$\Box \phi^2 = g^{\mu\nu} \nabla_\mu \nabla_\nu \phi^2, \tag{B.11}$$

$$= g^{00} \nabla_0 \nabla_0 \phi^2 + g^{ii} \nabla_i \nabla_i \phi^2. \tag{B.12}$$

Before going further we write

$$g_{00} = 1, \quad g_{11} = -\frac{a^2}{1 - kr^2}, \quad g_{22} = -a^2r^2, \quad g_{33} = -a^2r^2Sin^2\theta,$$
 (B.13)

$$g^{00} = 1, \quad g^{11} = -\frac{(1-kr^2)}{a^2}, \quad g^{22} = -\frac{1}{a^2r^2}, \quad g^{33} = -\frac{1}{a^2r^2Sin^2\theta}.$$
 (B.14)

We choose $\mu = \nu = 0$

$$(g_{00}\Box - \nabla_0\nabla_0)\phi^2 = g_{00}(g^{00}\nabla_0\nabla_0\phi^2 + g^{ii}\nabla_i\nabla_i\phi^2) - \nabla_0\nabla_0\phi^2, \qquad (B.15)$$
$$= g_{00}g^{ii}\nabla_i\nabla_i\phi^2,$$
$$= g^{ii}\nabla_i\nabla_i\phi^2.$$

Then we use the definition given in (A.77) and we obtain

$$g^{ii} \nabla_i \nabla_i \phi^2 = g^{ii} (\partial_i \partial_i \phi^2 - \partial_k \phi^2 \Gamma^k_{ii}),$$
(B.16)
$$= -g^{ii} \partial_0 \phi^2 \Gamma^0_{ii},$$
$$= -2\phi \dot{\phi} [g^{11} \Gamma^0_{11} + g^{22} \Gamma^0_{22} + g^{33} \Gamma^0_{33}],$$
$$= -2\phi \dot{\phi} \{ -\frac{(1-kr^2)}{a^2} \frac{a\dot{a}}{(1-kr^2)} - \frac{1}{a^2r^2} a\dot{a}r^2 - \frac{1}{a^2r^2Sin^2\theta} a\dot{a}r^2Sin^2\theta \},$$
$$= 6\frac{\dot{a}}{a}\phi \dot{\phi},$$

where we have assumed that scalar field depends only on the time and we have used (B.13-B.14). Hence

$$(g_{00}\Box - \nabla_0\nabla_0)\phi^2 = 6\frac{\dot{a}}{a}\phi\dot{\phi}.$$
(B.17)

Now we choose $\mu = \nu = i$. By following the similar steps we obtain

$$(g_{ii}\Box - \nabla_i\nabla_i)\phi^2 = g_{ii}(2\dot{\phi}^2 + 2\phi\ddot{\phi} + 4\phi\dot{\phi}\frac{\dot{a}}{a}).$$
(B.18)

B.1.2. Action Variation with Respect to the Scalar Field

As we explained before action variation with respect to the scalar field results in Euler-Lagrange equations. In Brans-Dicke cosmology we have

$$S_{\phi} = \int d^4x \sqrt{-g} \left[-\frac{1}{8\omega} \phi^2 R + \frac{1}{2} g^{\mu\nu} \partial_{\mu} \phi \partial_{\nu} \phi - V(\phi) \right]. \tag{B.19}$$

The differences between this action and the action given in (A.38) is the term $-\frac{1}{8\omega}\phi^2 R$. It only modifies the following term as

$$\frac{\partial \mathcal{L}}{\partial \phi} = -\frac{1}{4\omega} \phi R - \frac{dV(\phi)}{\phi}.$$
 (B.20)

Therefore Euler-Lagrange equation given in (A.79) becomes

$$\ddot{\phi} + 3\frac{\dot{a}}{a}\dot{\phi} + \frac{dV(\phi)}{d\phi} - \frac{3}{2\omega}\left(\frac{\ddot{a}}{a} + \frac{\dot{a}^2}{a^2} + \frac{k}{a^2}\right)\phi = 0, \tag{B.21}$$

where we have used $R = -6(\frac{k}{L^2a^2} + \frac{\dot{a}^2}{a^2} + \frac{\ddot{a}}{a})$ which is derived in Appendix A.

B.2. The Bernoulli Equation

A first order differential equation which is in the following form

$$\frac{dy}{dx} + P(x)y = Q(x)y^n, \tag{B.22}$$

is known as a Bernoulli equation. It is linearized by a transformation

$$v = y^{1-n}.\tag{B.23}$$

$$\frac{dv}{dx} + (1-n)P(x)v = (1-n)Q(x).$$
(B.24)

B.3. Some Necessary Calculations

In Section 3.3.1.1 Equation (3.16) is given as

$$\gamma'(a) + \frac{2a[(1+2\omega)\phi^{'2}(a) + \phi(a)\phi^{''}(a)]}{\phi(a)(\phi(a) + a\phi^{'}(a))}\gamma(a) = \frac{2(2\omega\rho^{'}a^{3} + 3k\phi^{2}(a))}{3a^{3}\phi(a)(\phi(a) + a\phi^{'}(a))}.$$
 (B.25)

For simplification we choose

$$\alpha = \phi^2 + a\phi\phi'. \tag{B.26}$$

Then we have

$$\phi\phi'' = \frac{\alpha' - 3\phi\phi' - a\phi'^2}{a}.$$
 (B.27)

Therefore we can write

$$\tilde{P}(a) = \frac{2a[(1+2\omega)\phi'^{2}(a) + \phi(a)\phi''(a)]}{\phi(a)(\phi(a) + a\phi'(a))},$$

$$= 2\frac{\alpha'}{\alpha} + \frac{2[2\omega a\phi'^{2} - 3\phi\phi']}{\phi(\phi + a\phi')},$$

$$= 2\frac{\alpha'}{\alpha} + P(a),$$
(B.28)

where

$$P(a) = \frac{2[2\omega a\phi'^2 - 3\phi\phi']}{\phi(\phi + a\phi')}.$$
 (B.29)

Thus we find

$$\gamma(a) = exp\Big(-\int_{a_{in}}^{a} \tilde{P}(a')da'\Big)\Big\{\int_{a_{in}}^{a} exp\Big(\int_{a_{in}}^{a'} \tilde{P}(a'')da''\Big)[\frac{2(2\omega\rho'a^{3}+3k\phi^{2}(a))}{3a^{3}\phi(a)(\phi(a)+a\phi'(a))}]da'$$
(B.30)
$$+\gamma(a_{in})\Big\}.$$

By using (B.28) we obtain

$$exp\left(-\int_{a_{in}}^{a} \tilde{P}(a')da'\right) = exp\left(-\int_{a_{in}}^{a} \left[\frac{2\alpha'}{\alpha} + P(a')\right]da'\right),$$
(B.31)
$$= \frac{\left[\phi(a_{in})\left(\phi(a_{in}) + a_{in}\phi'(a_{in})\right)\right]^{2}}{\left[\phi(a)\left(\phi(a) + a\phi'(a)\right)\right]^{2}}exp\left(-\int_{a_{in}}^{a} P(a')da'\right).$$

As a result we have

$$\gamma(a) = \frac{exp\left[-\int_{a_{in}}^{a} P(a')da'\right]}{\left[\phi(\phi + a\phi')\right]^{2}} \left\{\int_{a_{in}}^{a} exp\left[\int_{a_{in}}^{a'} P(a'')da''\right]Q(a')da' + \tilde{\gamma}(a_{in})\right\}, \quad (B.32)$$

$$P(a) = \frac{2[2\omega a\phi'^{2}(a) - 3\phi(a)\phi'(a)]}{\phi(a)(\phi(a) + a\phi'(a))},$$

$$Q(a) = \frac{2(2\omega\rho'a^{3} + 3k\phi^{2}(a))\left[\phi(a)(\phi(a) + a\phi'(a))\right]}{\phi(a)(\phi(a) + a\phi'(a))}, \quad (B.33)$$

$$\tilde{\gamma}(a_{in}) = \gamma(a_{in}) \left[\phi(a_{in}) \left(\phi(a_{in}) + a_{in} \phi'(a_{in}) \right) \right]^2.$$
(B.34)

APPENDIX C: APPENDIX FOR CHAPTER 4

Now we will solve the differential equation which is given in (4.23)

$$t^{2}\ddot{\theta} + 2\omega t\dot{\theta} - \frac{4\omega}{3}(2 + \lambda B^{2})\theta = -2t^{2}.$$
 (C.1)

It is recognized as non-homogeneous Cauchy-Euler equation. Thus we take

$$t = e^x \Rightarrow tD = D_x$$
, and $t^2D^2 = D_x(D_x - 1)$. (C.2)

First, we take right side zero so (C.1) becomes

$$[D_x^2 + (2\omega - 1)D_x - \frac{4\omega}{3}(2 + \lambda B^2)]\theta = 0.$$
 (C.3)

Then substitution $\theta = e^{mx}$ gives us the following auxiliary equation

$$m^{2} + (2\omega - 1)m - \frac{4\omega}{3}(2 + \lambda B^{2}) = 0.$$
 (C.4)

It's solution yields

$$m_{\pm} = \frac{1}{2} - \omega \pm \sqrt{\omega^2 - \omega + \frac{1}{4} + \frac{4\omega}{3}(2 + \lambda B^2)}.$$
 (C.5)

Hence we obtain the complementary solution as

$$\theta_c(t) = c_1 t^{m_+} + c_2 t^{m_-}.$$
(C.6)

where c_1 and c_2 are integration constants.

By inspection the particular solution is found as

$$\theta_p(t) = -\frac{3t^2}{2\omega(1-\lambda B^2) + 3}.$$
 (C.7)

Therefore

$$\theta(t) = \theta_c + \theta_p(t),$$
(C.8)
= $c_1 t^{m_+} + c_2 t^{m_-} + \frac{3t^2}{2\omega(\lambda B^2 - 1) - 3}.$
APPENDIX D: APPENDIX FOR CHAPTER 5

In chapter 5 we have introduced the right translation operator as u_{\pm} which satisfies

$$u_{+}|n\rangle = |n+1\rangle, \qquad n = 0, 1, ..., d-2,$$
 (D.1)

$$u_+ |d-1\rangle = 0.$$
 (D.2)

Similarly, the left translation operator have been denoted u_{-} which satisfies

$$u_{-}|n\rangle = |n-1\rangle, \qquad n = 1, ..., d-1,$$
 (D.3)

$$u_{-}\left|0\right\rangle = 0.\tag{D.4}$$

Then we have switched our notation as $u_{+} = u^{\dagger}$ and u = u. By regrading rules given in (D.1-D.4) one can easily construct the matrix form of the translation operators for d = 2 as

$$u^{\dagger} = \begin{pmatrix} 0 & 0 \\ 1 & 0 \end{pmatrix}, \qquad u = \begin{pmatrix} 0 & 1 \\ 0 & 0 \end{pmatrix}.$$
(D.5)

Then we obtain

$$u^{\dagger} + u = \begin{pmatrix} 0 & 1 \\ 1 & 0 \end{pmatrix} = \sigma_1, \tag{D.6}$$

$$i(u^{\dagger} - u) = \begin{pmatrix} 0 & -i \\ i & 0 \end{pmatrix} = \sigma_2, \qquad (D.7)$$

$$uu^{\dagger} - u^{\dagger}u = \begin{pmatrix} 1 & 0 \\ 0 & -1 \end{pmatrix} = \sigma_3.$$
 (D.8)

APPENDIX E: APPENDIX FOR CHAPTER 6

E.1. Projection Operators in Terms of a and a^{\dagger}

We have found two projection operators P_n and R_n ;

$$P_n = a^{\dagger n} a^n \quad \text{and} \quad R_n = a^n a^{\dagger n}. \tag{E.1}$$

The properties

$$P_m a^n = a^{\dagger n} P_m = 0 \quad \text{for} \quad n+m \ge d, \tag{E.2}$$
$$a^n R_m = R_m a^{\dagger n} = 0 \quad \text{for} \quad n+m \ge d,$$

are immediate results of definition of the projection operators and the algebra property $a^d = a^{\dagger d} = 0.$

The following properties have already been proved in chapter 5 and in [181]

$$P_n P_m = P_m \quad \text{where} \quad m \ge n,$$
 (E.3)

$$P_m a^{\dagger} = a^{\dagger} P_{m-1}$$
 and $P_m a = a P_{m+1}$. (E.4)

Now, we will prove $R_n R_m = R_m$ where $m \ge n$. Our method is proof by induction.

For n = 1,

$$R_1 R_m = (aa^{\dagger})(a^m a^{\dagger m}), \qquad (E.5)$$
$$= a(a^{\dagger}a)a^{m-1}a^{\dagger m},$$

$$= a(1 - a^{d-1}a^{\dagger d-1})a^{m-1}a^{\dagger m},$$
$$= aa^{m-1}a^{\dagger m},$$
$$= a^m a^{\dagger m},$$
$$R_1 R_m = R_m,$$

where we have used $a^d = 0$. For n = l, we assume

$$R_l R_m = R_m \quad \text{for} \quad m \ge l. \tag{E.6}$$

For n = l + 1, we have

$$R_{l+1}R_{m} = (a^{l+1}a^{\dagger l+1})(a^{m}a^{\dagger m}), \qquad (E.7)$$

$$= a(a^{l}a^{\dagger l})a^{\dagger}a(a^{m-1}a^{\dagger m-1})a^{\dagger}, \qquad (E.7)$$

$$= aR_{l}(a^{\dagger}a)R_{m-1}a^{\dagger}, \qquad (E.7)$$

$$= aR_{l}(1 - a^{d-1}a^{\dagger m-1})R_{m-1}a^{\dagger}, \qquad (E.7)$$

$$= aR_{l}(1 - a^{d-1}a^{\dagger m-1})R_{m-1}a^{\dagger}, \qquad (E.7)$$

$$= aR_{l}R_{m-1}a^{\dagger} - aR_{l}R_{d-1}R_{m-1}a^{\dagger}, \qquad (E.7)$$

where we have used (6.2) and (E.6). By using $aR_{d-1} = a^d a^{\dagger d-1} = 0$ and the assumption for n = l with the fact that $l + 1 \leq m$ implies $l \leq m - 1$, so we have $R_l R_{m-1} = R_{m-1}$ and $aR_{m-1}a^{\dagger} = R_m$. Therefore

$$R_{l+1}R_m = R_m \quad \text{Q.E.D.}. \tag{E.8}$$

Then we will show that $R_m R_n = R_m$ for $m \ge n$ by using the relation $R_n = 1 - P_{d-n}$ which is Equation (6.7). Thus

$$R_m R_n = (1 - P_{d-m})(1 - P_{d-n})$$
(E.9)
= 1 - P_{d-n} - P_{d-m} + P_{d-m} P_{d-n}

using $P_n P_m = P_m$ where $m \ge n$ with the fact that $m \ge n$ implies $d - n \ge d - m$

$$R_m R_n = 1 - P_{d-n} - P_{d-m} + P_{d-n}$$
 (E.10)
= $1 - P_{d-m}$
= R_m Q.E.D..

By using the last two proofs, we conclude that

$$R_n R_m = R_j$$
 where $j = \max(n, m)$. (E.11)

Similarly we will show that $P_m P_n = P_m$ where $m \ge n$ by using $P_n = 1 - R_{d-n}$ which is Equation (6.7). Thus

$$P_m P_n = (1 - R_{d-m})(1 - R_{d-n}),$$

$$= 1 - R_{d-m} - R_{d-n} + R_{d-m} R_{d-n},$$
(E.12)

using $R_n R_m = R_m$ where $m \ge n$ with the fact that $m \ge n$ implies $d - n \ge d - m$

$$P_m P_n = 1 - R_{d-m} - R_{d-n} + R_{d-n},$$
 (E.13)
= $1 - R_{d-m},$
= P_m Q.E.D..

This result and equation given by (E.3) can be expressed in one equation as

$$P_n P_m = P_j$$
 where $j = \max(n, m)$. (E.14)

Now, we will calculate relations between the projection operator R_m and shift operators a and a^{\dagger} .

$$R_{m}a = a^{m}a^{\dagger m}a, \qquad (E.15)$$

$$= a^{m}a^{\dagger m-1}(a^{\dagger}a), \qquad (E.15)$$

$$= a(a^{m-1}a^{\dagger m-1})P_{1}, \qquad (E.15)$$

$$= aR_{m-1}(1 - R_{d-1}), \qquad (E.15)$$

$$= aR_{m-1}(1 - R_{d-1}), \qquad (E.15)$$

$$= aR_{m-1}(1 - R_{d-1}), \qquad (E.15)$$

where we have used (6.7), (E.11) and the fact that $aR_{d-1} = a^d a^{\dagger d-1} = 0$. Then

$$a^{\dagger}R_{m} = a^{\dagger}a^{m}a^{\dagger m}, \qquad (E.16)$$

$$= (a^{\dagger}a)(a^{m-1}a^{\dagger m-1})a^{\dagger}, \qquad (E.16)$$

$$= P_{1}R_{m-1}a^{\dagger}, \qquad (E.16)$$

$$= P_{1}R_{m-1}a^{\dagger}, \qquad (E.16)$$

$$= R_{m-1}a^{\dagger}, \qquad (E.16)$$

$$= R_{m-1}a^{\dagger}, \qquad (E.16)$$

where we have used $R_{d-1}a^{\dagger} = a^{d-1}a^{\dagger d} = 0$. Let's summarize what we have derived about the projection operators P_n and R_n up to now;

$$P_n = a^{\dagger n} a^n, \quad P_0 = 1, \quad P_m a^{\dagger} = a^{\dagger} P_{m-1}, \quad a P_m = P_{m-1} a,$$
$$P_n P_m = P_j \quad \text{where} \quad j = \max(n, m),$$
$$P_m a^n = a^{\dagger n} P_m = 0 \quad \text{for} \quad n+m \ge d,$$

and

$$R_n = a^n a^{\dagger n}, \quad R_0 = 1, \quad R_m a^{\dagger} = a^{\dagger} R_{m+1}, \quad a R_m = R_{m+1} a,$$
$$R_n R_m = R_j \quad \text{where} \quad j = \max(n, m),$$
$$a^n R_m = R_m a^{\dagger n} = 0 \quad \text{for} \quad n+m \ge d,$$

where

$$P_n = 1 - R_{d-n}$$
 and $R_n = 1 - P_{d-n}$.

E.2. Projection Operators in Terms of \boldsymbol{U} and \boldsymbol{V}

We have found two projection operators \mathscr{P}_n and \mathscr{R}_n ;

$$\mathscr{P}_n = \frac{(1+q^nV+q^{2n}V^2+\dots+q^{(d-1)n}V^{(d-1)})}{d},$$
 (E.17)

$$\mathscr{R}_n = 1 - \mathscr{P}_n. \tag{E.18}$$

Then

$$(\mathscr{P}_{n})^{2} = \frac{(1+q^{n}V+q^{2n}V^{2}+\dots+q^{(d-1)n}V^{(d-1)})}{d}$$
(E.19)
+ $\frac{(q^{n}V+q^{2n}V^{2}+q^{3n}V^{3}+\dots+q^{dn}V^{d})}{d}$
+ \dots ,
+ $\frac{q^{(d-1)n}V^{(d-1)}+q^{dn}V^{d}+q^{(d+1)n}V^{(d+1)}+\dots+q^{2(d-n)}V^{2(d-1)}}{d}$,
= $d\frac{(1+q^{n}V+q^{2n}V^{2}+\dots+q^{(d-1)n}V^{(d-1)})}{d}$,
= \mathscr{P}_{n} ,

where we have used $V^d = 1$ and $q^d = 1$.

By using definition of \mathscr{P}_n and Schwinger equation VU = qUV, one can easily obtain

$$\mathscr{P}_n U^m = U^m \mathscr{P}_{n+m}. \tag{E.20}$$

Hermitian conjugate of \mathscr{P}_n is easily calculated as,

$$(\mathscr{P}_n)^{\dagger} = \frac{(1 + (q^n)^{\dagger}V^{\dagger} + (q^{2n})^{\dagger}(V^2)^{\dagger} + \dots + (q^{(d-1)n})^{\dagger}(V^{(d-1)})^{\dagger})}{d}, \qquad (E.21)$$
$$= \frac{(1 + q^{d-n}V^{d-1} + q^{(d-2)n}V^{d-2} + \dots + q^nV)}{d},$$
$$= \mathscr{P}_n,$$

where we have used the unitary property of V and properties of complex number $q = exp(\frac{2\pi i}{d})$. By using hermicity property of \mathscr{P}_n we will calculate,

$$(\mathscr{P}_n U^m = U^m \mathscr{P}_{n+m})^{\dagger}, \qquad (E.22)$$
$$U^{\dagger m} \mathscr{P}_n = \mathscr{P}_{n+m} U^{\dagger m}.$$

The other property of \mathcal{P}_n is found by the following steps,

$$\mathcal{P}_{0} + \mathcal{P}_{1} + \mathcal{P}_{2} + \dots + \mathcal{P}_{(d-1)} = \frac{(1+V+V^{2}+\dots+V^{(d-1)})}{d}$$
(E.23)
+ $\frac{(1+qV+q^{2}V^{2}+\dots+q^{(d-1)}V^{(d-1)})}{d}$
+ $\frac{(1+q^{2}V+q^{4}V^{2}+\dots+q^{2(d-1)}V^{(d-1)})}{d}$
...
+ $\frac{(1+q^{d-1}V+q^{2(d-1)}V^{2}+\dots+q^{(d-1)(d-1)}V^{(d-1)})}{d}$,

$$= d/d + (1 + q + q^{2} + \dots + q^{(d-1)})V/d$$

+ $(1 + q^{2} + q^{4} + \dots + q^{2(d-1)})V^{2}/d$
...
+ $(1 + q^{(d-1)} + q^{2(d-1)} + \dots + q^{(d-1)(d-1)})V^{(d-1)}/d.$
= 1.

In the last equation we used the fact that the sum of roots of unity gives zero,

$$(1+q+q^2) + \dots + q^{d-1} = 0$$
 with $q = exp(\frac{2\pi i}{d}).$ (E.24)

Furthermore, when we take unity as $1 = exp(2\pi in)$ where *n* is a integer, its roots will be $1, q^n, q^{2n}, \dots, q^{n(d-1)}$. Thus the sum of the terms appear in parentheses in the last lines are equal to 0.

Then sum of \mathcal{R}_n given by Equation (E.18) is found easily,

$$\mathcal{R}_0 + \mathcal{R}_1 + \dots + \mathcal{R}_{d-1} = (1 - \mathcal{P}_0) + (1 - \mathcal{P}_1) + \dots + (1 - \mathcal{P}_{d-1}), \quad (E.25)$$
$$= d - 1.$$

Now we will show that $\mathscr{P}_n \mathscr{P}_m = 0$ unless $n \neq m$ by direct substitution of definition of projection operators

$$\mathcal{P}_{n}\mathcal{P}_{m} = (1+q^{n}V+q^{2n}V^{2}+\dots+q^{(d-1)n}V^{d-1})(1+q^{m}V+q^{2m}V^{2}+\dots$$
(E.26)
+ $q^{(d-1)m}V^{d-1})/d^{2}$
= $\{(1+q^{m}V+q^{2m}V^{2}+\dots+q^{(d-1)m}V^{d-1})$
+ $(q^{n}V+q^{n+m}V^{2}+q^{n+2m}V^{3}\dots+q^{(d-1)m+n}V^{d})$
+ $(q^{2n}V^{2}+q^{2n+m}V^{3}+q^{2n+2m}V^{4}\dots+q^{(d-1)m+2n}V^{d+1})$

$$\begin{split} &+\cdots \\ &+ (q^{(d-1)n}V^{d-1} + q^{(d-1)n+m}V^d + q^{(d-1)n+2m}V^{(d+1)} \cdots + q^{(d-1)(m+n)}V^{2d-2}) \}/d^2, \\ &= \{1 + (q^{(d-1)m+n} + q^{(d-2)m+2n} + \cdots + q^{(d-1)n+m})V^d \\ &+ (q^m + q^n + q^{(d-1)m+2n} + \cdots + q^{(d-1)n+2m})V \\ &+ (q^{2m} + q^{n+m} + q^{2n} + \cdots + q^{(d-1)n+3m})V^2 \\ &+ \cdots \\ &+ (q^{(d-1)m} + q^{(d-2)m+n} + \cdots + q^{(d-1)n})V^{d-1}\}/d^2, \end{split}$$

with the help of $V^{d+n} = V^n$. Now replace n - m by k and use $q^d = 1$

$$\mathcal{P}_{n}\mathcal{P}_{m} = \{(1+q^{k}+q^{2k}+\dots+q^{(d-1)k})$$

$$+ q^{m}(1+q^{k}+q^{2k}+\dots+q^{(d-1)k})V$$

$$+ q^{2m}(1+q^{k}+q^{2k}+\dots+q^{(d-1)k})V^{2}$$

$$+ \dots$$

$$+ q^{(d-1)m}(1+q^{k}+q^{2k}+\dots+q^{(d-1)k})V^{d-1}\}/d^{2}.$$
(E.27)

Furthermore, taking unity as $1, 1 + q^k + \cdots + q^{(d-1)k}$ will be equal to Equation (E.24) in different order. Hence $(1 + q^k + \cdots + q^{(d-1)k}) = 0$. As a result each parentheses in the Equation (E.27) are equal to zero except k = 0 (corresponds n = m). At the exception each paratheses sum to d. We can summarize our results as

$$\mathscr{P}_{n}\mathscr{P}_{m} = 0 \quad \text{for } m \neq n,$$

$$\mathscr{P}_{n}\mathscr{P}_{n} = d(1 + q^{n}V + \dots + q^{(d-1)n}V^{d-1})/d^{2},$$

$$= \mathscr{P}_{n}.$$
(E.28)

E.3. $M_N(C)$ in Terms of a and a^{\dagger}

The standard basis of $M_N(C)$ is given by the operator e_{ij} which satisfies

$$e_{ij}e_{kl} = \delta_{jk}e_{il}, \qquad e_{ij}^{\dagger} = e_{ji}, \tag{E.29}$$

 e_{mn} can be expressed in terms of a and a^{\dagger} as,

$$e_{mn} = a^{\dagger m} R_{d-1} a^n \quad m, n = 0, 1, \cdots, d-1.$$
 (E.30)

Then,

$$e_{ij}e_{kl} = (a^{\dagger i}R_{d-1}a^{j})(a^{\dagger k}R_{d-1}a^{l}), \qquad (E.31)$$

$$= (a^{\dagger i}a^{j}R_{d-1-j})(R_{d-1-k}a^{\dagger k}a^{l}),$$

$$= a^{\dagger i}a^{j}(R_{d-1-j}R_{d-1-k})a^{\dagger k}a^{l},$$
if $j = k = a^{\dagger i}a^{j}R_{d-1-j}a^{\dagger j}a^{l},$

$$= a^{\dagger i}R_{d-1}a^{l},$$

$$= e_{il},$$
if $j > k = a^{\dagger i}a^{j}R_{d-1-k}a^{\dagger k}a^{l},$

$$= a^{\dagger i}a^{j}a^{\dagger k}R_{k-1}a^{l},$$

$$= a^{\dagger i}a^{j-k}R_{d-1}a^{l},$$

$$= 0,$$
if $j < k = a^{\dagger i}a^{j}R_{d-1-j}a^{\dagger k}a^{l},$

$$= a^{\dagger i} R_{d-1} a^{j} a^{\dagger k} a^{l},$$

= $a^{\dagger i} R_{d-1} R_{j} a^{k-j} a^{l},$
= $a^{\dagger i} R_{d-1} a^{k-j} a^{l},$
= $a^{\dagger i} R_{d-1} R_{j} a^{k-j} a^{l},$
= 0,

where we have used (6.9). Thus $e_{ij}e_{kl} = \delta_{jk}e_{il}$. It is the time to check second part of Equation (E.29)

$$(e_{ij} = a^{\dagger i} R_{d-1} a^{j})^{*},$$
 (E.32)
 $e_{ij}^{*} = a^{\dagger j} R_{d-1}^{*} a^{i},$
 $e_{ij}^{*} = a^{\dagger j} R_{d-1} a^{i},$
 $e_{ij}^{*} = e_{ji}$ Q.E.D..

E.4. $M_N(C)$ in Terms of U and V

 e_{mn} can be written in terms of operators U and V as

$$e_{mn} = \begin{cases} U^{m-n} \mathscr{P}_{d-n} & \text{for} \quad m > n, \\ \mathscr{P}_{d-n} & \text{for} \quad m = n, \\ U^{\dagger n-m} \mathscr{P}_{d-n} & \text{for} \quad m < n, \end{cases}$$
(E.33)

with

$$\mathscr{P}_n = \frac{(1+q^nV+q^{2n}V^2+\dots+q^{(d-1)n}V^{(d-1)})}{d}.$$

$$e_{ij}e_{kl} = U^{i-j}\mathscr{P}_{d-j}U^{k-l}\mathscr{P}_{d-l}, \qquad (E.34)$$

$$= U^{i-j}U^{k-l}\mathscr{P}_{d-j+k-l}\mathscr{P}_{d-l},$$
if $j = k = U^{i-j}U^{j-l}\mathscr{P}_{d-l}\mathscr{P}_{d-l},$

$$= U^{i-l}\mathscr{P}_{d-l},$$

$$= e_{il},$$
if $j \neq k = U^{i-j}U^{k-l}\mathscr{P}_{d-j+k-l}\mathscr{P}_{d-l},$

$$= 0.$$

Thus $e_{ij}e_{kl} = \delta_{jk}e_{il}$.

Now let us check second part of the Equation (E.29) for m > n

$$(e_{mn} = U^{m-n} \mathscr{P}_{d-n})^{\dagger}, \qquad (E.35)$$

$$e_{mn}^{\dagger} = \mathscr{P}_{d-n}^{\dagger} U^{\dagger m-n}, \qquad (E.35)$$

$$e_{mn}^{\dagger} = \mathscr{P}_{d-n} U^{\dagger m-n}, \qquad (E.35)$$

$$e_{mn}^{\dagger} = U^{\dagger m-n} \mathscr{P}_{d-n-(m-n)}, \qquad (E.35)$$

$$e_{mn}^{\dagger} = U^{\dagger m-n} \mathscr{P}_{d-m}, \qquad (E.35)$$

$$e_{mn}^{\dagger} = U^{\dagger m-n} \mathscr{P}_{d-m}, \qquad (E.35)$$

where we have used (E.21), (E.22) and (6.1). Next for m = n we have

$$e_{ii}e_{kk} = \mathscr{P}_{d-i}\mathscr{P}_{d-k},$$
 (E.36)
if $i = k, = 1,$
if $i \neq k, = 0,$

where we have used (E.28). Thus $e_{ij}e_{kl} = e_{il}$. Now, hermicity of property of the projection operators given by Equation (E.21) imply the second part of the matrix algebra given by the Equation (E.29). Finally for m < n we have

$$e_{ij}e_{kl} = U^{\dagger j-i}\mathscr{P}_{d-j}U^{\dagger l-k}\mathscr{P}_{d-l}, \qquad (E.37)$$

$$= U^{\dagger j-i}U^{\dagger l-k}\mathscr{P}_{d-j-l+k}\mathscr{P}_{d-l},$$
if $j = k = U^{\dagger j-i}U^{\dagger l-k}\mathscr{P}_{d-l}\mathscr{P}_{d-l},$

$$= U^{\dagger l-i}\mathscr{P}_{d-l},$$

$$= e_{il},$$
if $j \neq k = U^{\dagger j-i}U^{\dagger l-k}\mathscr{P}_{d-j-l+k}\mathscr{P}_{d-l},$

$$= 0.$$

Thus we can conclude that $e_{ij}e_{kl} = \delta_{jk}e_{il}$.

Now let us check second part of the Equation (E.29) for m < n we have

$$(e_{mn} = U^{\dagger n - m} \mathscr{P}_{d - n})^{\dagger}, \qquad (E.38)$$
$$= \mathscr{P}_{d - n}^{\dagger} U^{n - m},$$
$$= U^{n - m} \mathscr{P}_{d - n + n - m},$$
$$= U^{n - m} \mathscr{P}_{d - m},$$
$$= e_{nm} \quad Q.E.D..$$